#### COMMONWEALTH OF KENTUCKY

#### BEFORE THE PUBLIC SERVICE COMMISSION

In the Matter of:

Electronic Investigation Of The Service, Rates And Facilities Of Kentucky Power Company ) Case No. 2021-00370

Kentucky Power Company's Update Of West Virginia Proceedings For The Period July 5, 2022-July 11, 2022

\*\*\*\*

Kentucky Power Company provides the following update regarding proceedings before the Public Service Commission of West Virginia:

There were no documents filed by either Wheeling Power Company or the Public Service Commission of West Virginia in Case No. 20-1040-E-CN.

Kentucky Power has reviewed the order entered by the Public Service Commission of West Virginia ("WVPSC") on July 1, 2022 in Case No. 21-0810-E-PC regarding Wheeling Power Company's ("Wheeling Power") petition seeking the West Virginia commission's consent and approval of the Mitchell Ownership Agreement and the Mitchell Operating Agreement between Kentucky Power and Wheeling Power. A copy of the order was filed with Kentucky Power's update dated July 5, 2022 as Exhibit 1.

The basis of the Operating Agreement approved by the WVPSC was not the proposed Mitchell Plant Operations and Maintenance Agreement as approved by this Commission and submitted by Wheeling Power for the WVPSC's consideration and approval, but was instead the existing Mitchell Operating Agreement ("Current Operating Agreement"), with modifications by

<sup>&</sup>lt;sup>1</sup> The Current Operating Agreement became effective on December 31, 2014 and, as noted in the May 3, 2022 Order, sets forth the operation, maintenance, and joint ownership rights and obligations of Kentucky Power and Wheeling Power with respect to the Mitchell Plant. *See* May 3, 2022 Order at 3-4.

the WVPSC. The Ownership Agreement approved by the WVPSC was the Ownership Agreement submitted by Wheeling for the WVPSC's consideration and approval, with modifications by the WVPSC that are different from the modifications required by this Commission in its May 3, 2022 Order in Case No. 2021-00421.<sup>2</sup>

Kentucky Power has determined based on its review that:

- (1) The WVPSC-approved agreements are inconsistent in certain material respects with the agreements approved by this Commission;
- (2) The WVPSC-approved agreements approved include modifications related to the WVPSC's findings in its July 1, 2022 Order regarding the value of Kentucky Power's ownership interests in and rights to the capacity and energy from the Mitchell Plant after December 31, 2028, which are inconsistent with this Commission's May 3, 2022 Order;
- (3) The WVPSC-approved agreements, inclusive of the unexecuted compliance agreements filed by Wheeling Power today, contain substantive conflicts between the provisions of the WVPSC-created Operating Agreement and the WVPSC-approved Ownership Agreement, which address similar issues in different ways; and
- (4) Absent consistent new agreements, approved by the two state jurisdictions, and the Federal Energy Regulatory Commission, the existing agreement remains in place as the document governing operations and the contractual relationship between the two owners.

Accordingly, Kentucky Power has advised Wheeling Power that it is unable to enter into the agreements approved by the WVPSC, and has authorized Wheeling Power to state Kentucky Power's position in the compliance filing that Wheeling Power will make today as directed by the WVPSC. A copy of Wheeling Power's compliance filing is attached as **Exhibit 1**.

Because of the conflicting regulatory requirements between the WVPSC and this Commission regarding the proposed agreements for the jointly owned Mitchell generating station, the Current Operating Agreement must remain in effect in accordance with its terms until a

2

<sup>&</sup>lt;sup>2</sup> In the Matter of: Electronic Application Of Kentucky Power Company For Approval Of Affiliate Agreements Related To The Mitchell Generating Station, Case No. 2021-00421 (the "May 3, 2022 Order").

resolution of post-2028 operating and ownership issues can be resolved in an acceptable manner, subject to further state commission review in future proceedings. Steps need to be taken at the Mitchell Plant in the immediate future to meet federal environmental requirements, to carry out both state commissions' orders on the Coal Combustion Residuals Rule ("CCR") and Effluent Limitation Guidelines ("ELG") projects, and to implement customer protections related to those investments. Accordingly, Kentucky Power will move forward in the short term, as it must, under the Current Operating Agreement to achieve the immediate goals needed for Wheeling to carry out the ELG project to keep the plant running through 2028 and beyond, and ensure cost allocation is properly performed related to the ELG work, consistent with this Commission's and the WVPSC's orders on the use of the plant to protect customers.

Kentucky Power also recognizes that additional regulatory efforts to achieve fully consistent agreements between the parties and jurisdictions at this time could cause delays that would not be in the best interests of the Mitchell Plant. Such further efforts could delay the transfer of permits and operational rights to Wheeling which could, in turn, delay the ELG work, create more outages where the plant will not run, and jeopardize the in-service date for the ELG work and result in the plant being taken out of service until construction is completed and in compliance with the federal ELG rules. Following the path described above in the near term, until the longer-term issues can be resolved, will support the current construction schedule and not cause delays that could require additional plant outages that reduce the availability and add to the expense of operating the plant.

Therefore, because Kentucky Power is unable to implement changes to the Current Operating Agreement absent a common agreement in both states, and given the looming dates for completing ELG investment, the existing Operating Committee will act as needed to ensure the near term needs can be accomplished under the Current Operating Agreement. The Operating

Committee intends to use resolutions related to plant operations to achieve these near term needs and customer protections related to plant operation. This near-term path forward is rooted in the points of consistency between the agreements approved by this Commission and the WVPSC in their respective orders to ensure plant operations prior to 2028. Both commissions observed in their respective orders that the Current Operating Agreement could be used to achieve the immediate goals needed to keep the plant running through 2028 and ensure that cost allocation is properly performed under their respective orders on the use of the plant for customers.<sup>3</sup> Both commissions also supported a core body of concepts related to the operation of the plant prior to December 31, 2028, which could be implemented under the Current Operating Agreement, including updating cost allocation to reflect the commissions' approved cost responsibility for CCR and ELG projects, and to designate Wheeling Power as plant operator.<sup>4</sup>

The provisions to be implemented by the Operating Committee through appropriate resolutions based on these areas of commonality include: (1) authorizing Wheeling Power to act as the operator of the Mitchell Plant; (2) effecting the transfer of ELG and other essential permits from Kentucky Power to Wheeling Power to facilitate timely ELG and other work; (3) the performance of ELG work at Mitchell by Wheeling Power at its expense and not the expense of Kentucky Power or Kentucky customers; and (4) the implementation of appropriate accounting procedures to ensure that Kentucky Power and Wheeling Power share in the investments and

\_

<sup>&</sup>lt;sup>3</sup> See May 3, 2022 Order at 10-11 ("As an initial matter, the Commission notes that amendments to the existing Mitchell Operating Agreement would have sufficed and, had Kentucky Power filed only an amended Operating Agreement, the Commission could have reached its decision on or near the February 2022 date requested by Kentucky Power." (citations omitted) Regarding the necessity of the new agreements, the Commission also stated, "Also, in hearing testimony, AEP agreed that revisions to the existing Operating Agreement could have addressed these issues, but that additional terms would have been needed.... Rather than being necessary to comply with the Commission's orders, the Ownership Agreement is instead merely convenient for AEP to satisfy requirements it created as a result of its agreements related to the proposed Liberty Acquisition of Kentucky Power." (citations omitted)

<sup>&</sup>lt;sup>4</sup> See May 3, 2022 Order at 12 ("The Commission concurs that changes to the existing Operating Agreement are needed regarding cost allocation and to designate Wheeling Power as operator.")

expenditures needed to operate Mitchell through 2028, including the CCR work, and that the investments and expenditures needed to implement the ELG work that will enable Mitchell to operate beyond 2028 will be assigned to Wheeling Power.

In order to facilitate the construction of the ELG project by Wheeling Power this summer, Kentucky Power and Wheeling Power will obtain Operating Committee approval later this month of resolutions that implement in good faith this Commission's orders assigning the costs of and responsibility for that project exclusively to Wheeling Power and shielding Kentucky Power and Kentucky customers from those expenses, as well as ensuring the applicable permits and responsibility for plant operations are assigned to Wheeling Power. Kentucky Power will advise the Commission in future updates of any resolutions adopted by the Operating Committee under the Current Operating Agreement to implement the standards and procedures set forth above and will provide copies of such resolutions in its updates. Kentucky Power also confirms that Liberty Utilities Co. ("Liberty") has been informed of the WVPSC's July 1, 2022 Order and this update. The companies are in discussions regarding these developments.

For the longer term, as a result of this Commission's decision to grant a CPCN only for the CCR project and not the ELG project, Kentucky Power will no longer take energy and capacity from the Mitchell Plant beyond 2028. However, Kentucky Power is required to seek future approval of the terms and conditions associated with the disposition of its ownership share in the Mitchell Plant based on the facts and circumstances that will emerge in the future. Conversely, the WVPSC has directed Wheeling Power to protect its unilateral investment in ELG that enables post-2028 operations and to dispatch the plant should Kentucky Power seek to lower its

<sup>&</sup>lt;sup>5</sup> Kentucky Power expressly acknowledges the Commission's directive in its October 28, 2021 Order in Case No 2021-00370 that Kentucky Power should seek Commission approval prior to any change to the Current Operating Agreement. Kentucky Power will advise the Commission of the resolutions passed by the Operating Committee and will see approval prior to entering into any amendment of the agreement.

output. Kentucky Power believes that this Commission's immediate objectives can be accomplished in the near term through the Operating Committee and, in the longer term, by future good faith discussions with Wheeling Power based on the Current Operating Agreement, which must continue in effect until a resolution acceptable to both owners and their respective commissions can be reached.

Additionally, on July 8, 2022 Wheeling Power filed a motion for extension of deadline and the WVPSC filed a corrective order amending the fifth ordering paragraph of the July 1, 2022 Order. These filings are attached as **Exhibit 2** and **Exhibit 3** respectively.

Subsequent updates will be filed at ten-day intervals or more frequently as circumstances require.

Respectfully submitted,

Mark R. Overstreet

Katie M. Glass

STITES & HARBISON PLLC

421 West Main Street

P.O. Box 634

Frankfort, Kentucky 40602-0634

Telephone: (502) 223-3477 Facsimile: (502) 779-8349

moverstreet@stites.com

kglass@stites.com

COUNSEL FOR

KENTUCKY POWER COMPANY

### EXHIBIT 1



ANNE C. BLANKENSHIP ATTORNEY AT LAW

P.O. BOX 1791 CHARLESTON, WV 25326

CHARLESTON OFFICE: (304) 344-5800 DIRECT DIAL: (304) 347-8352 FACSIMILE: (304) 344-9566 E-MAIL: acb@ramlaw.com

July 11, 2022

#### Via Hand Delivery

Karen Buckley, Executive Secretary Public Service Commission of West Virginia 201 Brooks Street Charleston, WV 25301

Re: Case No. 21-0810-E-PC (Closed Entry)

Appalachian Power Company and Wheeling Power Company

Dear Ms. Buckley:

Please find enclosed for filing in the above-referenced case an original and twelve copies of the Compliance Filing of Appalachian Power Company and Wheeling Power Company. Copies of this filing have been provided to all parties of record.

Thank you for your assistance. Please do not hesitate to contact me with any questions.

Very truly yours,

Anne C. Blankenship

Counsel for Appalachian Power Company and

Wheeling Power Company

Enclosure

cc: Service List

#### PUBLIC SERVICE COMMISSION OF WEST VIRGINIA CHARLESTON

CASE NO. 21-0810-E-PC (Closed Filing)

APPALACHIAN POWER COMPANY and WHEELING POWER COMPANY, public utilities.

Petition for Commission Consent and Approval to Enter into Ownership and Operating Agreements for the Mitchell Plant Pursuant to W. Va. Code § 24-2-12

## COMPLIANCE FILING OF APPALACHIAN POWER COMPANY AND WHEELING POWER COMPANY

COME NOW Appalachian Power Company ("APCo") and Wheeling Power Company ("WPCo") (together, "the Companies"), and make this filing in compliance with the ordering paragraphs of the Commission's July 1, 2022 Order in this proceeding ("the Order"). The Companies state the following:

1. On November 19, 2021, the Companies requested an order authorizing WPCo to enter into new agreements with Kentucky Power Company ("KPCo") regarding the operation and ownership of the Mitchell Power Generation Facility ("Mitchell" or the "Plant") with KPCo. WPCo informed the Commission that, with the two owners in different states and with different regulators that had issued inconsistent orders regarding investments to comply with the requirements of the Coal Combustion Residuals Rule ("CCR") and Effluent Limitation Guidelines ("ELG), it was seeking common agreements to move forward beyond the existing agreement. Absent consistent new agreements, approved by the two state jurisdictions, and the Federal Energy

Regulatory Commission ("FERC"), the existing agreement remains in place as the document governing operations and the contractual relationship between the two owners.

- 2. On Friday, July 1, 2022, the Commission entered its Order, which approved an Operating Agreement ("the Operating Agreement") among WPCo, KPCo, and American Electric Power Service Corporation ("AEPSC") as agent, governing the operation of the Plant, co-owned equally by WPCo and KPCo. The basis of the approved Operating Agreement was not the proposed Mitchell Plant Operations and Maintenance Agreement submitted by the Companies in this case for the Commission's consideration and approval and approved by the Kentucky PSC, but was instead the current Mitchell Operating Agreement, with modifications by the Commission.
- 3. The Order also approved a Mitchell Plant Ownership Agreement ("the Ownership Agreement") between WPCo and KPCo. The basis for the approved Ownership Agreement was the Ownership Agreement submitted by the Companies in this case for the Commission's consideration and approval, with modifications by the Commission.
- 4. The Companies thank the Commission for its careful consideration of their filing and the evidentiary record in this proceeding and for the specificity of its July 1, 2022 Order. The Companies appreciate that this case presents a special challenge because the agreements at issue require the approval of both this Commission and the Kentucky Public Service Commission ("the KPSC") before the two different owners can execute and enter into the agreements. The owners' ability to execute is challenged by the fact that this Commission and the KPSC have approved different agreements. WPCo has only been authorized to enter into the agreements approved by this Commission in its Order. KPCo has only been authorized to enter into the agreements approved by the KPSC in its Order of May 3, 2022 in Case No. 2021-00421. KPCo has advised WPCo that it is unable to enter into the agreements approved by this Commission. KPCo has also

advised that on July 5, 2022 it provided the KPSC with this Commission's Order and the versions of the Operating Agreement and Ownership Agreement approved therein, and on the date hereof provided a further update to the KPSC that the agreements approved by this Commission are not fully consistent with the KPSC's orders.

- Nonetheless, as directed by the Commission in its Order, the Companies attach hereto clean copies of the Operating Agreement and Ownership Agreement approved by the Order for compliance purposes only; as discussed above, WPCo is unable to implement these agreements by entering into them with KPCo. The attached Ownership Agreement has been modified by the Companies with non-substantive edits to assure that it is internally consistent and consistent with the Operating Agreement approved by this Commission. The Companies have also attached redlined versions of these agreements, as instructed by this Commission. However, the Companies note that while they have made the ministerial, non-substantive changes directed by the Commission to make the agreements facially consistent, making the agreements fully consistent would require substantive changes due to conflicts between the provisions of the Operating Agreement which address similar issues in different ways than the Ownership Agreement. The Mitchell Plant Operations and Maintenance Agreement and Mitchell Plant Ownership Agreement submitted by the Companies for approval were drafted to complement each other and ensure their provisions did not cover the same issue in different ways, as is the case with the agreements adopted by the Commission. As a result, the agreements provided in this compliance filing are not ready for execution as a commercial matter, even if KPCo would be amenable to them.
- 6. Inasmuch as these are commercial agreements between separate entities and each are required to follow the orders of their respective commissions, WPCo is unable to require KPCo to enter into the agreements approved by this Commission, the same as KPCo is unable to require

WPCo to enter into the agreements adopted by the KPSC. The KPSC's Order provided a framework for the parties to operate the plant through December 2028, but committed matters related to the future of the plant after that date to the good faith commercial negotiations of the parties. The agreements this Commission issued by its Order in this proceeding diverge from the agreements approved by the KPSC primarily, though not exclusively, as they relate to a 2028 valuation of KPCo's share of the Mitchell plant. KPCo has further informed the Companies that this represents a material difference between the two sets of agreements approved by this Commission and the KPSC.

- 7. Because of the conflicting regulatory requirements between this Commission and this KPSC, the Current Mitchell Operating Agreement ("Current Operating Agreement") must remain in effect in accordance with its terms until a resolution of post-2028 operating and ownership issues can be resolved in an acceptable manner, subject to further state commission review. Accordingly, WPCo will move forward in the short term, as it must, under the Current Operating Agreement to achieve the immediate goals needed for WPCo to carry out the ELG project to keep the plant running through 2028 and beyond, and ensure cost allocation is properly performed related to ELG, consistent with this Commission's and the KPSC's orders on the use of the plant for customers.
- 8. In this manner, WPCo will carry out this Commission's directive that it construct the ELG at Mitchell as soon as possible to ensure the plant continues to operate after the date it would otherwise need to cease operations under the federal ELG rules. The Companies recognize that additional regulatory efforts to achieve fully consistent agreements between the parties and jurisdictions at this time could cause delays that would not be in the best interests of the Mitchell Plant. Such further efforts could delay the transfer of permits and operational rights to WPCo

which could, in turn, delay the ELG work, create more outages where the plant will not run, and jeopardize the in-service date for the ELG work. Following the path described above will support the current construction schedule and not cause delays that could require additional plant outages that reduce the availability of the plant.

- Because WPCo is unable to implement changes to the Current Operating Agreement absent a common agreement in both states, and given the looming dates for completing ELG investment, the existing Operating Committee will act as needed to ensure the near-term needs can be accomplished under the Current Operating Agreement. The Operating Committee intends to use resolutions related to plant operations to achieve these near-term needs and customer protections related to plant operations. This near-term path forward is rooted in the points of consistency between the agreements approved by this Commission and the KPSC in their respective orders to ensure plant operations prior to 2028. Both Commissions observed in their respective orders that the Current Operating Agreement could be used to achieve the immediate goals needed to keep the plant running through 2028 and ensure that cost allocation is properly performed under their respective orders on the use of the plant for customers. Both Commissions also supported a core body of concepts related to the operation of the plant prior to December 31, 2028, which could be implemented under the Current Operating Agreement, including updating cost allocation to reflect the commissions' approved cost responsibility for CCR and ELG projects, and to designate WPCo as plant operator.<sup>2</sup>
- 10. The areas of commonality that will be implemented by Operating Committee resolution include: (1) authorizing WPCo to act as the operator of Mitchell, (2) effecting the transfer of ELG and other essential permits from KPCo to WPCo to facilitate timely ELG and

<sup>&</sup>lt;sup>1</sup> See July 1, 2022 Order at 7.

<sup>&</sup>lt;sup>2</sup> See July 1, 2022 Order at 3.

other work, (3) the performance of ELG work at Mitchell by WPCo at its expense, following the direction of this Commission, and (4) the implementation of appropriate accounting procedures to ensure that WPCo and KPCo share in the investments and expenditures needed to operate Mitchell through 2028, including the CCR work, and that the investments and expenditures needed to implement the ELG work that will enable Mitchell to operate beyond 2028 will be assigned to WPCo. Any such resolutions reached by the Operating Committee would be shared with both commissions at the appropriate time.

- and not the ELG project, "[has] the result of Kentucky Power not being permitted to operate Mitchell after 2028." However, KPCo is required to seek future approval of the terms and conditions associated with any disposition of its ownership share in the Mitchell plant based on the facts and circumstances that will emerge in the future. Conversely, this Commission has very clearly directed WPCo to protect its unilateral investment in ELG that enables post 2028 operations and to dispatch the plant should KPCo seek to lower its output. WPCo believes that this Commissions' immediate objectives can be accomplished in the near term through the Operating Committee under the Current Operating Agreement and, in the longer term, by future good faith discussions with KPCo based on the Current Operating Agreement, which must continue in effect until a resolution acceptable to both owners and their respective commissions can be reached.
- 12. The Commission's Order also directed the Companies to file redacted public versions of the Companies' responses to CAD discovery requests 1.4 and 1.5. The Companies have been working on developing the required documents and on July 8, 2022 filed a motion

<sup>&</sup>lt;sup>3</sup> See Electronic Application Of Kentucky Power Company For Approval Of A Certificate Of Public Convenience And Necessity For Environmental Project Construction At The Mitchell Generating Station, An Amended Environmental Compliance Plan, And Revised Environmental Surcharge Tariff Sheets, Case No. 2021-00004, Order on Reh'g at 5 (May 3, 2022).

seeking an extension of the deadline for completing the formidable amount of work needed to accomplish that.

Respectfully submitted,

## APPALACHIAN POWER COMPANY WHEELING POWER COMPANY

By Counsel,

William C. Porth (WV State Bar #2943) wcp@ramlaw.com

Anne C. Blankenship (WV State Bar #9044) acb@ramlaw.com

Robinson & McElwee PLLC

P.O. Box 1791

Charleston, WV 25326

James R. Bacha American Electric Power Service Corporation 1 Riverside Plaza Columbus, Ohio 43215 jrbacha@aep.com

Keith D. Fisher (WV State Bar #11346) American Electric Power Service Corporation 500 Lee Street East, Suite 800 Charleston, West Virginia 25301 (304) 348-4154 kdfisher@aep.com

Counsel for Appalachian Power Company and Wheeling Power Company

Dated: July 11, 2022

#### MITCHELL OPERATING AGREEMENT

# KENTUCKY POWER COMPANY WHEELING POWER COMPANY

and

AMERICAN ELECTRIC POWER SERVICE CORPORATION, AS AGENT

THIS MITCHELL PLANT OPERATING AGREEMENT ("Agreement"), with an effective date of [\_\_] ("Effective Date"), is by and among Kentucky Power Company, a Kentucky corporation qualified as a foreign corporation in West Virginia ("KPCo"), and Wheeling Power Company, a West Virginia corporation ("WPCo") (such two parties hereinafter sometimes referred to as the "Owners"); and American Electric Power Service Corporation, a New York corporation qualified as a foreign corporation in West Virginia ("Agent"). KPCo, WPCo and Agent may hereinafter be referred to as a "Party" or collectively as the "Parties".

#### WITNESSETH:

WHEREAS, KPCo acquired a fifty percent (50%) undivided ownership interest in the Mitchell Power Generation Facility consisting of two 800MW generating units and associated plant, equipment and real estate, located in Moundsville, West Virginia (the "Mitchell Facility") on December 31, 2013; and

WHEREAS, AEP Generation Resources Inc. ("AEPGR"), an affiliate of the Parties, acquired a fifty percent (50%) undivided ownership interest in the Mitchell Facility, also on December 31, 2013; and

WHEREAS, pursuant to an Asset Contribution Agreement between AEPGR and Newco Wheeling Inc., a West Virginia corporation merged or to be merged into WPCo upon the closing of the transactions (the "Transfer Date") set forth in such Asset Contribution Agreement (the "ACA"), AEPGR transferred its fifty percent (50%) undivided interest in the Mitchell Facility to Newco Wheeling Inc., exclusive of its interest in the Conner Run Fly Ash Impoundment and Dam ("Conner Run"), which interest in Conner Run was retained on the Transfer Date by AEPGR; and

WHEREAS, the Agreement shall be effective upon the Effective Date;

WHEREAS, the Owners desire that WPCo shall operate and maintain the Mitchell Facility, exclusive of Conner Run (the "Mitchell Plant"), in accordance with the provisions set forth herein; and

WHEREAS, the Owners are subsidiaries of American Electric Power Company, Inc. ("AEP"), the parent company in an integrated public utility holding company system, and use the services of Agent (an affiliated company engaged solely in the business of furnishing essential services to the Owners and to other affiliated companies), as outlined in the service agreements between Agent and KPCo and between Agent and WPCo.

NOW THEREFORE, in consideration of the premises and for the purposes hereinabove recited, and in consideration of the mutual covenants hereinafter contained, the signatories agree as follows:

#### ARTICLE ONE

#### FUNCTION OF WPCO AND AGENT

- 1.1 WPCo shall operate and maintain the Mitchell Plant in accordance with good utility practice in conformity with the terms and conditions of this Agreement.
- 1.2 WPCo shall keep all necessary books of record, books of account and memoranda of all transactions involving the Mitchell Plant, and shall make computations and allocations on behalf of the Owners, as required under this Agreement. The books of record, books of account and memoranda shall be kept in such manner as to conform, where so required, to the Uniform System of Accounts as prescribed by the Federal Energy Regulatory Commission ("FERC") for Public Utilities and Licensees ("Uniform System of Accounts"), and to the rules and regulations of other regulatory bodies having jurisdiction as they may from time to time be in effect.
- 1.3 The Owners shall establish such bank accounts as may from time to time be required or appropriate.
- 1.4 As soon as practicable after the end of the month, WPCo shall furnish to KPCo a statement setting forth the dollar amounts associated with the operations and maintenance of the Mitchell Plant as allocated hereunder to KPCO and WPCO for such month. The Owners shall, on a timely basis, deposit sufficient dollar amounts in the appropriate bank accounts to cover their respective allocations of such costs.

- 1.5 WPCo shall be responsible for the day to day operation and maintenance of the Mitchell Plant. WPCo shall obtain such materials, labor and other services as it considers necessary in connection with the performance of the functions to be performed by it hereunder from such sources or through such persons as it may designate.
- 1.6 Agent, as directed by the Operating Committee and consistent with Agent's service agreements with KPCo and WPCo, shall provide services necessary for the safe and efficient operation and maintenance of the Mitchell Plant.

#### ARTICLE TWO

#### APPORTIONMENT OF CAPACITY AND ENERGY

- 2.1 The Total Net Capability of the Mitchell Plant at the Mitchell Unit 1 and Unit 2 low-voltage busses, after taking into account auxiliary load demand, is 1,560,000 kilowatts. The Owners may from time to time modify the Total Net Capability of the Mitchell Plant as they may mutually agree.
- 2.2 The Total Net Generation of the Mitchell Plant during a given period, as determined by the requirements of KPCo and WPCo, shall mean the electrical output of the Mitchell Plant generators during such period, measured in kilowatt hours by suitable instruments, reduced by the energy used by auxiliaries for the Mitchell Unit 1 and Unit 2 during such period.
- 2.3 Except as set forth in Section 7.6 (including Section 7.6 Subsections), in any hour, KPCo and WPCo shall share the minimum load responsibility of Mitchell Unit 1 and Unit 2 in respective amounts proportionate to their ownership interests in the Mitchell Plant at such time. Each Owner may independently dispatch its share of the generating capacity between minimum and full load.
- 2.4 In any hour during which the Mitchell Units are out of service, the energy used by the outof-service Units' auxiliaries during such hour shall be provided by KPCo and WPCo in respective amounts proportionate to their ownership interests in the Mitchell Plant at such time.

#### ARTICLE THREE

#### REPLACEMENTS, ADDITIONS, AND RETIREMENTS

- WPCo shall from time to time make or cause to be made any additions to, replacements of, and retirements of, capitalizable facilities associated with the Mitchell Plant in accordance with the approved annual budget.
- 3.2 The dollar amounts associated with any additions to, replacements of, or retirements of, capitalizable facilities associated with the Mitchell Plant shall be allocated to KPCo and WPCo in respective amounts proportionate to their ownership interests in the Mitchell Plant at the time such additions, replacements, or retirements are made.

#### ARTICLE FOUR

#### WORKING CAPITAL REQUIREMENTS

- 4.1 KPCo and WPCo shall periodically mutually determine the amount of funds required for use as working capital in meeting payrolls and other expenses incurred in the operation and maintenance of the Mitchell Plant, and in buying materials and supplies (exclusive of fuel) for the Mitchell Plant.
- 4.2 KPCo and WPCo shall from time to time provide their share of working capital requirements in respective amounts proportionate to their ownership interests at such time in the Mitchell Plant.

#### ARTICLE FIVE

#### INVESTMENT IN FUEL

MPCo and Agent shall establish and maintain reserves of coal in stock piles for the Mitchell Plant of such quality and in such quantities as the Operating Committee shall determine to be required to provide adequate fuel reserves against interruptions of normal fuel supply, provided each Owner, subject to the approval of the Operating Committee and subject to no adverse impact on the operation of the Mitchell Plant, will have the right, but not the obligation, to directly purchase coal, transportation and consumables for its

- ownership interest. For the purposes of this Agreement, "consumables" shall be as defined in FERC account 502.
- 5.2 Except as provided in Section 5.1 for an Owner to elect to procure coal for its own interest, the Owners shall make such monthly investments in the common coal stock piles associated with the Mitchell Plant as are necessary to maintain the number of tons in such coal stock piles, after taking into account the coal consumption from the common coal stock piles by Mitchell Unit 1 and Unit 2 during such month.
- 5.3 At any time, KPCo's and WPCo's respective shares of the investment in the common coal stock piles shall be proportionate to their ownership interests in the Mitchell Plant, unless an Owner elects to procure its own coal as provided in Section 5.1, in which case inventories will be separately maintained for accounting purposes.
- 5.4 Fuel oil and consumables charged to operation for the Mitchell Plant shall be owned and accounted for between the Owners in the same manner as coal.

#### ARTICLE SIX

#### APPORTIONMENT OF STATION COSTS

- 6.1 Except in the case where an Owner has elected to purchase coal for its own interest as provided for in Section 5.1 (in which case the allocation to the Owners of fuel expense shall be in accordance with procedures and processes approved by the Operating Committee), the allocation to the Owners of fuel expense associated with Mitchell Unit 1 and Unit 2 shall be determined by WPCo and Agent as follows:
  - (a) In any calendar month, the average unit cost of coal available for consumption from the Mitchell Plant common coal stock piles shall be determined based on the prior month's ending inventory dollar and ton balances plus current month receipts delivered to the Mitchell Plant common coal stock piles. Each Owner's average unit cost will be the same, and receipts and inventory available for consumption amounts will be allocated to each Owner based on monthly usage.

- (b) The number of tons of coal consumed by the Mitchell Plant in each calendar month from the Mitchell Plant common coal stock piles shall be determined and shall be converted into a dollar amount equal to the product of (i) the average cost per ton of coal associated with the Mitchell Plant in the Mitchell Plant common coal stock pile at the close of such month, and (ii) the number of tons of coal consumed by the Mitchell Plant from the Mitchell Plant common coal stock piles during such month. Such dollar amount shall be credited to the Mitchell Plant fuel in stock pile and charged to Mitchell Plant fuel consumed.
- (c) In each calendar month, KPCo's and WPCo's respective shares of the Mitchell Plant fuel consumed expense as determined by the provisions of Section 6.1(b) shall be proportionate to each Owner's dispatch of the Mitchell Plant in such month.
- (d) Fuel oil reserves will be owned and accounted for in the same manner as coal stock piles, and fuel oil consumed will be allocated to the Owners in the same manner as coal consumed.
- 6.2 Except for adjustments related to assignment of facilities necessary to maintain capacity and energy production that are directly assigned to one owner pursuant to Section 7.9, for purposes of this Agreement, KPCo's Assigned Capacity in the Mitchell Plant shall be equal to 50% of the Total Net Capability, and WPCo's Assigned Capacity shall be equal to 50% of the Total Net Capability.
- 6.3 For each calendar month WPCo and Agent will, to the extent practicable, determine all Mitchell Plant operations expenses and associated overheads, as accounted for under the FERC Uniform System of Accounts.
- 6.4 For each calendar month WPCo and Agent will, to the extent practicable, determine all Mitchell Plant maintenance expenses and associated overheads, as accounted for under the FERC Uniform System of Accounts.

- 6.5 In each calendar month, KPCo's and WPCo's respective shares of operations and maintenance expenses associated with the Mitchell Plant, as determined in accordance with Sections 6.3 and 6.4, shall be allocated as follows:
  - (a) In each calendar month, KPCo's and WPCo's respective shares of the Mitchell Plant steam expenses as recorded in FERC Account 502, and emission tons, with allowance expenses as recorded in FERC Account 509, shall be proportionate to each Owner's dispatch of the Mitchell Plant in such month.
  - (b) In each calendar month, the maintenance of boiler plant expenses as recorded in FERC Account 512, and maintenance of electric plant expenses as recorded in FERC Account 513, shall be directly assigned to Mitchell Unit 1 or Unit 2 or designated as a common expense attributable to both units. In each calendar month, KPCo's and WPCo's respective shares of these expenses shall be proportionate to each Owner's capacity entitlement as adjusted pursuant to Section 7.9.
  - (c) On or prior to December 31, 2028, in each calendar month, KPCo's and WPCo's respective shares of all other operations, maintenance, administrative and general expenses shall be proportionate to their respective ownership interests. After December 31, 2028, these expenses will be assigned proportionate to each Owner's capacity entitlement as adjusted pursuant to Section 7.9.
- 6.6 Each Owner shall bear the cost of all taxes attributable to its capacity entitlement as adjusted pursuant to Section 7.9.

#### ARTICLE SEVEN

#### OPERATING COMMITTEE AND OPERATIONS

7.1 By written notice to each other, the Owners and Agent each shall name one representative ("Operating Representative") and one alternate to act for it in matters pertaining to operating arrangements under this Agreement. Any Party may change its Operating

Representative or alternate at any time by written notice to the other Parties. The Operating Representatives for the respective Parties, or their alternates, shall comprise the Operating Committee. All decisions, directives, or other actions by the Operating Committee must be by unanimous agreement of the Operating Representatives of the Owners. The Operating Representative of Agent, or of any third party that provides services in replacement of Agent, shall be free to express the views of Agent or such third party on any matter, but shall not have a vote on the Operating Committee. Except as otherwise provided in Sections 11.1, 11.2 and 11.3 with respect to a dispute referred to the Operating Committee by an Owner, the failure of the Owners' respective Operating Representatives to unanimously agree with respect to a matter pending before the Operating Committee shall not be considered to be a dispute that would be subject to resolution under Article Eleven.

- 7.2 The Operating Committee shall have the following responsibilities:
  - (a) Review and approval of an annual budget and annual operating plan, including determination of the emission allowances required to be acquired by KPCo and WPCo. If the Operating Committee fails to approve an annual budget, the approved annual budget from the previous year will continue to apply until such time as the new annual budget is approved.
  - (b) Establishment and review of procedures and systems for dispatch, notification of dispatch, and unit commitment under this Agreement, including any commitment of Called Capacity pursuant to Section 7.6.2.
  - (c) Establishment and monitoring of procedures for communication and coordination with respect to the Mitchell Plant capacity availability, fuelfiring options, and scheduling of outages for maintenance, repairs, equipment replacements, scheduled inspections, and other foreseeable cause of outages, as well as the return to availability following an unplanned outage.
  - (d) Decisions on capital expenditures, including unit upgrades and re-powering.

- (e) Determinations as to changes in the unit capability and decisions on unit retirement.
- (f) Establishment and modification of billing procedures under this Agreement.
- (g) Approval of material contracts for fuel, transportation or consumable supply. Establishment of specification of fuels, oversight of fuel inspection and certification procedures, management of fuel inventories, and allocation of rights under fuel supply, transportation and consumable contracts. Establishment of an Owner's procurement rights and procedures if the Owner elects to purchase coal, transportation or consumables for its own interest.
- (h) Establishment of, termination of, and approval of any change or amendment to the operating arrangements between KPCo and Agent or any replacement third party with respect to the Mitchell Plant generating units; provided, however, that Agent or any replacement third party shall participate in discussions pursuant to this subsection 7.2(h) only if and to the extent requested to do so by both Owners.
- (i) Review and approval of plans and procedures designed to ensure compliance with any environmental law, regulation, ordinance or permit, including procedures for allocating and using emission allowances or for any programs that permit averaging at more than one unit for compliance.
- (j) Other duties as assigned by agreement of the Owners.
- 7.3 The Operating Committee shall meet at least annually, and at such other times as any Party may reasonably request.
- 7.4 The Parties shall cooperate in providing to the Operating Committee the information it reasonably needs to carry out its duties, and to supplement or correct such information on a timely basis.
- 7.5 The Owners will each make an initial unit commitment one business day ahead of real-time dispatch.

- 7.6 Application of this Section 7.6 (including subsections) is subject to (i) the receipt of any necessary regulatory approvals or waivers expressly granted for this Section 7.6; and (ii) the Operating Committee establishing and approving procedures and systems for dispatch. As used in this Section and subsections of this Section, the terms "Party" or "Parties" refers only to KPCo and WPCo, or both of them, as the case may be.
- 7.6.1 If Mitchell Unit 1 or Unit 2 is designated to be committed by both Parties, such unit will be brought on line or kept on line. If neither Party designates Mitchell Unit 1 or Unit 2 to be committed, such unit will remain off line or be taken offline.
- 7.6.2 When a Mitchell Unit is designated to be committed by one Party, but designated not to be committed by the other Party, the unit will be brought on line or kept on line if the Party designating the unit for commitment undertakes to pay any applicable start-up costs for the unit, as well as any applicable minimum running costs for the unit thereafter, in which event the unit shall be brought on line or kept on line, as the case may be. The Party so designating the unit to be committed shall have the right to schedule and dispatch up to all of the Available Capacity of the unit. Available Capacity means that portion of the Owners' aggregate Assigned Capacity that is currently capable of being dispatched. The Party exercising this right shall be referred to as the "Calling Party," and the capacity called by that Party in excess of its Assigned Capacity Percentage of the Available Capacity of that unit shall be referred to as its "Called Capacity." The other Party shall be referred to as the "Non-Calling Party". The Calling Party shall provide reasonable notice to the Non-Calling Party of its call, including any start-up or shut-down time for the Unit. For purposes of this Agreement, KPCo's Assigned Capacity Percentage shall be 50%, and WPCo's Assigned Capacity Percentage shall be 50% as adjusted pursuant to section 7.9.
- 7.6.3 The Non-Calling Party can reclaim any Called Capacity attributable to its Assigned Capacity share by giving the Calling Party notice equal to the normal cold start-up time for the unit. At the end of the notice period, the Non-Calling Party shall have the right to schedule and dispatch the recalled capacity. At that point, the Non-Calling Party shall resume its responsibility for its share of any applicable start-up costs for the unit and

- prospectively shall bear its responsibility for the costs associated with its Assigned Capacity from the unit.
- 7.6.4 If any capacity remains available but is not dispatched from a Party's Available Capacity committed as a result of the initial unit commitment, the other Party may only schedule and dispatch such capacity pursuant to 7.6.1 through 7.6.3, above.
- 7.7 KPCo and WPCo shall be individually responsible for any fees charged by FERC on the basis of the sales or transmission by each of capacity or energy at wholesale in interstate commerce.
- 7.8 Emission Allowances. On the Transfer Date pursuant to the ACA, AEPGR, the previous owner of WPCo's interest in the Mitchell Plant, will assign to WPCo all Emission Allowances allocated to AEPGR for the Mitchell Plant for each vintage year after 2014, issued by the U.S. Environmental Protection Agency ("USEPA") pursuant to Title IV of the Clean Air Act Amendments of 1990 and any regulations thereunder, and any other emission allowance trading program created under the Clean Air Act and administered by USEPA or the State of West Virginia, including but not limited to the Clean Air Interstate Rule 40 CFR Parts 96 and 97, and any amendments thereto ("Emission Allowances"), and all Emission Allowances for 2014 and any vintage year prior to 2014 that were allocated to the Mitchell Plant and that have not been expended as of the date of assignment. To the extent that additional Emission Allowances are required for operation of the Mitchell Plant, KPCo and WPCo will each be responsible for acquiring sufficient Emission Allowances to satisfy the Emission Allowances required because of its dispatch of energy from the Mitchell Plant, and the Emission Allowances required to satisfy the Emission Allowance surrender obligations attributable to the Mitchell Plant imposed under the Consent Decree between USEPA and Ohio Power Company entered on December 10, 2007, in Civil Action No. C2-99-1182 and consolidated cases by the U.S. District Court in the Southern District of Ohio. On or before January 10 of each year, Agent shall determine and notify KPCo and WPCo of the number of additional annual Emission Allowances consumed by each of them through December 31 of the previous year, and KPCo and WPCo shall each transfer into the Mitchell Plant U.S. EPA Allowance Transfer System account that number of

Emission Allowances with a small compliance margin by January 31 of that year. For seasonal Emission Allowance programs, Agent shall determine and notify KPCo and WPCo of the number of additional seasonal Emission Allowances consumed by each of them during the applicable compliance period by the 10th day of the first month following the end of the compliance period, and KPCo and WPCo shall each transfer into the appropriate Mitchell Plant U.S. EPA Allowance Transfer System Account that number of Emission Allowances with a small compliance margin by the last day of the first month following the end of the compliance period. In the event that KPCo or WPCo fails to surrender the required number of Emission Allowances by January 31 or the last day of the first month following any seasonal compliance period, Agent shall purchase the required number of Emission Allowances, and KPCo or WPCo, as the case may be, shall reimburse Agent for such purchases, with interest at the Federal Funds Rate (as published by the Board of Governors of the Federal Reserve System as from time to time in effect) running from the date of such purchases to the date of payment. The Operating Committee will develop procedures to be implemented after the end of each calendar year to account for the Emission Allowances required by the use of the Mitchell Plant by KPCo and WPCo and to correct any imbalance between Emission Allowances supplied and Emission Allowances used through the end of the preceding year by settlement or payment.

- 7.9 Capital repairs and improvements to the Mitchell Plant will be determined by the Operating Committee pursuant to the annual budgeting process set forth in Section 7.10. Expenditures that the Operating Committee determines have been or will be incurred exclusively for one Owner shall be assigned exclusively to that Owner. To the extent that capacity and energy production would not be available but for the investments and costs assigned exclusively to one owner, such capacity and energy production shall be assigned solely to the owner to which the costs are assigned.
- 7.10 At least 90 days before the start of each operating year, WPCo and Agent shall submit to the Operating Committee a proposed annual budget with respect to the Mitchell Plant, a proposed annual operating plan, and an estimate and schedule of costs to be incurred for major maintenance or replacement items during the next six-year period. The annual

budget shall be presented on a month-by-month basis for each month during the next operating year, and shall include an operating budget, a capital budget, an estimate of the cost of any major repairs that are anticipated will occur during such operating year with respect to the Mitchell Plant, and an itemized estimate of all projected non-fuel variable operating expenses relating to the operation of the Mitchell Plant during that operating year. The members of the Operating Committee will meet and work in good faith to agree upon the final annual budget and final annual operating plan. Once approved, the annual budget and annual operating plan shall remain in effect throughout the applicable operating year, subject to such changes, revisions, amendments, and updating as the Operating Committee may determine.

#### ARTICLE EIGHT

#### EFFECTIVE DATE AND TERM

- 8.1 Subject to receipt of necessary regulatory approvals, the effective date of this Agreement shall be the Effective Date.
- 8.2 Subject to necessary regulatory approvals, this Agreement shall remain in force until such time as (i) KPCo or WPCo has divested itself of all or any portion of its ownership interest in the Mitchell Plant, other than assignment or other transfer of such ownership interests to another AEP affiliate; or (ii) either KPCo or WPCo is no longer a direct or indirect wholly owned subsidiary of AEP; or (iii) KPCo and WPCo may mutually agree to terminate this Agreement.

#### ARTICLE NINE

#### **GENERAL**

9.1 This Agreement shall inure to the benefit of and be binding upon the signatories hereto and their respective successors and assigns, but this Agreement may not be assigned by any signatory without the written consent of the others, which consent shall not be unreasonably withheld.

- 9.2 This Agreement is subject to the regulatory authority of any State or Federal agency having jurisdiction.
- 9.3 The interpretation and performance of this Agreement shall be in accordance with the laws of the State of West Virginia, excluding conflict of laws principles that would require the application of the laws of a different jurisdiction.
- 9.4 This Agreement supersedes all previous representations, understandings, negotiations, and agreements, either written or oral between the signatories or their representatives with respect to operation of the Mitchell Plant, and constitutes the entire agreement of the signatories with respect to the operation of the Plant. Notwithstanding the foregoing, this Agreement does not supersede any previous agreements among any of the signatories allocating or transferring rights to capacity and associated energy, or ownership, of the Mitchell Plant.
- 9.5 Each Party shall designate in writing a representative to receive any and all notices required under this Agreement. Notices shall be in writing and shall be given to the representative designated to receive them, either by personal delivery, certified mail, facsimile, e-mail or any similar means, properly addressed to such representative at the address specified below:

KENTUCKY POWER COMPANY		
Attn:		
Phone:		
Facsimile:		
E-Mail:		

WHEELING POWER COM	PANY
A	
Attn:	
Phone:	
Facsimile:	
E-Mail:	
AMERICAN ELECTRIC PO	WER
SERVICE CORPORATION	
Attn:	
W- 4	
Phone:	
Phone:Facsimile:	

All notices shall be effective upon receipt, or upon such later date following receipt as set forth in the notice. Any Party may, by written notice to the other Parties, change the representative or the address to which such notices are to be sent.

#### ARTICLE TEN

#### LIMITATION OF LIABILITY

10.1 Notwithstanding anything in this Agreement to the contrary, neither of the Owners or Agent shall be liable under this Agreement for special, consequential, indirect, punitive or exemplary damages, or for lost profits or business interruption damages, whether arising by statute, in tort or contract or otherwise.

#### ARTICLE ELEVEN

#### DISPUTE RESOLUTION

- 11.1 If either Owner believes that a dispute has arisen as to the meaning or application of this Agreement, it shall submit a written description of the disputed matter to the Operating Committee, and shall provide a copy of that submission to the other Owner.
- 11.2 If the Operating Committee does not reach agreement on the resolution of a dispute within thirty (30) days after the dispute is presented to it, the matter shall be referred to senior executive officers with the authority to resolve such dispute of each of the Owners for resolution in the manner that such individuals shall agree is appropriate; provided, <a href="https://docs.nih.gov/however">however</a>, that either Owner may exercise any and all rights at law or equity at any time after the end of the thirty (30) day period provided for the Operating Committee to reach agreement if the Operating Committee has not reached agreement.
- 11.3 Except as provided in this Article Eleven, the existence, contents, or results of any settlement negotiations or the results thereof under this Article Eleven may not be disclosed without the prior written consent of the Owners, provided, however, that either Owner may make disclosures as may be required to fulfill regulatory obligations to any governmental authority having jurisdiction, and may inform its lenders, affiliates, auditors, and insurers, as necessary, under pledge of confidentiality, and may consult with expert consultants as required in connection with any proceeding under pledge of confidentiality.
- Nothing in this Agreement shall be construed to preclude either Owner from filing a petition or complaint with FERC with respect to any claim over which FERC has jurisdiction. In such case, the other Owner may request that FERC reject the petition or complaint or otherwise decline to exercise its jurisdiction. If FERC declines to act with respect to all or part of a claim, the portion of the claim not so accepted by FERC may be resolved through an action at law or equity. To the extent that FERC asserts or accepts jurisdiction over all or part of a claim, the decisions, findings of fact, or orders of FERC shall be final and binding, subject to judicial review under the Federal Power Act, 16 U.S.C. §§ 791a et seq., as amended from time to time, and any proceedings that may have commenced prior to the assertion or acceptance of jurisdiction by FERC shall be stayed,

pending the outcome of FERC proceedings. To the extent that any decisions, findings of fact, or orders of FERC do not provide a final or complete remedy to an Owner seeking relief, such Owner may proceed at law or equity to secure such a remedy, subject to any FERC decisions, findings, or orders.

11.5 If an Owner (the "Contesting Owner") contests in good faith any amount paid pursuant to the terms of this Agreement, and any portion of such amount is determined or resolved (including pursuant to the dispute resolution procedures of this Article Eleven) to be in excess of the actual amount due pursuant to the terms of this Agreement, then the Contesting Owner may charge the other Owner interest with respect to such excess amount from the day the payment was made until it is repaid to the Contesting Owner, at the rate equal to the prime rate as published from time to time *in The Wall Street Journal* (or any successor publication) plus five (5) percentage points per annum, calculated daily, regardless of whether the Contesting Owner has notified the other Owner in advance of its intention to charge interest with respect to such excess amount, and the other Owner shall make payment in full in respect of such excess amount and interest within thirty (30) days of written demand therefor.

IN WITNESS WHEREOF, the Parties hereto have caused this Agreement to be executed by their officers thereunto duly authorized as of the date first above written.

KENTUCKY POWER COMPANY
By:
Title:
WHEELING POWER COMPANY
By:
Title:
AMERICAN ELECTRIC POWER SERVICE
CORPORATION
Ву:
Title:

#### MITCHELL OPERATING AGREEMENT

KENTUCKY POWER COMPANY
WHEELING POWER COMPANY

and

AMERICAN ELECTRIC POWER SERVICE CORPORATION, AS AGENT

THIS MITCHELL PLANT OPERATING AGREEMENT ("Agreement"), with an effective date of [\_\_] ("Effective Date"), is by and among Kentucky Power Company, a Kentucky corporation qualified as a foreign corporation in West Virginia ("KPCo"), and Wheeling Power Company, a West Virginia corporation ("WPCo") (such two parties hereinafter sometimes referred to as the "Owners"); and American Electric Power Service Corporation, a New York corporation qualified as a foreign corporation in West Virginia ("Agent"). KPCo, WPCo and Agent may hereinafter be referred to as a "Party" or collectively as the "Parties".

#### WITNESSETH:

WHEREAS, KPCo acquired a fifty percent (50%) undivided ownership interest in the Mitchell Power Generation Facility consisting of two 800MW generating units and associated plant, equipment and real estate, located in Moundsville, West Virginia (the "Mitchell Facility") on December 31, 2013; and

WHEREAS, AEP Generation Resources Inc. ("AEPGR"), an affiliate of the Parties, acquired a fifty percent (50%) undivided ownership interest in the Mitchell Facility, also on December 31, 2013; and

WHEREAS, pursuant to an Asset Contribution Agreement between AEPGR and Newco Wheeling Inc., a West Virginia corporation merged or to be merged into WPCo upon the closing of the transactions (the "Transfer Date") set forth in such Asset Contribution Agreement (the "ACA"), AEPGR transferred its fifty percent (50%) undivided interest in the Mitchell Facility to Newco Wheeling Inc., exclusive of its interest in the Conner Run Fly Ash Impoundment and Dam ("Conner Run"), which interest in Conner Run was retained on the Transfer Date by AEPGR; and

WHEREAS, the Agreement shall be effective upon the Effective Date;

WHEREAS, the Owners desire that WPCo shall operate and maintain the Mitchell Facility, exclusive of Conner Run (the "Mitchell Plant"), in accordance with the provisions set forth herein; and

WHEREAS, the Owners are subsidiaries of American Electric Power Company, Inc. ("AEP"), the parent company in an integrated public utility holding company system, and use the services of Agent (an affiliated company engaged solely in the business of furnishing essential services to the Owners and to other affiliated companies), as outlined in the service agreements between Agent and KPCo and between Agent and WPCo.

NOW THEREFORE, in consideration of the premises and for the purposes hereinabove recited, and in consideration of the mutual covenants hereinafter contained, the signatories agree as follows:

# ARTICLE ONE

# FUNCTION OF WPCO AND AGENT

- 1.1 WPCo shall operate and maintain the Mitchell Plant in accordance with good utility practice in conformity with the terms and conditions of this Agreement.
- 1.2 WPCo shall keep all necessary books of record, books of account and memoranda of all transactions involving the Mitchell Plant, and shall make computations and allocations on behalf of the Owners, as required under this Agreement. The books of record, books of account and memoranda shall be kept in such manner as to conform, where so required, to the Uniform System of Accounts as prescribed by the Federal Energy Regulatory Commission ("FERC") for Public Utilities and Licensees ("Uniform System of Accounts"), and to the rules and regulations of other regulatory bodies having jurisdiction as they may from time to time be in effect.
- 1.3 The Owners shall establish such bank accounts as may from time to time be required or appropriate.
- 1.4 As soon as practicable after the end of the month, WPCo shall furnish to KPCo a statement setting forth the dollar amounts associated with the operations and maintenance of the Mitchell Plant as allocated hereunder to KPCO and WPCO for such month. The

- Owners shall, on a timely basis, deposit sufficient dollar amounts in the appropriate bank accounts to cover their respective allocations of such costs.
- 1.5 WPCo shall be responsible for the day to day operation and maintenance of the Mitchell Plant. WPCo shall obtain such materials, labor and other services as it considers necessary in connection with the performance of the functions to be performed by it hereunder from such sources or through such persons as it may designate.
- 1.6 Agent, as directed by the Operating Committee and consistent with Agent's service agreements with KPCo and WPCo, shall provide services necessary for the safe and efficient operation and maintenance of the Mitchell Plant.

# ARTICLE TWO

# APPORTIONMENT OF CAPACITY AND ENERGY

- 2.1 The Total Net Capability of the Mitchell Plant at the Mitchell Unit 1 and Unit 2 low-voltage busses, after taking into account auxiliary load demand, is 1,560,000 kilowatts. The Owners may from time to time modify the Total Net Capability of the Mitchell Plant as they may mutually agree.
- 2.2 The Total Net Generation of the Mitchell Plant during a given period, as determined by the requirements of KPCo and WPCo, shall mean the electrical output of the Mitchell Plant generators during such period, measured in kilowatt hours by suitable instruments, reduced by the energy used by auxiliaries for the Mitchell Unit 1 and Unit 2 during such period.
- 2.3 Except as set forth in Section 7.6 (including Section 7.6 Subsections), in any hour, KPCo and WPCo shall share the minimum load responsibility of Mitchell Unit 1 and Unit 2 in respective amounts proportionate to their ownership interests in the Mitchell Plant at such time. Each Owner may independently dispatch its share of the generating capacity between minimum and full load.
- 2.4 In any hour during which the Mitchell Units are out of service, the energy used by the out-of-service Units' auxiliaries during such hour shall be provided by KPCo and WPCo

in respective amounts proportionate to their ownership interests in the Mitchell Plant at such time.

# ARTICLE THREE

# REPLACEMENTS, ADDITIONS, AND RETIREMENTS

- 3.1 WPCo shall from time to time make or cause to be made any additions to, replacements of, and retirements of, capitalizable facilities associated with the Mitchell Plant in accordance with the approved annual budget.
- 3.2 The dollar amounts associated with any additions to, replacements of, or retirements of, capitalizable facilities associated with the Mitchell Plant shall be allocated to KPCo and WPCo in respective amounts proportionate to their ownership interests in the Mitchell Plant at the time such additions, replacements, or retirements are made.

# ARTICLE FOUR

# WORKING CAPITAL REQUIREMENTS

- 4.1 KPCo and WPCo shall periodically mutually determine the amount of funds required for use as working capital in meeting payrolls and other expenses incurred in the operation and maintenance of the Mitchell Plant, and in buying materials and supplies (exclusive of fuel) for the Mitchell Plant.
- 4.2 KPCo and WPCo shall from time to time provide their share of working capital requirements in respective amounts proportionate to their ownership interests at such time in the Mitchell Plant.

# ARTICLE FIVE

# INVESTMENT IN FUEL

5.1 WPCo and Agent shall establish and maintain reserves of coal in stock piles for the Mitchell Plant of such quality and in such quantities as the Operating Committee shall determine to be required to provide adequate fuel reserves against interruptions of normal fuel supply, provided each Owner, subject to the approval of the Operating Committee and subject to no adverse impact on the operation of the Mitchell Plant, will have the

- right, but not the obligation, to directly purchase coal, transportation and consumables for its ownership interest. For the purposes of this Agreement, "consumables" shall be as defined in FERC account 502.
- 5.2 Except as provided in Section 5.1 for an Owner to elect to procure coal for its own interest, the Owners shall make such monthly investments in the common coal stock piles associated with the Mitchell Plant as are necessary to maintain the number of tons in such coal stock piles, after taking into account the coal consumption from the common coal stock piles by Mitchell Unit 1 and Unit 2 during such month.
- 5.3 At any time, KPCo's and WPCo's respective shares of the investment in the common coal stock piles shall be proportionate to their ownership interests in the Mitchell Plant, unless an Owner elects to procure its own coal as provided in Section 5.1, in which case inventories will be separately maintained for accounting purposes.
- Fuel oil and consumables charged to operation for the Mitchell Plant shall be owned and accounted for between the Owners in the same manner as coal.

# ARTICLE SIX

# APPORTIONMENT OF STATION COSTS

- 6.1 Except in the case where an Owner has elected to purchase coal for its own interest as provided for in Section 5.1 (in which case the allocation to the Owners of fuel expense shall be in accordance with procedures and processes approved by the Operating Committee), the allocation to the Owners of fuel expense associated with Mitchell Unit 1 and Unit 2 shall be determined by WPCo and Agent as follows:
  - (a) In any calendar month, the average unit cost of coal available for consumption from the Mitchell Plant common coal stock piles shall be determined based on the prior month's ending inventory dollar and ton balances plus current month receipts delivered to the Mitchell Plant common coal stock piles. Each Owner's average unit cost will be the same, and receipts and inventory available for consumption amounts will be allocated to each Owner based on monthly usage.

- (b) The number of tons of coal consumed by the Mitchell Plant in each calendar month from the Mitchell Plant common coal stock piles shall be determined and shall be converted into a dollar amount equal to the product of (i) the average cost per ton of coal associated with the Mitchell Plant in the Mitchell Plant common coal stock pile at the close of such month, and (ii) the number of tons of coal consumed by the Mitchell Plant from the Mitchell Plant common coal stock piles during such month. Such dollar amount shall be credited to the Mitchell Plant fuel in stock pile and charged to Mitchell Plant fuel consumed.
- (c) In each calendar month, KPCo's and WPCo's respective shares of the Mitchell Plant fuel consumed expense as determined by the provisions of Section 6.1(b) shall be proportionate to each Owner's dispatch of the Mitchell Plant in such month.
- (d) Fuel oil reserves will be owned and accounted for in the same manner as coal stock piles, and fuel oil consumed will be allocated to the Owners in the same manner as coal consumed.
- 6.2 Except for adjustments related to assignment of facilities necessary to maintain capacity and energy production that are directly assigned to one owner pursuant to Section 7.9, for purposes of this Agreement, KPCo's Assigned Capacity in the Mitchell Plant shall be equal to 50% of the Total Net Capability, and WPCo's Assigned Capacity shall be equal to 50% of the Total Net Capability.
- 6.3 For each calendar month WPCo and Agent will, to the extent practicable, determine all Mitchell Plant operations expenses and associated overheads, as accounted for under the FERC Uniform System of Accounts.
- 6.4 For each calendar month WPCo and Agent will, to the extent practicable, determine all Mitchell Plant maintenance expenses and associated overheads, as accounted for under the FERC Uniform System of Accounts.

- 6.5 In each calendar month, KPCo's and WPCo's respective shares of operations and maintenance expenses associated with the Mitchell Plant, as determined in accordance with Sections 6.3 and 6.4, shall be allocated as follows:
  - (a) In each calendar month, KPCo's and WPCo's respective shares of the Mitchell Plant steam expenses as recorded in FERC Account 502, and emission tons, with allowance expenses as recorded in FERC Account 509, shall be proportionate to each Owner's dispatch of the Mitchell Plant in such month.
  - (b) In each calendar month, the maintenance of boiler plant expenses as recorded in FERC Account 512, and maintenance of electric plant expenses as recorded in FERC Account 513, shall be directly assigned to Mitchell Unit 1 or Unit 2 or designated as a common expense attributable to both units. In each calendar month, KPCo's and WPCo's respective shares of these expenses shall be proportionate to each Owner's capacity entitlement as adjusted pursuant to Section 7.9.
  - (c) PriorOn or prior to (date when generation must cease but for ELG and other WPCo investments)December 31, 2028, in each calendar month, KPCo's and WPCo's respective shares of all other operations, maintenance, administrative and general expenses shall be proportionate to their respective ownership interests. After (date)December 31, 2028, these expenses will be assigned proportionate to each Owner's capacity entitlement as adjusted pursuant to Section 7.9.
- 6.6 Each Owner shall bear the cost of all taxes attributable to its capacity entitlement as adjusted pursuant to Section 7.9.

# ARTICLE SEVEN

# OPERATING COMMITTEE AND OPERATIONS

7.1 By written notice to each other, the Owners and Agent each shall name one representative ("Operating Representative") and one alternate to act for it in matters pertaining to

operating arrangements under this Agreement. Any Party may change its Operating Representative or alternate at any time by written notice to the other Parties. The Operating Representatives for the respective Parties, or their alternates, shall comprise the Operating Committee. All decisions, directives, or other actions by the Operating Committee must be by unanimous agreement of the Operating Representatives of the Owners. The Operating Representative of Agent, or of any third party that provides services in replacement of Agent, shall be free to express the views of Agent or such third party on any matter, but shall not have a vote on the Operating Committee. Except as otherwise provided in Sections 11.1, 11.2 and 11.3 with respect to a dispute referred to the Operating Committee by an Owner, the failure of the Owners' respective Operating Representatives to unanimously agree with respect to a matter pending before the Operating Committee shall not be considered to be a dispute that would be subject to resolution under Article Eleven.

# 7.2 The Operating Committee shall have the following responsibilities:

- (a) Review and approval of an annual budget and annual operating plan, including determination of the emission allowances required to be acquired by KPCo and WPCo. If the Operating Committee fails to approve an annual budget, the approved annual budget from the previous year will continue to apply until such time as the new annual budget is approved.
- (b) Establishment and review of procedures and systems for dispatch, notification of dispatch, and unit commitment under this Agreement, including any commitment of Called Capacity pursuant to Section 7.6.2.
- (c) Establishment and monitoring of procedures for communication and coordination with respect to the Mitchell Plant capacity availability, fuel-firing options, and scheduling of outages for maintenance, repairs, equipment replacements, scheduled inspections, and other foreseeable cause of outages, as well as the return to availability following an unplanned outage.

- (d) Decisions on capital expenditures, including unit upgrades and re-powering.
- (e) Determinations as to changes in the unit capability and decisions on unit retirement.
- (f) Establishment and modification of billing procedures under this Agreement.
- (g) Approval of material contracts for fuel, transportation or consumable supply. Establishment of specification of fuels, oversight of fuel inspection and certification procedures, management of fuel inventories, and allocation of rights under fuel supply, transportation and consumable contracts. Establishment of an Owner's procurement rights and procedures if the Owner elects to purchase coal, transportation or consumables for its own interest.
- (h) Establishment of, termination of, and approval of any change or amendment to the operating arrangements between KPCo and Agent or any replacement third party with respect to the Mitchell Plant generating units; provided, however, that Agent or any replacement third party shall participate in discussions pursuant to this subsection 7.2(h) only if and to the extent requested to do so by both Owners.
- (i) Review and approval of plans and procedures designed to ensure compliance with any environmental law, regulation, ordinance or permit, including procedures for allocating and using emission allowances or for any programs that permit averaging at more than one unit for compliance.
- (j) Other duties as assigned by agreement of the Owners.
- 7.3 The Operating Committee shall meet at least annually, and at such other times as any Party may reasonably request.
- 7.4 The Parties shall cooperate in providing to the Operating Committee the information it reasonably needs to carry out its duties, and to supplement or correct such information on a timely basis.

- 7.5 The Owners will each make an initial unit commitment one business day ahead of real-time dispatch.
- 7.6 Application of this Section 7.6 (including subsections) is subject to (i) the receipt of any necessary regulatory approvals or waivers expressly granted for this Section 7.6; and (ii) the Operating Committee establishing and approving procedures and systems for dispatch. As used in this Section and subsections of this Section, the terms "Party" or "Parties" refers only to KPCo and WPCo, or both of them, as the case may be.
- 7.6.1 If Mitchell Unit 1 or Unit 2 is designated to be committed by both Parties, such unit will be brought on line or kept on line. If neither Party designates Mitchell Unit 1 or Unit 2 to be committed, such unit will remain off line or be taken offline.
- 7.6.2 When a Mitchell Unit is designated to be committed by one Party, but designated not to be committed by the other Party, the unit will be brought on line or kept on line if the Party designating the unit for commitment undertakes to pay any applicable start-up costs for the unit, as well as any applicable minimum running costs for the unit thereafter, in which event the unit shall be brought on line or kept on line, as the case may be. The Party so designating the unit to be committed shall have the right to schedule and dispatch up to all of the Available Capacity of the unit. Available Capacity means that portion of the Owners' aggregate Assigned Capacity that is currently capable of being dispatched. The Party exercising this right shall be referred to as the "Calling Party," and the capacity called by that Party in excess of its Assigned Capacity Percentage of the Available Capacity of that unit shall be referred to as its "Called Capacity." The other Party shall be referred to as the "Non-Calling Party". The Calling Party shall provide reasonable notice to the Non-Calling Party of its call, including any start-up or shut-down time for the Unit. For purposes of this Agreement, KPCo's Assigned Capacity Percentage shall be 50%, and WPCo's Assigned Capacity Percentage shall be 50% as adjusted pursuant to section 7.9.
- 7.6.3 The Non-Calling Party can reclaim any Called Capacity attributable to its Assigned Capacity share by giving the Calling Party notice equal to the normal cold start-up time for the unit. At the end of the notice period, the Non-Calling Party shall have the right to

- schedule and dispatch the recalled capacity. At that point, the Non-Calling Party shall resume its responsibility for its share of any applicable start-up costs for the unit and prospectively shall bear its responsibility for the costs associated with its Assigned Capacity from the unit.
- 7.6.4 If any capacity remains available but is not dispatched from a Party's Available Capacity committed as a result of the initial unit commitment, the other Party may only schedule and dispatch such capacity pursuant to 7.6.1 through 7.6.3, above.
- 7.7 KPCo and WPCo shall be individually responsible for any fees charged by FERC on the basis of the sales or transmission by each of capacity or energy at wholesale in interstate commerce.
- 7.8 Emission Allowances. On the Transfer Date pursuant to the ACA, AEPGR, the previous owner of WPCo's interest in the Mitchell Plant, will assign to WPCo all Emission Allowances allocated to AEPGR for the Mitchell Plant for each vintage year after 2014, issued by the U.S. Environmental Protection Agency ("USEPA") pursuant to Title IV of the Clean Air Act Amendments of 1990 and any regulations thereunder, and any other emission allowance trading program created under the Clean Air Act and administered by USEPA or the State of West Virginia, including but not limited to the Clean Air Interstate Rule 40 CFR Parts 96 and 97, and any amendments thereto ("Emission Allowances"), and all Emission Allowances for 2014 and any vintage year prior to 2014 that were allocated to the Mitchell Plant and that have not been expended as of the date of assignment. To the extent that additional Emission Allowances are required for operation of the Mitchell Plant, KPCo and WPCo will each be responsible for acquiring sufficient Emission Allowances to satisfy the Emission Allowances required because of its dispatch of energy from the Mitchell Plant, and the Emission Allowances required to satisfy the Emission Allowance surrender obligations attributable to the Mitchell Plant imposed under the Consent Decree between USEPA and Ohio Power Company entered on December 10, 2007, in Civil Action No. C2-99-1182 and consolidated cases by the U.S. District Court in the Southern District of Ohio. On or before January 10 of each year, Agent shall determine and notify KPCo and WPCo of the number of additional

annual Emission Allowances consumed by each of them through December 31 of the previous year, and KPCo and WPCo shall each transfer into the Mitchell Plant U.S. EPA Allowance Transfer System account that number of Emission Allowances with a small compliance margin by January 31 of that year. For seasonal Emission Allowance programs, Agent shall determine and notify KPCo and WPCo of the number of additional seasonal Emission Allowances consumed by each of them during the applicable compliance period by the 10th day of the first month following the end of the compliance period, and KPCo and WPCo shall each transfer into the appropriate Mitchell Plant U.S. EPA Allowance Transfer System Account that number of Emission Allowances with a small compliance margin by the last day of the first month following the end of the compliance period. In the event that KPCo or WPCo fails to surrender the required number of Emission Allowances by January 31 or the last day of the first month following any seasonal compliance period, Agent shall purchase the required number of Emission Allowances, and KPCo or WPCo, as the case may be, shall reimburse Agent for such purchases, with interest at the Federal Funds Rate (as published by the Board of Governors of the Federal Reserve System as from time to time in effect) running from the date of such purchases to the date of payment. The Operating Committee will develop procedures to be implemented after the end of each calendar year to account for the Emission Allowances required by the use of the Mitchell Plant by KPCo and WPCo and to correct any imbalance between Emission Allowances supplied and Emission Allowances used through the end of the preceding year by settlement or payment.

7.9 Capital repairs and improvements to the Mitchell Plant will be determined by the Operating Committee pursuant to the annual budgeting process set forth in Section 7.10. Expenditures that the Operating Committee determines have been or will be incurred exclusively for one Owner shall be assigned exclusively to that Owner. To the extent that capacity and energy production would not be available but for the investments and costs assigned exclusively to one owner, such capacity and energy production shall be assigned solely to the owner to which the costs are assigned.

At least 90 days before the start of each operating year, WPCo and Agent shall submit to the Operating Committee a proposed annual budget with respect to the Mitchell Plant, a proposed annual operating plan, and an estimate and schedule of costs to be incurred for major maintenance or replacement items during the next six-year period. The annual budget shall be presented on a month-by-month basis for each month during the next operating year, and shall include an operating budget, a capital budget, an estimate of the cost of any major repairs that are anticipated will occur during such operating year with respect to the Mitchell Plant, and an itemized estimate of all projected non-fuel variable operating expenses relating to the operation of the Mitchell Plant during that operating year. The members of the Operating Committee will meet and work in good faith to agree upon the final annual budget and final annual operating plan. Once approved, the annual budget and annual operating plan shall remain in effect throughout the applicable operating year, subject to such changes, revisions, amendments, and updating as the Operating Committee may determine.

# ARTICLE EIGHT

# EFFECTIVE DATE AND TERM

- 8.1 Subject to receipt of necessary regulatory approvals, the Effective Date of this Agreement shall be \_\_\_\_\_the Effective Date.
- 8.2 Subject to necessary regulatory approvals, this Agreement shall remain in force until such time as (i) KPCo or WPCo has divested itself of all or any portion of its ownership interest in the Mitchell Plant, other than assignment or other transfer of such ownership interests to another AEP affiliate; or (ii) either KPCo or WPCo is no longer a direct or indirect wholly owned subsidiary of AEP; or (iii) KPCo and WPCo may mutually agree to terminate this Agreement.

# ARTICLE NINE

# **GENERAL**

- 9.1 This Agreement shall inure to the benefit of and be binding upon the signatories hereto and their respective successors and assigns, but this Agreement may not be assigned by any signatory without the written consent of the others, which consent shall not be unreasonably withheld.
- 9.2 This Agreement is subject to the regulatory authority of any State or Federal agency having jurisdiction.
- 9.3 The interpretation and performance of this Agreement shall be in accordance with the laws of the State of West Virginia, excluding conflict of laws principles that would require the application of the laws of a different jurisdiction.
- 9.4 This Agreement supersedes all previous representations, understandings, negotiations, and agreements, either written or oral between the signatories or their representatives with respect to operation of the Mitchell Plant, and constitutes the entire agreement of the signatories with respect to the operation of the Plant. Notwithstanding the foregoing, this Agreement does not supersede any previous agreements among any of the signatories allocating or transferring rights to capacity and associated energy, or ownership, of the Mitchell Plant.
- 9.5 Each Party shall designate in writing a representative to receive any and all notices required under this Agreement. Notices shall be in writing and shall be given to the representative designated to receive them, either by personal delivery, certified mail, facsimile, e-mail or any similar means, properly addressed to such representative at the address specified below:

KENTUCKY POWER COMPANY				
President & CEO				
Attn:				
Phone:				

Facsimile:
E-Mail:
WHEELING POWER COMPANY
President
Attn:
Phone:
Facsimile:
E-Mail:
<u>AMERICAL</u> <u>AMERICAN</u>
ELECTRIC POWER SERVICE
CORPORATION
Executive Vice President Generation
Attn:
Phone:
Facsimile:
E-Mail:

All notices shall be effective upon receipt, or upon such later date following receipt as set forth in the notice. Any Party may, by written notice to the other Parties, change the representative or the address to which such notices are to be sent.

# ARTICLE TEN

# LIMITATION OF LIABILITY

10.1 Notwithstanding anything in this Agreement to the contrary, neither of the Owners or Agent shall be liable under this Agreement for special, consequential, indirect, punitive

or exemplary damages, or for lost profits or business interruption damages, whether arising by statute, in tort or contract or otherwise.

11.1 [Use language from Appendix B, modified Ownership Agreement or reference modified Ownership agreement]

# ARTICLE ELEVEN

# **DISPUTE RESOLUTION**

- 11.1 If either Owner believes that a dispute has arisen as to the meaning or application of this

  Agreement, it shall submit a written description of the disputed matter to the Operating

  Committee, and shall provide a copy of that submission to the other Owner.
- If the Operating Committee does not reach agreement on the resolution of a dispute within thirty (30) days after the dispute is presented to it, the matter shall be referred to senior executive officers with the authority to resolve such dispute of each of the Owners for resolution in the manner that such individuals shall agree is appropriate; provided. however, that either Owner may exercise any and all rights at law or equity at any time after the end of the thirty (30) day period provided for the Operating Committee to reach agreement if the Operating Committee has not reached agreement.
- Except as provided in this Article Eleven, the existence, contents, or results of any settlement negotiations or the results thereof under this Article Eleven may not be disclosed without the prior written consent of the Owners, provided, however, that either Owner may make disclosures as may be required to fulfill regulatory obligations to any governmental authority having jurisdiction, and may inform its lenders, affiliates, auditors, and insurers, as necessary, under pledge of confidentiality, and may consult with expert consultants as required in connection with any proceeding under pledge of confidentiality.
- Nothing in this Agreement shall be construed to preclude either Owner from filing a petition or complaint with FERC with respect to any claim over which FERC has jurisdiction. In such case, the other Owner may request that FERC reject the petition or complaint or otherwise decline to exercise its jurisdiction. If FERC declines to act with

respect to all or part of a claim, the portion of the claim not so accepted by FERC may be resolved through an action at law or equity. To the extent that FERC asserts or accepts jurisdiction over all or part of a claim, the decisions, findings of fact, or orders of FERC shall be final and binding, subject to judicial review under the Federal Power Act, 16 U.S.C. §§ 791a et seq., as amended from time to time, and any proceedings that may have commenced prior to the assertion or acceptance of jurisdiction by FERC shall be stayed, pending the outcome of FERC proceedings. To the extent that any decisions, findings of fact, or orders of FERC do not provide a final or complete remedy to an Owner seeking relief, such Owner may proceed at law or equity to secure such a remedy, subject to any FERC decisions, findings, or orders.

If an Owner (the "Contesting Owner") contests in good faith any amount paid pursuant to the terms of this Agreement, and any portion of such amount is determined or resolved (including pursuant to the dispute resolution procedures of this Article Eleven) to be in excess of the actual amount due pursuant to the terms of this Agreement, then the Contesting Owner may charge the other Owner interest with respect to such excess amount from the day the payment was made until it is repaid to the Contesting Owner, at the rate equal to the prime rate as published from time to time in The Wall Street Journal (or any successor publication) plus five (5) percentage points per annum, calculated daily, regardless of whether the Contesting Owner has notified the other Owner in advance of its intention to charge interest with respect to such excess amount, and the other Owner shall make payment in full in respect of such excess amount and interest within thirty (30) days of written demand therefor.

IN WITNESS WHEREOF, the Parties hereto have caused this Agreement to be executed by their officers thereunto duly authorized as of the date first above written.

KENTUCK	YPOWER	COMPAN	Y
Ву:			
Title: Presi	dent & CO		
WHEELIN	G POWER	COMPANY	ď
Ву:			
Title: Presi	<del>dent</del>		
AMERICA	N ELECTR	IC POWER	R SERVICE
CORPORA	TION		
Ву:			
Title: Exec	utive Vice I	President -	Generation

# APPENDIX B

# [RATE SCHEDULE NO. 303] MITCHELL PLANT OWNERSHIP AGREEMENT KENTUCKY POWER COMPANY

and

WHEELING POWER COMPANY

# TABLE OF CONTENTS

	Page
ARTICLE ONE OWNERSHIP AND OPERATIONS	2
ARTICLE TWO APPORTIONMENT OF CAPACITY AND ENERGY	3
ARTICLE THREE REPLACEMENTS, ADDITIONS, AND RETIREMENTS	4
ARTICLE FOUR WORKING CAPITAL REQUIREMENTS	5
ARTICLE FIVE INVESTMENT IN FUEL	6
ARTICLE SIX APPORTIONMENT OF STATION COSTS	6
ARTICLE SEVEN OPERATING COMMITTEE AND OPERATIONS	9
ARTICLE EIGHT EFFECTIVE DATE AND TERM	14
ARTICLE NINE TRANSFERS	
ARTICLE TEN DEFAULTS AND REMEDIES	16
ARTICLE ELEVEN LIMITATION OF LIABILITY	
ARTICLE TWELVE DISPUTE RESOLUTION	18
ARTICLE THIRTEEN GENERAL	20
ARTICLE FOURTEEN DEFINITIONS	23

Exhibit B — Form of Monthly Sample Report

THIS MITCHELL PLANT OWNERSHIP AGREEMENT (this "<u>Agreement</u>"), with an effective date of [] (the "<u>Effective Date</u>"), is by and among Kentucky Power Company, a Kentucky corporation qualified as a foreign corporation in West Virginia ("<u>KPCo</u>"); Wheeling Power Company, a West Virginia corporation ("<u>WPCo</u>") (such parties hereinafter sometimes referred to as an "<u>Owner</u>" and together the "<u>Owners</u>"); and, solely with respect to <u>Section 13.4</u>, American Electric Power Service Corporation, a New York corporation ("<u>AEPSC</u>").

# WITNESSETH:

WHEREAS, KPCo and WPCo, as of the date hereof, each own a fifty percent (50%) undivided ownership interest in the Mitchell Power Generation Facility (each such percentage interest, an Owner's "Ownership Interest"), which consists of two coal-fired generating units (each, a "Unit"), with each Unit having a nominal nameplate capacity of 800 MW, located in Moundsville, West Virginia (as further defined herein, the "Mitchell Plant");

WHEREAS, KPCo, WPCo and AEPSC are parties to that certain Mitchell Plant Operating Agreement, dated as of December 31, 2014 (the "Original Operating Agreement");

WHEREAS, the Original Operating Agreement sets forth certain rights and obligations of the Owners and AEPSC with respect to the Mitchell Plant and the Owners' ownership thereof;

WHEREAS, pursuant to the Original Operating Agreement, KPCo is responsible for the day-to-day operations and maintenance of the Mitchell Plant;

WHEREAS, the Owners and AEPSC desire to replace the Original Operating Agreement to set forth the rights and obligations of the Owners with respect to the Mitchell Plant and their ownership thereof and to remove AEPSC as a party thereto;

WHEREAS, in connection with the execution of this Agreement, the Owners desire to execute a separate operations and management agreement to provide for the day-to-day operation and maintenance responsibilities in respect of the Mitchell Plant (as may be amended from time to time the "O&M Agreement"); and

WHEREAS, the Owners have agreed that, subject to the terms and conditions of the O&M Agreement, on and after the Effective Date WPCo shall replace KPCo as the operator of the Mitchell Plant (the "Operator").

NOW THEREFORE, in consideration of the premises and for the purposes hereinabove recited, and in consideration of the mutual covenants hereinafter contained, the signatories hereto agree as follows:

# ARTICLE ONE

# OWNERSHIP AND OPERATIONS

- To the greatest extent permitted by Applicable Law, the Mitchell Plant and all assets (tangible and intangible) and property (real and personal) owned, leased, held, developed, constructed or acquired solely for or in connection with the Mitchell Plant or the operation, maintenance or Decommissioning of the Mitchell Plant by or on behalf of an Owner or the Owners (together, the "Project Assets") shall be owned and held and deemed to be owned and held by the Owners as tenants in common in proportion to their respective Ownership Interests (except for any capital items owned in a different proportion in accordance with Section 1.8) or, in the event any Project Asset cannot be held directly by both of the Owners due to, inter alia, any pre-existing legal or contractual restrictions that cannot be altered or satisfied or where effectuating such ownership structure would result in unreasonable additional expense to the Owners, by the Operator as trustee for the Owners as tenants in common in proportion to their respective Ownership Interest. If the ownership of any Project Asset is registered or recorded in the name of one of the Owners, and notwithstanding the Owners' efforts such Project Asset cannot be held directly by both Owners as contemplated above, then such Owner in whose name ownership is registered or recorded shall hold such Project Asset in trust for itself and the other Owner in proportion to their respective Ownership Interests and, to the extent necessary or requested by the Operator or other Owner, make such Project Assets (or the benefits thereof) available for the use and benefit of the Owners (in proportion with their respective Ownership Interests), including, to the extent consistent with the foregoing, by such Owner subcontracting, sublicensing, subleasing, delegating or granting a limited power of attorney or similar appointment as agent to Operator to administer such Project Assets.
- 1.2 At the request of either Owner, and in accordance with <u>Section 1.1</u>, each Owner and the Operator shall execute all documents and do all things necessary or appropriate to register or record the Project Assets in the names of the Owners in proportion to their respective Ownership Interests (or such different proportion as any capital item may be owned in accordance with <u>Section 1.8</u>).
- 1.3 All assets (tangible and intangible) and property (real and personal) held, developed, constructed or acquired by or on behalf of the Operator for or on behalf of the Owners jointly, or any of them, shall constitute "Project Assets" subject to the ownership of both Owners as set forth in Sections 1.1 and 1.2. Except as otherwise agreed by the Owners, the Operator shall not have any right, title or interest in or to any such assets, or in or to any money paid to, collected or received by the Operator for or on behalf of either Owner, except as the agent or representative of, or for the use and benefit of, such Owners as set forth in this Agreement and in proportion to each Owner's respective Ownership Interest.
- 1.4 Each Owner hereby waives any rights it may have at law or equity to bring an action for partition or division of the Mitchell Plant or any Project Asset or any contracts related thereto, and agrees that it shall not (a) seek partition or division of the Mitchell Plant or any Project Asset or any contracts related thereto, or (b) take any action, whether by way of any court order or otherwise, for physical partition or judicial sale in lieu of partition of the Mitchell Plant or any Project Asset or any contracts related thereto. Nothing in this Section 1.4 shall affect the right of

either Owner to dispatch its respective share of the Total Net Capability under <u>Article Two</u> or to Dispose of its Ownership Interest in accordance with Article Nine.

- 1.5 On and after the Effective Date, WPCo shall be the Operator responsible for the day-to-day operations and maintenance of the Mitchell Plant and shall operate, maintain and Decommission the Mitchell Plant for the sole benefit (and on behalf) of the Owners and in accordance with the terms and conditions of this Agreement and the O&M Agreement. KPCo agrees to take all actions reasonably necessary to facilitate WPCo's operation, maintenance and Decommissioning of the Mitchell Plant pursuant to the terms of the O&M Agreement, including providing or permitting reasonable access to the Mitchell Plant to third party contractors and other contract counterparties of each Owner or the Operator with respect to the administration, implementation and satisfaction of such contracts or agreements executed or assumed by the Operator on behalf of either Owner relating to the Mitchell Plant, including all Facility Agreements (as defined in the O&M Agreement).
- 1.6 The Owners shall establish and maintain such bank accounts as may from time to time be required or appropriate for paying the costs and expenses, including capital expenditures, in respect of the ownership, operation, maintenance and Decommissioning of the Mitchell Plant. The Owners shall designate only the Operator, and its representatives as reasonably requested by the Operator, as authorized signatories to such bank accounts. All withdrawals made by the Operator (or its representatives) from such bank accounts shall be made only in connection with the performance of the Operator's obligations set forth in this Agreement and the O&M Agreement.
- 1.7 The initial capital budget for the period from the Effective Date through December 31, 2028 (including agreed allocations of costs for capital projects between the Owners) (the "Capital Budget"), the initial annual operating budget and the initial forecast of operating and capital costs to be incurred for the period from the Effective Date through December 31, 2028 are attached hereto as Exhibit A.
- 1.8 Notwithstanding the provisions of this Article One, to the extent that either Owner funds or bears an amount greater than 50% of any capital expenditures or ELG Capital Expenditures as contemplated in the Capital Budget or this Agreement, the directly resulting portion of any property, plant and equipment, or improvements thereto shall be owned by the Owners in proportion to their respective amounts funded and shall be included only in such proportion in each Owner's ownership accounts for regulatory, accounting, tax and other purposes.

# ARTICLE TWO

# APPORTIONMENT OF CAPACITY AND ENERGY

2.1 The total net capability of the Mitchell Plant at low-voltage busses of the Units, after taking into account auxiliary load demand, is 1,560,000 kilowatts (the "Total Net Capability") as of the Effective Date. The Owners may from time to time modify the Total Net Capability of the Mitchell Plant as they may mutually agree.

- 2.2 The total net generation of the Mitchell Plant during a given period, as determined by the requirements of each Owner, shall mean the electrical output of the Mitchell Plant generators during such period, measured in kilowatt hours by suitable instruments, reduced by the energy used by auxiliaries for each Unit during such period (the "Total Net Generation").
- 2.3 Each Owner shall be entitled to receive 50% of the Total Net Capability and the Total Net Generation (with respect to each Owner, such Owner's "Assigned Capacity"), and all associated energy, capacity, ancillary services and other energy products, in accordance with this Agreement. Provided that such percentage shall be adjusted in accordance with Section 7.9 of the O&M Agreement.
- 2.4 Except as may be determined by the Operating Committee in accordance with Section 7.6, in any hour, each Owner shall share 50% of the minimum load responsibility of each Unit.
- 2.5 In any hour during which any Unit is out of service, the Owners shall bear equally the cost of energy used by the out-of-service Unit's auxiliaries during such hour, which may be provided by the applicable local utility Affiliate of an Owner. Alternatively, the Owners may mutually agree in writing to each provide 50% of such energy.

# ARTICLE THREE

# REPLACEMENTS, ADDITIONS, AND RETIREMENTS

- 3.1 The Owners shall take all actions within their respective control to cause the Operator, pursuant to the O&M Agreement, from time to time to make or cause to be made any necessary or appropriate additions to, replacements of, and retirements of, capitalizable facilities associated with the Mitchell Plant in accordance with the Capital Budget and the O&M Agreement or as may otherwise be mutually agreed upon by the Owners.
- 3.2 Each Owner shall, upon the retirement of the Mitchell Plant (or any individual Unit), (a) cooperate in good faith and take all actions reasonably necessary to facilitate the relevant Decommissioning thereof, including negotiating in good faith any contracts or agreements (including liability transfer arrangements) on behalf of either Owner or Operator, including transfers, conveyances or assignments of Facility Equipment (as defined in the O&M Agreement), as reasonably requested by either Owner or Operator to facilitate such Decommissioning and (b) take, and/or instruct the Operator pursuant to the O&M Agreement to take, such actions, at the sole cost and expense of WPCo, to continue operating and maintaining the barge loading facilities and gypsum conveyor system at the Mitchell Plant and providing use of such facilities and system to the applicable contract counterparty and its representatives in accordance with, and until the expiration or earlier termination of, the CertainTeed Contract.

# ARTICLE FOUR

# WORKING CAPITAL REQUIREMENTS

- 4.1 The Owners shall periodically mutually determine the amount, timing and invoicing processes for funds required for use as working capital, for operating, capital and other expenses incurred in the operation, maintenance and Decommissioning (including the Decommissioning Costs) of the Mitchell Plant, and in buying equipment, materials, parts, fuel and other supplies and services necessary to operate, maintain and Decommission the Mitchell Plant and to make the timely payments of any expenses required under the O&M Agreement.
- 4.2 Each Owner shall, in accordance with the timing set forth in a determination made pursuant to Section 4.1, promptly provide 50% of any such amount required by the Owners pursuant to Section 4.1, except as otherwise provided for in Section 6.7.
- Each Owner agrees that if such Owner fails at any time during the Term to satisfy the Ratings Requirement, it will, within thirty (30) days of such failure, provide in favor of the other Owner and maintain credit support in the form of (a) a cash deposit, (b) a guaranty issued by an Affiliate of such Owner that satisfies the Ratings Requirement in form and substance reasonably acceptable to the other Owner or (c) a letter of credit in form and substance reasonably acceptable to other Owner, issued by a commercial bank or other financial institution with a Credit Rating of at least "A-" by S&P Global Ratings, or any successor thereto ("S&P") or at least "A3" by Moody's Investors Service, Inc., or any successor thereto ("Moody's"), and in an amount equal to (i) one-half (1/2) of the then-applicable annual operating budget for the Mitchell Plant established pursuant to Section 7.2 from time to time, plus (ii) the sum of such Owner's allocated amount of capital expenditures for such year contained in the then-applicable Capital Budget, plus (iii) an amount equal to the latest estimate of Decommissioning Costs prepared by the Operator, determined on a net present value basis using a discount rate equal to the WACC as of the date of determination. Such credit support posted in favor of an Owner shall be promptly returned within thirty (30) days of the other Owner furnishing written evidence demonstrating that it satisfies the Ratings Requirement.
- 4.4 The Operator shall provide such credit support, including guarantees, cash deposits, letters of credit or other forms of credit support, to third parties (including contractual counterparties and Governmental Authorities) as required for the Owners' ownership, operation, maintenance and Decommissioning of the Mitchell Plant. To the extent that the Operator is required to provide such credit support to a third party in connection with any activity performed in respect of the Mitchell Plant under this Agreement (including the procurement of fuel as described in Section 5.1), the Owners shall share the reasonable and documented out-of-pocket cost of the third-party credit support incurred by the Operator (including of any credit support furnished by an Affiliate of the Operator) in accordance with their respective Ownership Interests.

# ARTICLE FIVE

# **INVESTMENT IN FUEL**

- 5.1 The Operator shall procure, establish and maintain reserves of coal in common stock piles for the Mitchell Plant of such quality and in such quantities as the Operating Committee shall determine to be required to provide adequate fuel reserves against interruptions of normal fuel supply and as is necessary to maintain the number of tons in such coal stock piles, after taking into account the coal consumption from such coal stock piles by each Unit during each month. For purposes of this Agreement, "consumables" shall be as defined in account 502 of the Uniform System of Accounts administered by the Federal Energy Regulatory Commission ("FERC").
- 5.2 The quality of any coal or consumable product provided by the Operator must be reasonably acceptable to both Owners. Any coal being utilized shall be deemed to be acceptable to the Owners if it meets the following requirements: (a) coal previously utilized at the Mitchell Plant with satisfactory operating performance shall be considered acceptable for use in the Mitchell Plant, unless deemed unacceptable due to a required change of the engineering specifications making the coal no longer viable; (b) coal from any new seam or source shall be acceptable if such supply is shown to perform satisfactorily in the Mitchell Plant and is mutually acceptable to each Owner; or (c) as otherwise mutually agreed to by each Owner. Consumables from any new seam or source shall be acceptable if such supply is shown to perform satisfactorily to both Owners in the Mitchell Plant and conform to the then current engineering specifications for the Mitchell Plant or as otherwise mutually agreed by each Owner.
- 5.3 Each Owner shall be responsible for, and own, 50% of the investment in the common coal stock piles or such percentage as adjusted pursuant to <u>Article Five</u> of the O&M Agreement.
- 5.4 Fuel oil and consumables charged to operation for the Mitchell Plant shall be owned and accounted for between the Owners in the same manner as coal.

# ARTICLE SIX

# APPORTIONMENT OF STATION COSTS

- 6.1 The allocation to the Owners of fuel expense associated with each Unit shall be determined by the Operating Committee as follows:
- (a) In any calendar month, the average unit cost of coal available for consumption from the Mitchell Plant common coal stock piles shall be determined based on the prior month's ending inventory dollar and ton balances plus current month receipts delivered to the Mitchell Plant common coal stock piles. Each Owner's average unit-cost will be the same, and receipts and inventory available for consumption amounts will be allocated to each Owner based on monthly usage.

- (b) The number of tons of coal consumed by the Mitchell Plant in each calendar month from the Mitchell Plant common coal stock piles shall be determined and shall be converted into a dollar amount equal to the product of (i) the average cost per ton of coal associated with the Mitchell Plant in the Mitchell Plant common coal stock pile at the close of such month, and (ii) the number of tons of coal consumed by the Mitchell Plant from the Mitchell Plant common coal stock piles during such month. Such dollar amount shall be credited to the Mitchell Plant fuel in the stock pile and charged to the Mitchell Plant fuel consumed.
- (c) In each calendar month, each Owner's respective shares of the Mitchell Plant fuel consumed expense as determined by the provisions of <u>Section 6.1(b)</u> shall be proportionate to each Owner's dispatch of the Mitchell Plant in such month.
- (d) Fuel oil reserves will be owned and accounted for in the same manner as coal stock piles, and fuel oil consumed will be allocated to the Owners in the same manner as coal consumed.
- 6.2 For each calendar month, the Operator will, to the extent practicable and in accordance with the O&M Agreement, determine all of the Mitchell Plant's operations expenses and associated overheads, as accounted for under the FERC Uniform System of Accounts.
- 6.3 For each calendar month, the Operator will, to the extent practicable and in accordance with the O&M Agreement, determine all the Mitchell Plant's maintenance expenses and associated overheads, as accounted for under the FERC Uniform System of Accounts.
- 6.4 In each calendar month, each Owner's respective shares of operations and maintenance expenses associated with the Mitchell Plant, as determined in accordance with this <u>Article Six</u>, shall be allocated as follows:
- (a) Each Owner's respective share of the Mitchell Plant steam expenses as recorded in FERC Account 502, and emission tons, with allowance expenses as recorded in FERC Account 509, shall be proportionate to each Owner's dispatch of the Mitchell Plant in such month.
- (b) In each calendar month, the maintenance of boiler plant expenses as recorded in FERC Account 512, and maintenance of electric plant expenses as recorded in FERC Account 513, shall be directly assigned to each Unit or designated as a common expense attributable to both Units. In each calendar month, each Owner's respective share of these expenses shall be proportionate to each Owner's dispatch of the applicable Unit, or both Units in the case of common expenses, over the previous sixty (60) calendar months.
- (c) In each calendar month, each Owner shall be responsible for 50% of all other Steam Power Generation Expenses (FERC Accounts 500 515) not addressed in Section 6.4(a) and Section 6.4(b). Administrative and General Expenses (FERC Accounts 920 935) shall be assigned to the Mitchell Plant through an annual wages and salaries allocator applied to monthly Administrative & General Expenses. Each Owner shall be responsible for 50% of this monthly amount; provided, however, that, for the avoidance of doubt, each Owner shall be individually responsible for any fees, costs or other charges, including but not limited to those imposed by PJM Interconnection, L.L.C. ("PJM") or any regional transmission operator or any

other Governmental Authority in respect of, or which are attributable to, the sale or transmission of the capacity or energy associated with its Ownership Interest, as the case may be.

- (d) Notwithstanding the foregoing clauses (a) through (c) or anything else in this Agreement or the O&M Agreement to the contrary, in each calendar month, any operations and maintenance or other expenses to the extent attributable to any ELG Upgrade (regardless of the FERC Account to which it is charged) shall be allocated exclusively to and paid by WPCo.
- (e) In each calendar month, each Owner's respective share of Construction Work In Progress charged to FERC Account 107 shall be allocated on the same basis as capital expenditures, as set forth in <u>Section 6.7</u>.
- (f) In each calendar month, the net change in Mitchell Plant storeroom inventory (inventory purchases less issuances of inventory) charged to FERC Account 154 shall be allocated 50% to each Owner.
- (g) Each Owner shall be charged 50% of Operating Costs, as defined in and in accordance with Section 7.2 of the O&M Agreement, except to the extent a different allocation for specific FERC Accounts or otherwise is specified in this Article Six.
- 6.5 All taxes, duties or assessments levied against or with respect to each Owner's Ownership Interest, or an Owner's purchase, use, ownership or beneficial interest in, or income from, the Mitchell Plant shall be the sole responsibility of, and shall be paid by, the Owner upon whose purchase, use, ownership interest or beneficial interest or income said taxes or assessments are levied. Without limiting the foregoing, in each calendar month, each Owner's respective share of Employee Payroll Taxes charged to FERC Account 408 shall be 50%.
- 6.6 Notwithstanding any other provision of this Agreement or any other agreement to the contrary, each Owner hereby acknowledges and agrees that (a) each Owner prior to the Effective Date has treated, and subsequent to such date shall continue to treat, the co-ownership and operation of the Mitchell Plant as excluded from Subchapter K of the Internal Revenue Code of 1986, as amended (the "Tax Code"), pursuant to Section 761(a) thereof, for all federal, state and local income tax purposes, (b) each Owner prior to the Effective Date affirmatively elected not to apply any of the provisions of Subchapter K of the Tax Code to such Owner's interest in the Mitchell Plant, with such election having been formally filed in connection with the Owners' applicable income tax returns for the taxable year ending on December 31, 2020 and each Owner has taken all actions necessary to implement such election and (c) each Owner prior to the Effective Date has reported, and subsequent to such date shall report, its share of all income, gains, deductions, losses, credits, etc. from its Ownership Interest on its tax returns consistent with such exclusion from the provisions of Subchapter K of the Tax Code.
- 6.7 Subject to <u>clauses (b)</u> and <u>(c)</u> below the cost of any replacement, addition, improvement or upgrade of each Unit or any portion of the Mitchell Plant, and any restoration or remediation required in connection therewith, shall be allocated between the Owners in accordance with the allocations for such capital items contained in the Capital Budget. With respect to any such capital item not contained in the Capital Budget, the costs of such capital item shall be allocated as follows, unless the Operating Committee agrees upon a different allocation:

- (a) Capital expenditures (other than ELG Capital Expenditures) that the Operating Committee determines have been or will be incurred exclusively for one Owner shall be allocated exclusively to, and paid for by, that Owner.
- (b) Notwithstanding anything to the contrary herein, ELG Capital Expenditures shall be allocated exclusively to, and paid for exclusively by, WPCo and CCR Capital Expenditures shall be allocated 50% to (and paid for by) each Owner; provided, that, the Operating Committee shall engage or retain a Technical Expert to make recommendations with respect to determining which capital expenditures are ELG Capital Expenditures and which capital expenditures are CCR Capital Expenditures.
- (c) Notwithstanding anything to the contrary herein, if the in-service date of a capital item is reasonably anticipated by the Operating Committee to be after December 31, 2028, then the capital expenditures for such capital item shall be allocated exclusively to, and paid for by, WPCo.
- (d) If the Operating Committee determines, including based on Depreciable Lives of similar assets previously approved by applicable Governmental Authorities, that a capital item (other than an ELG Upgrade) has a Depreciable Life that extends beyond December 31, 2028, then (i) KPCo shall be responsible for and shall pay 50% of the expenditures for such capital item, multiplied by (A) the number of months (not to exceed the Depreciable Life of such capital item) between the reasonably anticipated in-service date of such capital item and December 31, 2028, divided by (B) the Depreciable Life of such capital item and (ii) WPCo shall be responsible for the remaining amount of such capital expenditure not allocated to KPCo pursuant to the foregoing clause (i).
- (e) Any other capital expenditures shall be allocated 50% to (and paid for by) each Owner, subject to the written approval of the Operating Committee for budget overruns to the extent required pursuant to Section 5.3.2 of the O&M Agreement.
- 6.8 Each Owner shall be responsible for 50% of all Decommissioning Costs, unless a different allocation is expressly specified for such item in the Capital Budget (as agreed by the Owners) or the Owners mutually agree to allocate such costs in another manner.
- 6.9 Notwithstanding anything contained in this Agreement, an Owner's obligation to pay its obligations under this Agreement shall not in any way be conditioned upon or affected by any regulatory order or other determination disallowing, limiting or deferring rate recovery of the costs and expenses paid or payable by an Owner in respect of its Ownership Interest.

# ARTICLE SEVEN

# **OPERATING COMMITTEE AND OPERATIONS**

7.1 By written notice to each other, each Owner shall name one representative (the "Operating Representative") and one alternate to act for it in matters pertaining to operating arrangements under this Agreement and the O&M Agreement. An Owner may change its

Operating Representative or alternate at any time by written notice to the other Owner. The Operating Representatives for the respective Owners, or their alternates, shall comprise the "Operating Committee". All decisions, directives, or other actions by the Operating Committee must be by unanimous agreement of the Operating Representatives of the Owners. If the Operating Representatives are unable to agree on any matter, such matter will be resolved through the dispute resolution procedures set forth in Article Twelve.

- 7.2 The Operating Committee shall have the following responsibilities, which decisions are reserved exclusively for the Operating Committee and may not be made individually by the Operator or any Owner:
- (a) Review and approval of any amendments to the Capital Budget, and adoption of an annual operating budget, annual operating plan and a six-year forecast of operating and capital expenses, each as delivered to the Operating Committee by the Operator pursuant to Section 7.8, including determination of the emission allowances required to be acquired by each Owner with respect to their Ownership Interests; provided, that an Owner's Operating Representative shall have the right to amend the Capital Budget solely to include any capital expenditures for which such Owner shall be allocated greater than 75% of the costs pursuant to Section 6.7, up to an aggregate amount of such capital expenditures that does not exceed \$3 million per year allocated to the other Owner. Allocations of new capital expenditures added to the Capital Budget shall be consistent with Section 6.7; provided, that if the Operating Committee cannot agree upon the Depreciable Life of a capital item or the allocation of a capital expenditure between the Owners (including determining which capital expenditures are ELG Capital Expenditures and which capital expenditures are CCR Capital Expenditures), the matter shall be resolved in accordance with the Technical Dispute resolution procedures set forth in Section 12.1 and Section 12.3 and the Owners shall implement any resolution of the Technical Dispute through adjustments or true-up payments, as appropriate. If the Operating Committee fails to adopt an annual operating budget, the approved annual operating budget from the previous year (other than one-time or other non-recurring or inapplicable items) shall apply until such time as the new annual operating budget is approved.
- (b) Establishment and monitoring of procedures for communication and coordination with respect to the Mitchell Plant capacity availability, fuel-firing options, and scheduling of outages for maintenance, repairs, equipment replacements, scheduled inspections, and other foreseeable cause of outages at the Mitchell Plant, as well as the return the Mitchell Plant to availability following an unplanned outage. The Operating Committee shall use commercially reasonable efforts, consistent with Prudent Operation and Maintenance Practices (as defined in the O&M Agreement), to schedule the implementation of ELG Upgrades during planned maintenance and repair outages so as to eliminate or minimize incremental outages.
- (c) To the extent not included in the Capital Budget, decisions on capital projects, including Unit upgrades and re-powering, except that an Owner's Operating Representative shall have the right to approve any such capital projects for which such Owner shall be allocated greater than 75% of the costs pursuant to Section 6.7 and Section 7.2(a).
- (d) Determinations as to allocations between the Owners of expenses pursuant to <u>Section 6.1</u>.

- (e) Determinations as to changes in the Unit capability.
- (f) Establishment and modification of billing procedures under this Agreement or under the O&M Agreement.
  - (g) Approval of material contracts for fuel supply or transportation.
- (h) Establishment and modification of specifications of fuels; oversight of fuel procurement, inspection and certification arrangements, policies and procedures; and management of fuel inventories for the Mitchell Plant.
- (i) Establishment of, termination of, and approval of any change or amendment to the operating arrangements (including the O&M Agreement) between the Owners and the Operator (or any successor Operator or replacement third-party Operator) and selection of any replacement Operator, except as otherwise permitted by Section 7.9.
- (j) Review and approval of plans and procedures designed to ensure compliance at the Mitchell Plant with all Applicable Law, including procedures for allocating and using emission allowances or for any programs that permit averaging at more than one Unit for compliance.
- (k) Amendment, termination, extension or modification of the O&M Agreement, and supervision of the performance of, and provision of direction as needed to, the Operator.
- (l) Decisions regarding the retirement, permanent removal from service or Decommissioning of a Unit or any material portion of the Mitchell Plant and any restoration or remediation required in connection therewith.
- (m) Establishment of an insurance program to provide property and general liability insurance on behalf of each Owner, to be procured by the Operator pursuant to the O&M Agreement.
  - (o) Other duties as assigned by agreement of the Owners.
- 7.3 The Operating Committee shall meet at least quarterly, or at such other frequency as determined by the Operating Committee, and at such other times as an Owner may reasonably request. The Operator shall provide operations reports to the Operating Committee each month (presented on a monthly basis) and each quarter (presented on a quarterly basis) substantially in the form of Exhibit B hereto.
- 7.4 The Owners and the Operator shall cooperate in providing to the Operating Committee the information it reasonably needs to carry out its duties, and to supplement or correct such information on a timely basis.
- 7.5 The Owners will each make an initial unit commitment one business day ahead of real-time dispatch.

- 7.6 Application of this <u>Section 7.6</u> (including subsections) is subject to (i) the receipt of any necessary regulatory approvals or waivers expressly granted for this <u>Section 7.6</u>; and (ii the Operating Committee establishing and approving procedures and systems for dispatch. As used in this Section and subsections of this Section, the terms "Party" or "Parties" refers only to KPCo and WPCo, or both of them, as the case may be.
- 7.6.1 If Mitchell Unit 1 or Unit 2 is designated to be committed by both Parties, such unit will be brought on line or kept on line. If neither Party designates Mitchell Unit 1 or Unit 2 to be committed, such unit will remain offline or be taken offline.
- 7.6.2 When a Mitchell Unit is designated to be committed by one Party, but designated not to be committed by the other Party, the unit will be brought on line or kept on line if the Party designating the unit for commitment undertakes to pay any applicable start-up costs for the unit, as well as any applicable minimum running costs for the unit thereafter, in which event the unit shall be brought on line or kept on line, as the case may be. The Party so designating the unit to be committed shall have the right to schedule and dispatch up to all of the Available Capacity of the unit. Available Capacity means that portion of the Owners' aggregate Assigned Capacity that is currently capable of being dispatched. The Party exercising this right shall be referred to as the "Calling Party," and the capacity called by that Party in excess of its Assigned Capacity Percentage of the Available Capacity of that unit shall be referred to as its "Called Capacity." The other Party shall be referred to as the "Non-Calling Party". The Calling Party shall provide reasonable notice to the Non-Calling Party of its call, including any start-up or shut-down time for the Unit. For purposes of this Agreement, KPCo's Assigned Capacity Percentage shall be 50%, and WPCo's Assigned Capacity Percentage shall be 50% as adjusted pursuant to section 7.9 of the O&M Agreement.
- 7.6.3 The Non-Calling Party can reclaim any Called Capacity attributable to its Assigned Capacity share by giving the Calling Party notice equal to the normal cold start-up time for the unit. At the end of the notice period, the Non-Calling Party shall have the right to schedule and dispatch the recalled capacity. At that point, the Non-Calling Party shall resume its responsibility for its share of any applicable start-up costs for the unit and prospectively shall bear its responsibility for the costs associated with its Assigned Capacity from the unit.
- 7.6.4 If any capacity remains available but is not dispatched from a Party's Available Capacity committed as a result of the initial unit commitment, the other Party may schedule and dispatch such capacity pursuant to 7.6.1 through 7.6.3, above.
- 7.7 Emission Allowances. Prior to the earlier of any Buyout Transaction or December 31, 2028 (or earlier retirement of the Facility), to the extent that emission allowances issued by the U.S. Environmental Protection Agency ("USEPA") pursuant to Title IV of the Clean Air Act Amendments of 1990 and any regulations thereunder, and any other emission allowance trading program created under the Clean Air Act and administered by USEPA or the State of West Virginia, including but not limited to the Cross-State Air Pollution Rule 40 C.F.R. Part 97, and any amendments thereto (the "Emission Allowances"), are required for operation of the Mitchell Plant, each Owner will be entitled to receive for its own benefit 50% of any Emissions Allowances allocated to the Mitchell Plant. Each Owner will be responsible for acquiring any additional Emission Allowances needed to satisfy the Emission Allowances required because of such

12

Owner's dispatch of energy from the Mitchell Plant. Additionally, prior to such time, each Owner will be responsible for acquiring the Emission Allowances required, to the extent necessary in addition to its share of the Emissions Allowances allocated to the Mitchell Plant, to satisfy 50% of the Emission Allowance surrender obligations attributable to the Mitchell Plant imposed under the Consent Decree entered in United States, et al. v. American Electric Power Service Corp., et al., Civil Action Nos. C2-99-1182 and C2-99-1250 and United States, et al. v. American Electric Power Service Corp., et al., Civil Action No. C2 05-360 and Ohio Citizen Action, et al. v. American Electric Power Service Corp., Civil Action No. C2-04-1098 dated December 10, 2007 as subsequently modified or amended, it being understood that the Owners may be subject to additional rights and obligations under any applicable agreement among the Owners (and/or their Affiliates) and American Electric Power Company, Inc. (and/or its Affiliates) pertaining to the allocation of emission limitations associated with the Mitchell Plant. As early as possible, but no later than three business days after the deadline for submitting final electronic data to the EPA for compliance purposes, the Operator shall notify each Owner of the number of annual or seasonal Emission Allowances that are needed to offset each Owner's share of emissions for the previous year or season. Each Owner shall supply its respective share of allowances, with a reasonable compliance margin as determined by the Operating Committee, by transferring the applicable allowances to the Mitchell Plant's Allowance Facility Account on or before 15 days prior to the remittance date. In the event that an Owner fails to surrender the required number of Emission Allowances in accordance with the prior paragraph, the other Owner shall have the option to purchase the required number of Emission Allowances, and the Owner that failed to surrender the required number of Emission Allowances shall reimburse the other Owner for any amounts it shall have incurred to make such purchases, with interest at the "Federal Funds Rate" (as published by the Board of Governors of the Federal Reserve System as from time to time in effect) running from the date of such purchases to the date of payment. The Operating Committee will develop procedures to be implemented after the end of each calendar year to account for each Owner's share of the Emission Allowances required by the use of the Mitchell Plant and to correct any imbalance between the Emission Allowances supplied and the Emission Allowances used through the end of the preceding year by settlement or payment.

At least ninety (90) days before the start of each operating year, the Operator shall submit to the Operating Committee any proposed amendments to the Capital Budget and an annual operating budget for such operating year with respect to the Mitchell Plant, a proposed annual operating plan with respect to the Mitchell Plant for such operating year, and a forecast of operating and capital costs to be incurred during the next six-year period. The annual operating budget and amendments to the Capital Budget shall be presented on a month-by-month basis, and shall include an operating budget, a capital budget, and an estimate of the cost of any major repairs or improvements that are anticipated to occur during the relevant period with respect to the Mitchell Plant, and an itemized estimate of all projected fixed and variable operating expenses relating to the operation of the Mitchell Plant during that operating year. The members of the Operating Committee will meet and work in good faith to agree upon the final annual operating budget, final annual operating plan and any amendments to the Capital Budget. Once approved, the annual operating budget and annual operating plan shall remain in effect throughout the applicable operating year, subject to such changes, revisions, amendments, and updating as the Operating Committee may determine. If the Operating Committee determines to retire the Mitchell Plant prior to December 31, 2028, the members of the Operating Committee will meet and work in good faith to amend the Capital Budget to remove any future ELG Capital Expenditures and any other future capital expenditures no longer required, to the extent practicable and consistent with Applicable Law. The Capital Budget shall remain in effect throughout the Term, subject to such amendments as the Operating Committee may determine.

Notwithstanding anything in this Agreement to the contrary, (i) in the case of the O&M Agreement or any other agreement relating to the Mitchell Plant that is entered into jointly by or on behalf of the Owners, on one hand, with an Affiliate of an Owner (or with an Owner itself, as in the case of the O&M Agreement) on the other hand, the non-Affiliate Owner shall have the sole and exclusive right to exercise any and all affirmative or elective rights of the Owners, including remedies (including delivering notices of and pursuing or settling disputes or delivering notices of default or making and pursuing claims for indemnification) and any termination rights (including rights of termination for convenience, if any) thereunder (for the avoidance of doubt, without first obtaining the consent of the other Owner or the Operating Committee) and (ii) in the case the O&M Agreement is terminated pursuant to Section 8.2 thereof, KPCo shall have the sole and exclusive right to select and designate any successor "Operator" or replacement third-party Operator, in each case so long as such successor replacement is a "Qualified Replacement Operator" (as defined in the O&M Agreement); provided, however, that notice of any such action described in this Section 7.9 shall be sent to the other Owner at the time such action is taken if such other Owner is not the Operator. For purposes of this Agreement, "Affiliate" shall mean, with respect to any person or entity, any other person or entity that directly or indirectly, controls, is controlled by, or is under common control with such person or entity. As used in this definition, "control" (including, with its correlative meanings, "controlled by" and "under common control with'') means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a person or entity, whether through the ownership of securities or partnership or other ownership interests, by contract or otherwise.

# ARTICLE EIGHT EFFECTIVE DATE AND TERM

- 8.1 This Agreement shall be effective as of the Effective Date.
- 8.2 Subject to FERC approval or acceptance of any termination, if necessary, this Agreement shall remain in force until the date on which this Agreement is terminated by mutual agreement of the Owners (the period from the Effective Date through such date, the "Term").

# ARTICLE NINE TRANSFERS

9.1 Neither Owner may assign, transfer or otherwise dispose of its Ownership Interest, either in whole or part, whether by sale, lease, division, declaration or creation of a trust, by operation of law or otherwise ("<u>Dispose</u>" or a "<u>Disposition</u>") to any person or entity (the "<u>Proposed Purchaser</u>") without the prior written consent of the other Owner (the "<u>Non-Offering Owner</u>" and

the Owner proposing the Disposition, the "Offering Owner"), which consent may be granted or withheld in the Non-Offering Owner's sole discretion. Notwithstanding the foregoing, either Owner may Dispose of, all (but not less than all) of its Ownership Interest to a state regulated utility Affiliate, provided that (i) the Disposition shall not relieve the Offering Owner of its obligations under this Agreement, (ii) the Disposition shall be made in compliance with the Consent Decree entered in United States, et al. v. American Electric Power Service Corp., et al., Civil Action Nos. C2-99-1182 and C2-99-1250 and United States, et al. v. American Electric Power Service Corp., et al., Civil Action Nos. C2-04-1098 and C2-05-360, and all amendments or modifications thereto, as in effect as of the date of the Disposition, (iii) the Proposed Purchaser shall agree to and assume, in respect of the Ownership Interest subject to the Disposition, the rights and obligations of the Offering Owner and its Affiliates under any applicable agreement with American Electric Power Company, Inc. (and/or its Affiliates) pertaining to the allocation of emission limitations associated with the Mitchell Plant, and (iv) in the event the Offering Owner (or any Affiliate thereof) shall be the Operator, the Proposed Purchaser shall also have been assigned, and agreed to have assumed, the rights and obligations of the Operator under this Agreement and the O&M Agreement as of the effective date of such Disposition; provided, that in the case of this clause (iv), a written consent from the Non-Offering Owner (which consent shall not be unreasonably withheld, conditioned or delayed) shall be obtained prior to such Disposition to the extent such Disposition results in the change of the Operator.

- 9.2 No Disposition shall be made unless all requisite regulatory and other approvals, consents and authorizations from all Governmental Authorities that are required to be obtained in connection with such Disposition have been obtained and as to which all conditions to the consummation of Disposition thereunder have been satisfied.
- 9.3 All costs associated with any Disposition of an Ownership Interest by an Owner shall be borne solely by the Offering Owner, <u>provided</u> that the foregoing shall not limit the Offering Owner's right to seek reimbursement of any costs from the Proposed Purchaser in connection with any such Disposition.
- Each Owner shall have the right to seek financing for all or a portion of such Owner's Ownership Interest and to provide general security for such financing of its Ownership Interest, including through the creation of any Encumbrance thereon (and the right of the beneficiary thereof to enforce thereon, but not to foreclose upon or transfer such Owner's Ownership Interest without the prior written consent of the other Owner), without the prior consent of the other Owner. Each Owner further agrees to cooperate reasonably and in good faith, and to cause its Affiliates to so cooperate, with an Owner seeking financing in connection with such modifications and other rights and consents customary in transactions of such type, and not unreasonably to withhold its consent to such modifications as may be reasonably necessary or appropriate to allow such Owner to obtain such financing upon reasonably competitive terms, including obtaining consents to the assignment of such Owner's Ownership Interest in any of the Project Assets reasonably requested by such Owner's lender; provided that none of such proposed modifications shall (a) relieve the financing Owner of any of its obligations under this Agreement, the O&M Agreement or any other agreement related to the Mitchell Plant or any Project Asset, (b) decrease the economic benefits, or increase the costs, of the ownership and operation of the Mitchell Plant to the other Owner, (c) create any increased economic or legal risk to the other Owner in connection with the ownership and operation of the Mitchell Plant, (d) permit or allow

any Encumbrances relating to any such financing to be placed upon any portion of or interest in the Project Assets other than the financing Owner's Ownership Interest; or (e) permit partition of the Project Assets or any of them, including any partition upon a default by the financing Owner under any of the relevant financing documents.

9.5 Notwithstanding anything else herein to the contrary, no Disposition shall constitute a release of the Offering Owner from any liabilities to the Non-Offering Owner or the Operator arising from events occurring prior to or in connection with the Disposition.

# ARTICLE TEN

#### **DEFAULTS AND REMEDIES**

- 10.1 An Owner shall be deemed to be in default hereunder upon the occurrence of any of the following events with respect to such Owner (each of the following events to be referred to as an "Event of Default," the Owner in default to be referred to as the "Defaulting Owner" and the Owner not in default to be referred to as the "Non-Defaulting Owner"):
- (a) an Owner fails to make any payment required by it as and when due and payable in accordance with the terms of this Agreement, the O&M Agreement or any other agreement related to the Mitchell Plant or any Project Asset and such failure is not remedied within ten (10) days after receipt of written notice thereof by such Owner from the other Owner; provided, that any such notice shall include a statement of the amount the Defaulting Owner has failed to pay (a "Payment Default"); or
- (b) an Owner fails to perform any material obligation (other than as described in Section 10.1(a)) imposed upon such Owner under this Agreement and such failure is not remedied within thirty (30) days after such Owner receives written notice thereof from the other; provided that, if such thirty (30) day period is not sufficient to enable the remedy or cure of such failure in performance, and such Owner shall have upon receipt of the initial notice promptly commenced and diligently continues thereafter to remedy such failure, then such Owner shall have a reasonable additional period of time (but in no event longer than an additional ninety (90) days from the end of the initial thirty (30) day cure period) to remedy or cure such failure; provided, however, that an Owner shall not be in default of its obligations hereunder to the extent such failure is caused by or is otherwise attributable to a breach by the other Owner of its obligations under this Agreement.
- 10.2 Without limiting the rights and remedies available to the Non-Defaulting Owner under Applicable Law, in the case of an Event of Default, the Non-Defaulting Owner shall have the right (but not the obligation) to (x) pay all or a portion of the amounts that were the subject of the Payment Default on behalf of the Defaulting Owner and (y) perform the obligation(s) which the Defaulting Owner has failed to perform on behalf of and at the expense of the Defaulting Owner (in any such case subject to all limits on liability benefiting the Defaulting Owner as set forth in this Agreement); and, if such payment is made (the portion as so paid or expended in connection with such performance, the "Paid Amount"), to:

- (a) charge the Defaulting Owner interest with respect to the Paid Amount, from the day the payment was made by the Non-Defaulting Owner until it is paid in full by the Defaulting Owner to the Non-Defaulting Owner, at the rate equal to the prime rate as published from time to time in *The Wall Street Journal* (or any successor publication) plus five (5) percentage points per annum, calculated daily, regardless of whether the Non-Defaulting Owner has notified the Defaulting Owner in advance of its intention to charge interest with respect to such Paid Amount;
- (b) set off against the Paid Amount any sums due or accruing to the Defaulting Owner by the Non-Defaulting Owner in accordance with this Agreement;
- (c) maintain an action or actions for the Paid Amount and interest thereon on a continuing basis as the Paid Amount becomes payable but is not paid by the Defaulting Owner, as if the obligation to pay those amounts and the interest thereon was a liquidated demand due and payable on the date the amounts were due to be paid, without any right or resort of the Defaulting Owner to set-off or counter-claim against the Non-Defaulting Owner; and any obligation to pay interest under this Section 10.2 shall apply until the Payment Default is rectified or remedied; and
- at the Non-Defaulting Owner's option, (i) draw on any letter of credit posted by the Defaulting Owner pursuant to Section 4.3 in an amount equal to the Paid Amount, including all interest accrued thereon or (ii) receive one hundred percent (100%) of any revenues arising from or attributable to the sale of capacity, energy, ancillary services or other energy products from the Mitchell Plant that the Defaulting Owner would otherwise be entitled to receive in respect of its Assigned Capacity until the Non-Defaulting Owner receives an amount equal to the Paid Amount, including all interest accrued thereon, plus all costs of collection incurred in connection therewith, and the Owners shall cooperate with each other, the Operator, applicable Governmental Authorities (including in respect of securing any regulatory approvals) or other third parties (including lenders) as may be reasonably necessary to facilitate the Non Defaulting Owner's right to be paid and receive the revenues attributable to the Defaulting Owner's Assigned Capacity until the applicable Paid Amount, including all interest accrued thereon and all costs of collection incurred in connection therewith has been paid to the Non-Defaulting Owner in full, including facilitating any appropriate changes in the applicable settlement accounts with respect to which market revenues are credited or paid by PJM or other applicable regional transmission organizations and executing any documents required to assign over such market revenues to the Non-Defaulting Owner.

# ARTICLE ELEVEN LIMITATION OF LIABILITY

11.1 Without limiting any other provision of this Agreement, each Owner's liability under this Agreement shall be limited to direct actual damages only. Such direct actual damages shall be the sole and exclusive remedy with respect to all claims arising under this Agreement and all other remedies or damages at law or in equity with respect to claims arising under this Agreement are waived, and unless expressly provided herein, no Owner shall be liable for

consequential, punitive, incidental, exemplary or indirect damages, lost profits or other business interruption damages, by statute, in tort or in contract, under any indemnity provision or otherwise, with respect to claims arising under this Agreement. It is the intent of the Owners that the limitations herein imposed on remedies and the measure of damages be without regard to the cause or causes related thereto, including the negligence of any Owner, whether such negligence be sole, joint or concurrent, or active or passive. Notwithstanding anything herein to the contrary, the limitations set forth in this Section 11.1 shall not limit or preclude any indemnification obligations of an Owner pursuant to Article Ten of the O&M Agreement, including with respect to indemnification for third-party claims.

# ARTICLE TWELVE DISPUTE RESOLUTION

- 12.1 If either Owner believes that a dispute (including a Technical Dispute) has arisen as to the meaning or application of this Agreement, it shall submit a written description of the disputed matter to the Operating Committee, and shall provide a copy of that submission to the other Owner.
- 12.2 If the Operating Committee does not reach agreement on the resolution of a dispute not constituting a Technical Dispute submitted to the Operating Committee pursuant to Section 12.1 within thirty (30) days after the dispute is presented to it, the matter shall be referred to senior executive officers with the authority to resolve such dispute of each of the Owners for resolution in the manner that such individuals shall agree is appropriate; provided, however, that either Owner may exercise any and all rights at law or equity at any time after the end of the thirty (30) day period provided for the Operating Committee to reach agreement if the Operating Committee has not reached agreement.
- If the Operating Committee does not reach agreement on the resolution of a Technical Dispute submitted to the Operating Committee within ten (10) business days after such Technical Dispute is presented to it, then either Owner may refer such Technical Dispute to a Technical Expert. Within ten (10) business days following receipt of an Owner's notice referring a Technical Dispute to a Technical Expert, the Operating Representatives shall confer to agree upon a Technical Expert to hear the Technical Dispute. If the Owners are unable to agree upon the appointment of a Technical Expert, then at the end of such ten (10) business day period each Owner shall, within five (5) business days, notify the other Owner in writing of its designation of a proposed Technical Expert. The two proposed Technical Experts shall, within five (5) business days, select a Technical Expert (who may be one of the two Technical Experts designated by the Owners or another Technical Expert) and such Technical Expert shall hear the Technical Dispute. Each Owner shall be required to put forth and endorse one proposal, budget or solution, as the case may be, as its proposed resolution to the Technical Dispute, based on an agreed statement of the nature of the Technical Dispute and agreed facts surrounding such Technical Dispute. Each Owner's proposal, budget or solution shall be delivered to the Technical Expert and the other Owner no later than twenty (20) business days after the date of the notice of the Owner submitting the Technical Dispute to the Technical Expert. The Technical Expert shall be guided by

consideration of (a) this Agreement, (b) all other agreements between the Owners relating to the Mitchell Plant, including the O&M Agreement and (c) Prudent Operation and Maintenance Practices (as defined in the O&M Agreement), and be required to select one of the proposals, budgets or solutions, as the case may be, and shall not be able to select any other proposal, budget or solution, except to the extent mutually agreed by the Owners. The Technical Expert shall render a decision resolving the matter within forty-five (45) days of the date of the notice of the Owner submitting such matter. The Technical Expert shall not award to either Owner any relief greater than that initially sought by such Owner. The decision of the Technical Expert shall be final and binding upon the Owners and not subject to appeal or review. The Owners shall bear equally all costs and expenses of the Technical Expert procedure and the Technical Expert shall not have the authority to award costs or attorneys' fees to either Owner. The Technical Expert shall act as an expert and not as an arbitrator and the provisions of the Federal Arbitration Act and the laws relating to arbitration shall not apply to the Technical Expert or the Technical Expert's determination or the procedure by which a determination is reached. Except as provided in Section 7.2(a), the Technical Expert's decision shall not in any event result in deviations from the agreed allocations of costs between the Owners as set forth in this Agreement.

- 12.4 Except as provided in this <u>Article Twelve</u>, the existence, contents, or results of any settlement negotiations or the results thereof under this <u>Article Twelve</u> may not be disclosed without the prior written consent of the Owners, <u>provided</u>, <u>however</u>, that either Owner may make disclosures as may be required to fulfill regulatory obligations to any Governmental Authority having jurisdiction, and may inform its lenders, affiliates, auditors, and insurers, as necessary, under pledge of confidentiality, and may consult with expert consultants as required in connection with any proceeding under pledge of confidentiality.
- a petition or complaint with FERC with respect to any claim over which FERC has jurisdiction. In such case, the other Owner may request that FERC reject the petition or complaint or otherwise decline to exercise its jurisdiction. If FERC declines to act with respect to all or part of a claim, the portion of the claim not so accepted by FERC may be resolved through an action at law or equity. To the extent that FERC asserts or accepts jurisdiction over all or part of a claim, the decisions, findings of fact, or orders of FERC shall be final and binding, subject to judicial review under the Federal Power Act, 16 U.S.C. §§ 791a et seq., as amended from time to time, and any proceedings that may have commenced prior to the assertion or acceptance of jurisdiction by FERC shall be stayed, pending the outcome of FERC proceedings. To the extent that any decisions, findings of fact, or orders of FERC do not provide a final or complete remedy to an Owner seeking relief, such Owner may proceed at law or equity to secure such a remedy, subject to any FERC decisions, findings, or orders.
- 12.6 If an Owner (the "Contesting Owner") contests in good faith any amount paid pursuant to the terms of this Agreement following receipt of the written notice of the other Owner delivered pursuant to Section 10.1(a), and any portion of such amount is determined or resolved (including pursuant to the dispute resolution procedures of this Article Twelve) to be in excess of the actual amount due pursuant to the terms of this Agreement, then the Contesting Owner may charge the other Owner interest with respect to such excess amount from the day the payment was made until it is repaid to the Contesting Owner, at the rate equal to the prime rate as published from time to time in The Wall Street Journal (or any successor publication) plus five (5) percentage

19

points per annum, calculated daily, regardless of whether the Contesting Owner has notified the other Owner in advance of its intention to charge interest with respect to such excess amount, and the other Owner shall make payment in full in respect of such excess amount and interest within thirty (30) days of written demand therefor.

# ARTICLE THIRTEEN GENERAL

- 13.1 This Agreement shall inure to the benefit of and be binding upon the signatories hereto and their respective successors and permitted assigns, but this Agreement may not be assigned by any signatory without the written consent of the other parties hereto or as permitted by Article Nine hereof.
- 13.2 This Agreement and all of its terms, including those addressing future events, is subject to the regulatory authority of any state or federal agency having jurisdiction, including the KPSC and the WVPSC. Each Owner is responsible for compliance with the orders of its state public service commission regarding ELG Upgrades and associated costs, including maintenance capital and landfill capital expense, incurred to keep the Mitchell Plant operating after the date the Mitchell Plant would otherwise be required to retire under the ELG Rule in the absence of the ELG Upgrades, including KPSC Case Nos. 2021-00004 and 2021-00421, with respect to KPCo and WVPSC Case Nos. 20-1040-E-CN and 21-0810-E-PC with respect to WPCo.
- The interpretation and performance of this Agreement is governed by and shall be construed in accordance with the laws of the State of New York, exclusive of the conflicts of law provisions thereof that would require the application of the laws of a different jurisdiction. Each Owner hereby agrees that any Action arising out of or relating to this Agreement brought by an Owner (or any of their respective successors or assigns) shall be brought and determined in any state or federal court sitting in the State of New York, within the Borough of Manhattan, City of New York, and the Owners hereby irrevocably submit to the exclusive jurisdiction of the aforesaid courts for themselves and with respect to their property, generally and unconditionally, with regard to any such Action arising out of or relating to this Agreement and the transactions contemplated hereby, and the appellate courts from any thereof in connection with any action arising out of or relating to this Agreement or any other agreement related to the Mitchell Plant or any Project Asset and the transactions contemplated hereby, and consents that any such action may be brought in such courts and waives any objection it may now or hereafter have to the venue of any such action in any such court or that such action was brought in an inconvenient court. EACH OWNER HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ALL RIGHTS TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE O&M AGREEMENT, OR ANY OTHER AGREEMENT RELATED TO THE MITCHELL PLANT OR ANY PROJECT ASSET.
- 13.4 This Agreement supersedes all previous representations, understandings, negotiations, and agreements, either written or oral between the signatories hereto or their representatives with respect to operation of the Mitchell Plant, including the Original Operating

Agreement. Notwithstanding the foregoing, the amendment and restatement of the Original Operating Agreement effected hereby shall not relieve any party thereto of any undischarged obligation or liability of such party in respect of the period prior to the Effective Date under the Original Operating Agreement. This Agreement, together with the O&M Agreement (and any replacements thereof), constitutes the entire agreement of the signatories hereto with respect to the operation of the Mitchell Plant and the ownership thereof. The signatories hereto hereby agree that this Agreement shall amend the Original Operating Agreement to irrevocably remove AEPSC as a party thereto and, on and after the Effective Date, AEPSC shall no longer be a party thereto or hereto or entitled to rights, or subject to obligations, as a party to this Agreement; provided, however, that Operator shall be permitted to delegate any of its rights, duties and obligations under this Agreement and the O&M Agreement to AEPSC without the consent of KPCo, but without relieving Operator of any of its obligations hereunder.

- 13.5 No amendments or modifications of this Agreement are valid unless in writing and signed by duly authorized representatives of the Owners.
- 13.6 Each Owner shall designate in writing a representative to receive any and all notices required under this Agreement. Notices shall be in writing and shall be given to the representative designated to receive them, either by personal delivery, certified mail, e-mail or any similar means, properly addressed to such representative at the address specified below:

# KENTUCKY POWER COMPANY [\_\_\_\_]\_\_\_ Attn:\_\_\_\_ Phone: [\_\_\_\_]\_\_ Email:[]\_\_\_\_ WHEELING POWER COMPANY [\_\_\_\_]\_\_ Attn:\_\_\_\_ Phone: [\_\_\_\_]\_\_ Email:[]\_\_\_\_

All notices shall be deemed to have been given (a) when personally delivered, (b) when transmitted (except if not a Business Day then the next Business Day) via electronic mail (provided

that no error message or other notification of non-delivery is generated with respect to the intended recipient), (c) the day following the day (except if not a Business Day then the next Business Day) on which the same has been delivered prepaid to a reputable national overnight air courier service or (d) the third Business Day following the day on which the same is sent by certified or registered mail, postage prepaid, in each case to the respective parties hereto at the address set forth below, or at such other address as such Owner may specify by written notice to the other Owner (or at such other address for an Owner as shall be specified in a notice given in accordance with this Section 13.6). Each Owner may, by written notice to the other Owner, change the representative or the address to which such notices are to be sent.

- 13.7 This Agreement may be executed in any number of counterparts, and each such counterpart hereof shall be deemed to be an original instrument, but all of such counterparts shall constitute for all purposes one agreement. Any signature hereto delivered by a party hereto by facsimile or other electronic transmission shall be deemed an original signature hereto.
- 13.8 Except as otherwise specifically provided, all fees, costs and expenses incurred by the parties hereto in negotiating this Agreement shall be paid by the party incurring the same, including legal and accounting fees, costs and expenses.
- 13.9 Any of the terms, covenants, or conditions hereof may be waived only by a written instrument executed by or on behalf of the Owners waiving compliance. No course of dealing on the part of any Owners, or its respective officers, employees, agents, accountants, attorneys, investment bankers, consultants or other authorized representatives, nor any failure by an Owner to exercise any of its rights under this Agreement shall operate as a waiver thereof or affect in any way the right of such Owner at a later time to enforce the performance of such provision. No waiver by any Owner of any condition, or any breach of any term or covenant contained in this Agreement, in any one or more instances, shall be deemed to be or construed as a further or continuing waiver of any such condition or breach or a waiver of any other condition or of any breach of any other term or covenant. The rights of the Owners under this Agreement shall be cumulative, and the exercise or partial exercise of any such right shall not preclude the exercise of any other right.
- 13.10 This Agreement shall be binding upon and inure to the benefit of the Owners and their respective successors and permitted assigns.
- 13.11 No Owner will issue, or permit any of its Affiliates, its or its Affiliate's directors, officers, employees, consultants, agents or other representatives to issue, any press releases or otherwise make, or cause any of its Affiliates, its or its Affiliate's directors, officers, employees, consultants, agents or other representatives to make, any public statements or other public disclosures with respect to this Agreement, or the transactions contemplated hereby without the prior written consent of the other Owner; provided, however, that the foregoing requirement to obtain prior written consent shall not apply where such release, statement or disclosure is deemed in good faith by the releasing or disclosing Owner to be required by Applicable Law or under the rules and regulations of a recognized stock exchange on which shares of such Owner (or any of its Affiliates) are listed, so long as prior to making any such release, statement or disclosure and to the extent legally permitted, the releasing or disclosing Owner shall provide prompt notice to the other Owner, consult the other Owner as to the form, contents and timing of such release or

disclosure and, when available, provide a copy of such release, statement or disclosure containing such information to the other Owner.

- 13.12 If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Owners shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Owners as closely as possible in an acceptable manner to the end that the transactions contemplated hereby are fulfilled to the extent possible.
- 13.13 Each Owner acknowledges that it shall be inadequate or impossible, or both, to measure in money the damage to the Members if any of them or any transferee or any legal representative of any Owner fails to comply with any of the restrictions or obligations imposed by Article Nine that every such restriction and obligation is material, and that in the event of any such failure, the Owners shall not have an adequate remedy at law or in damages. Therefore, each Owner consents to the issuance of an injunction or the enforcement of other equitable remedies against such Owner at the suit of an aggrieved party without the posting of any bond or other security, to compel specific performance of all of the terms of Article Nine and to prevent any Disposition in contravention of any terms of Article Nine, and waives any defenses thereto, including the defenses of: (i) failure of consideration, (ii) breach of any other provision of this Agreement and (iii) availability of relief in monetary damages.

# ARTICLE FOURTEEN DEFINITIONS

For all purposes of this Agreement (including the preceding sections and recitals), unless otherwise required by the context in which any defined term appears or otherwise defined in the body of this Agreement, capitalized terms have the meanings specified in this Article Fourteen. In this Agreement, unless expressly stated otherwise: (a) reference to any agreement (including this Agreement), document or instrument means such agreement, document or instrument as has been, or may be, amended, supplemented or otherwise modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof; (b) reference to any Applicable Law means such Applicable Law as has been, or may be, amended, modified, codified or reenacted, in whole or in part, and in effect from time to time, including rules and regulations, promulgated thereunder; (c) the singular includes the plural, as the context requires; (d) the terms "includes" and "including" mean "including, but not limited to"; (e) "Day" (regardless of capitalization) shall mean a calendar month; (vii) references to articles, sections and appendices mean the articles and sections of, and appendices to, this Agreement.

"AEPSC" shall have the meaning given to such term in the Preamble. "Agreement" shall have the meaning given to such term in the Preamble.

"Applicable Law" shall mean all laws (including common law), statutes, codes, acts, treaties, ordinances, orders, judgments, writs, decrees, injunctions, rules, regulations, governmental approvals, permits, directives, and requirements of all Governmental Authorities (including with respect to the environment) having jurisdiction over an Owner, any other person or entity (as to that person or entity), this Agreement, any Project Asset or the Mitchell Plant, as applicable

"Assigned Capacity" shall have the meaning given to such term in Section 2.3.

"Capital Budget" shall have the meaning given to such term in Section 1.7.

"CertainTeed Contract" shall mean that certain Supply Agreement dated March 11, 2005, by and between CertainTeed Gypsum West Virginia, Inc. (formerly BPB West Virginia Inc.) and KPCo (as assignee of Ohio Power Company), as amended by Amendment No. 2010-1 dated August 2, 2010, as further amended by Amendment No. 2012-1 dated February 20, 2012 and as further amended by Amendment No. 2013-1 dated June 5, 2013, as may be amended, amended and restated, supplemented or modified from time to time, and as may be assigned to Operator or an Affiliate of Operator.

"CCR Capital Expenditures" shall mean all capital expenditures associated with implementation of the CCR Upgrades.

"CCR Rule" means the Coal Combustion Residuals Rule, 40 CFR Part 257 (April 17, 2015, as amended), and any regulations thereunder promulgated by the USEPA or the State of West Virginia.

"CCR Upgrades" shall mean any improvements or upgrades to the Mitchell Plant to enable KPCo and WPCo to comply with the CCR Rule.

"Control" shall have the meaning given to such term in Section 7.9.

"Credit Rating" means with respect to any entity, the rating then assigned to such entity's unsecured, senior long-term debt or deposit obligations (not supported by third party credit enhancements) by S&P or Moody's. If no rating is assigned to such entity's unsecured, senior long-term debt or deposit obligations by S&P or Moody's, then "Credit Rating" means the general corporate credit rating or long-term issuer rating assigned to such entity by S&P or Moody's. If an entity is rated by both S&P and Moody's and the ratings are at different levels, then "Credit Rating" means the lowest such rating.

"Decommission" or "Decommissioning" shall mean the retirement, dismantlement and permanent removal of the Units and other property, plant, and equipment comprising the Mitchell Plant, including any common facilities associated with each Unit that are to be permanently removed from service, the restoration of the Mitchell Plant site and the removal or remediation of any hazardous materials or other contaminated equipment, materials, coal ash or wastes associated therewith, in a manner that meets the requirements of Applicable Law.

"<u>Decommissioning Costs</u>" shall mean all costs and obligations expended or incurred in the performance of all work reasonably necessary or undertaken to Decommission the Mitchell Plant,

including work associated with the preparation and implementation of Decommissioning plans and the preparation, submittal and prosecution of all necessary applications with Governmental Authorities as required to Decommission the Mitchell Plant in accordance with Applicable Law.

"Defaulting Owner" shall have the meaning given to such term in Section 10.1.

"Depreciable Life" means, with respect to a capital item, the shorter of (a) the reasonably expected depreciable life (in months) of such capital item and (b) the number of months between the anticipated in-service date of such capital item and December 31, 2040 (or such earlier anticipated date of the permanent cessation of operations of the Units filed with the WVPSC).

"Dispose" or "Disposition" shall have the meaning given to such term in Section 9.1.

Effective Date" shall have the meaning given to such term in the Preamble.

"ELG Capital Expenditures" shall mean all capital expenditures associated with implementation of the ELG Upgrades.

"ELG Rule" shall mean the Steam Electric Reconsideration Rule, 85 Fed. Reg. 64,650 (Oct. 13, 2020), and any regulations thereunder promulgated by the USEPA or the State of West Virginia.

"ELG Upgrades" shall mean any improvements or upgrades to the Mitchell Plant to enable WPCo to comply with the ELG Rule.

"Emission Allowances" shall have the meaning given to such term in Section 7.7.

"Encumbrance" shall mean with respect to any property or asset (a) any mortgage, deed of trust, charge, lien, pledge, hypothecation, title retention arrangement or other security interest, as or in effect as security for the payment of a monetary obligation or the observance of any other obligation; (b) any easement, servitude, restrictive covenant, equity or interest in the nature of an encumbrance, garnishee order, writ of execution, right of set-off, lease, license to use or occupy, assignment of income or monetary claim, whether or not filed, recorded or otherwise perfected under Applicable Law; and (c) any agreement to create any of the foregoing or allow any of the foregoing to exist.

"Event of Default" shall have the meaning given to such term in Section 10.1.

"FERC" shall have the meaning given to such term in Section 5.1.

"FERC Accounting Requirements" means the accounting requirements of FERC, including with respect to the Uniform System of Accounts, established by FERC under the FPA.

"FPA" means the Federal Power Act.

"Governmental Authority" means any federal, national, regional, state, municipal or local government authority, tribunal, court, agency, body, board or instrumentality, or any regulatory, administrative or other department, commission, bureau or agency, taxing authority or power, or

any political or other subdivision, department or branch of the foregoing, including any independent system operator, regional transmission organization or electric reliability organization.

"KPCo" shall have the meaning given to such term in the Preamble.

"KPSC" shall mean the Kentucky Public Service Commission.

"<u>Mitchell Plant</u>" shall mean the Mitchell Power Generation Facility, which consists of the Units and associated plant, equipment, real estate and other related facilities, located in Moundsville, West Virginia, but excluding the real property and operation known as the Conner Run Fly Ash Impoundment and Dam.

"Moody's" shall have the meaning given to such term in <u>Section 4.3</u>.

"Non-Defaulting Owner" shall have the meaning given to such term in Section 10.1.

"Non-Offering Owner" shall have the meaning given to such term in Section 9.1.

"O&M Agreement" shall have the meaning given to such term in the Recitals.

"Offering Owner" shall have the meaning given to such term in Section 9.1.

"Operating Committee" shall have the meaning given to such term in Section 7.1.

"Operating. Representative" shall have the meaning given to such term in Section 7.1.

"Operator" shall have the meaning given to such term in the Recitals.

"Original Operating Agreement" shall have the meaning given to such term in the Recitals.

"Owner" or "Owners" shall have the meaning given to such term in the Preamble.

"Ownership Interest" shall have the meaning given to such term in the Recitals.

"Paid Amount" shall have the meaning given to such term in Section 10.2.

"Payment Default" shall have the meaning given to such term in Section 10.1(a).

"Project Assets" shall have the meaning given to such term in Section 1.1.

"Proposed Purchaser" shall have the meaning given to such term in <u>Section 9.1</u>.

"Ratings Requirement" shall mean a Credit Rating for such Owner (or if such Owner has provided a guaranty issued by an Affiliate to satisfy its obligations under this Section 4.3, such Owner's Affiliate guarantor) of at least "BBB-" by S&P or at least Baa3 by Moody's, and if such Credit Rating is "BBB-" by S&P or "Baa3" by Moody's then such Credit Rating must not be on negative credit watch by S&P or Moody's.

"S&P" shall have the meaning given to such term in Section 4.3.

"<u>Tax Code</u>" shall have the meaning given to such term in <u>Section 6.6</u>.

"<u>Technical Dispute</u>" shall mean any dispute which this Agreement expressly provides shall be a Technical Dispute.

"Technical Expert" shall mean any individual selected in accordance with the procedure specified in Section 12.3 and who (a) has significant professional qualifications and practical experience in the subject matter of the Technical Dispute, (b) has no interest, financial or otherwise, or duty which conflicts or may conflict with such individual's functions as a Technical Expert (such individual being required to fully disclose any such interest or duty prior to any appointment) and (c) is not currently and has not been (i) during the five (5) years prior to the date of appointment, an employee of any of the Owners or any of their Affiliates and (ii) during the three (3) years prior to the date of appointment, a contractor or consultant of either of the Owners or any of their Affiliates, unless otherwise mutually agreed by the Owners.

"Term" shall have the meaning given to such term in Section 8.2.

"Total Net Capability" shall have the meaning given to such term in <u>Section 2.1</u>.

"Total Net Generation" shall have the meaning given to such term in Section 2.2.

"<u>Unit</u>" shall have the meaning given to such term in the Recitals.

"USEPA" shall have the meaning given to such term in Section 7.7.

"WACC" shall mean, as of any date, WPCo's then-applicable WVPSC-authorized weighted average cost of capital, compounded semiannually (consistent with the compounding of Allowance for Funds Used During Construction (AFUDC)).

"<u>WPCo</u>" shall have the meaning given to such term in the Preamble. "<u>WVPSC</u>" shall mean the Public Service Commission of West Virginia

[Signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their officers thereunto duly authorized as of the date first above written.

By:	
Title:	
WHEELING POWER COMPANY	
By:	
Title:	

KENTUCKY POWER COMPANY

Solely with respect to <u>Section 13.4</u> :
AMERICAN ELECTRIC POWER SERVIC CORPORATION
By:
Title:

### Exhibit A

### Capital Budget, Initial Budgets and Forecast

[To Be Attached as of the Effective Date.]

### Exhibit B

### Form of Monthly Sample Report

[To Be Attached as of the Effective Date.]

# [RATE SCHEDULE NO. 303] MITCHELL PLANT OWNERSHIP AGREEMENT KENTUCKY POWER COMPANY

and

WHEELING POWER COMPANY

### TABLE OF CONTENTS

	Page
ARTICLE ONE OWNERSHIP AND OPERATIONS	2
ARTICLE TWO APPORTIONMENT OF CAPACITY AND ENERGY	3
ARTICLE THREE REPLACEMENTS, ADDITIONS, AND RETIREMENTS	4
ARTICLE FOUR WORKING CAPITAL REQUIREMENTS	5
ARTICLE FIVE INVESTMENT IN FUEL	6
ARTICLE SIX APPORTIONMENT OF STATION COSTS	6
ARTICLE SEVEN OPERATING COMMITTEE AND OPERATIONS	
ARTICLE EIGHT EFFECTIVE DATE AND TERM	14
ARTICLE NINE TRANSFERS	14
ARTICLE TEN DEFAULTS AND REMEDIES	16
ARTICLE ELEVEN LIMITATION OF LIABILITY	
ARTICLE TWELVE DISPUTE RESOLUTION	18
ARTICLE THIRTEEN GENERAL	20
ARTICLE FOURTEEN DEFINITIONS	23

Exhibit A — Capital Budget, Initial Budgets and Forecast

Exhibit B — Form of Monthly Sample Report

THIS MITCHELL PLANT OWNERSHIP AGREEMENT (this "<u>Agreement</u>"), with an effective date of [] (the "<u>Effective Date</u>"), is by and among Kentucky Power Company, a Kentucky corporation qualified as a foreign corporation in West Virginia ("<u>KPCo</u>"); Wheeling Power Company, a West Virginia corporation ("<u>WPCo</u>") (such parties hereinafter sometimes referred to as an "<u>Owner</u>" and together the "<u>Owners</u>"); and, solely with respect to <u>Section 13.4</u>, American Electric Power Service Corporation, a New York corporation ("<u>AEPSC</u>").

### WITNESSETH:

WHEREAS, KPCo and WPCo, as of the date hereof, each own a fifty percent (50%) undivided ownership interest in the Mitchell Power Generation Facility (each such percentage interest, an Owner's "Ownership Interest"), which consists of two coal-fired generating units (each, a "Unit"), with each Unit having a nominal nameplate capacity of 800 MW, located in Moundsville, West Virginia (as further defined herein, the "Mitchell Plant");

WHEREAS, KPCo, WPCo and AEPSC are parties to that certain Mitchell Plant Operating Agreement, dated as of December 31, 2014 (the "Original Operating Agreement");

WHEREAS, the Original Operating Agreement sets forth certain rights and obligations of the Owners and AEPSC with respect to the Mitchell Plant and the Owners' ownership thereof;

WHEREAS, pursuant to the Original Operating Agreement, KPCo is responsible for the day-to-day operations and maintenance of the Mitchell Plant;

WHEREAS, the Owners and AEPSC desire to replace the Original Operating Agreement to set forth the rights and obligations of the Owners with respect to the Mitchell Plant and their ownership thereof and to remove AEPSC as a party thereto;

WHEREAS, in connection with the execution of this Agreement, the Owners desire to execute a separate operations and management agreement to provide for the day-to-day operation and maintenance responsibilities in respect of the Mitchell Plant (as may be amended from time to time the "O&M Agreement"); and

WHEREAS, the Owners have agreed that, subject to the terms and conditions of the O&M Agreement, on and after the Effective Date WPCo shall replace KPCo as the operator of the Mitchell Plant (the "Operator").

NOW THEREFORE, in consideration of the premises and for the purposes hereinabove recited, and in consideration of the mutual covenants hereinafter contained, the signatories hereto agree as follows:

### ARTICLE ONE

### OWNERSHIP AND OPERATIONS

- To the greatest extent permitted by Applicable Law, the Mitchell Plant and all assets (tangible and intangible) and property (real and personal) owned, leased, held, developed, constructed or acquired solely for or in connection with the Mitchell Plant or the operation, maintenance or Decommissioning of the Mitchell Plant by or on behalf of an Owner or the Owners (together, the "Project Assets") shall be owned and held and deemed to be owned and held by the Owners as tenants in common in proportion to their respective Ownership Interests (except for any capital items owned in a different proportion in accordance with Section 1.8) or, in the event any Project Asset cannot be held directly by both of the Owners due to, inter alia, any pre-existing legal or contractual restrictions that cannot be altered or satisfied or where effectuating such ownership structure would result in unreasonable additional expense to the Owners, by the Operator as trustee for the Owners as tenants in common in proportion to their respective Ownership Interest. If the ownership of any Project Asset is registered or recorded in the name of one of the Owners, and notwithstanding the Owners' efforts such Project Asset cannot be held directly by both Owners as contemplated above, then such Owner in whose name ownership is registered or recorded shall hold such Project Asset in trust for itself and the other Owner in proportion to their respective Ownership Interests and, to the extent necessary or requested by the Operator or other Owner, make such Project Assets (or the benefits thereof) available for the use and benefit of the Owners (in proportion with their respective Ownership Interests), including, to the extent consistent with the foregoing, by such Owner subcontracting, sublicensing, subleasing, delegating or granting a limited power of attorney or similar appointment as agent to Operator to administer such Project Assets.
- 1.2 At the request of either Owner, and in accordance with <u>Section 1.1</u>, each Owner and the Operator shall execute all documents and do all things necessary or appropriate to register or record the Project Assets in the names of the Owners in proportion to their respective Ownership Interests (or such different proportion as any capital item may be owned in accordance with <u>Section 1.8</u>).
- 1.3 All assets (tangible and intangible) and property (real and personal) held, developed, constructed or acquired by or on behalf of the Operator for or on behalf of the Owners jointly, or any of them, shall constitute "Project Assets" subject to the ownership of both Owners as set forth in Sections 1.1 and 1.2. Except as otherwise agreed by the Owners, the Operator shall not have any right, title or interest in or to any such assets, or in or to any money paid to, collected or received by the Operator for or on behalf of either Owner, except as the agent or representative of, or for the use and benefit of, such Owners as set forth in this Agreement and in proportion to each Owner's respective Ownership Interest.
- 1.4 Each Owner hereby waives any rights it may have at law or equity to bring an action for partition or division of the Mitchell Plant or any Project Asset or any contracts related thereto, and agrees that it shall not (a) seek partition or division of the Mitchell Plant or any

Project Asset or any contracts related thereto, or (b) take any action, whether by way of any court order or otherwise, for physical partition or judicial sale in lieu of partition of the Mitchell Plant or any Project Asset or any contracts related thereto. Nothing in this Section 1.4 shall affect the right of either Owner to dispatch its respective share of the Total Net Capability under Article Two or to Dispose of its Ownership Interest in accordance with Article Nine.

- 1.5 On and after the Effective Date, WPCo shall be the Operator responsible for the day-to-day operations and maintenance of the Mitchell Plant and shall operate, maintain and Decommission the Mitchell Plant for the sole benefit (and on behalf) of the Owners and in accordance with the terms and conditions of this Agreement and the O&M Agreement. KPCo agrees to take all actions reasonably necessary to facilitate WPCo's operation, maintenance and Decommissioning of the Mitchell Plant pursuant to the terms of the O&M Agreement, including providing or permitting reasonable access to the Mitchell Plant to third party contractors and other contract counterparties of each Owner or the Operator with respect to the administration, implementation and satisfaction of such contracts or agreements executed or assumed by the Operator on behalf of either Owner relating to the Mitchell Plant, including all Facility Agreements (as defined in the O&M Agreement).
- 1.6 The Owners shall establish and maintain such bank accounts as may from time to time be required or appropriate for paying the costs and expenses, including capital expenditures, in respect of the ownership, operation, maintenance and Decommissioning of the Mitchell Plant. The Owners shall designate only the Operator, and its representatives as reasonably requested by the Operator, as authorized signatories to such bank accounts. All withdrawals made by the Operator (or its representatives) from such bank accounts shall be made only in connection with the performance of the Operator's obligations set forth in this Agreement and the O&M Agreement.
- 1.7 The initial capital budget for the period from the Effective Date through December 31, 2028 (including agreed allocations of costs for capital projects between the Owners) (the "Capital Budget"), the initial annual operating budget and the initial forecast of operating and capital costs to be incurred for the period from the Effective Date through December 31, 2028 are attached hereto as Exhibit A.
- 1.8 Notwithstanding the provisions of this Article One, to the extent that either Owner funds or bears an amount greater than 50% of any capital expenditures or ELG Capital Expenditures as contemplated in the Capital Budget or this Agreement, the directly resulting portion of any property, plant and equipment, or improvements thereto shall be owned by the Owners in proportion to their respective amounts funded and shall be included only in such proportion in each Owner's ownership accounts for regulatory, accounting, tax and other purposes.

### ARTICLE TWO

### APPORTIONMENT OF CAPACITY AND ENERGY

- 2.1 The total net capability of the Mitchell Plant at low-voltage busses of the Units, after taking into account auxiliary load demand, is 1,560,000 kilowatts (the "<u>Total Net Capability</u>") as of the Effective Date. The Owners may from time to time modify the Total Net Capability of the Mitchell Plant as they may mutually agree.
- 2.2 The total net generation of the Mitchell Plant during a given period, as determined by the requirements of each Owner, shall mean the electrical output of the Mitchell Plant generators during such period, measured in kilowatt hours by suitable instruments, reduced by the energy used by auxiliaries for each Unit during such period (the "Total Net Generation").
- 2.3 Each Owner shall be entitled to receive 50% of the Total Net Capability and the Total Net Generation (with respect to each Owner, such Owner's "Assigned Capacity"), and all associated energy, capacity, ancillary services and other energy products, in accordance with this Agreement. Provided that such percentage shall be adjusted in accordance with Section 7.9 of the O&M Agreement.
- 2.4 Except as may be determined by the Operating Committee in accordance with Section 7.6, in any hour, each Owner shall share 50% of the minimum load responsibility of each Unit.
- 2.5 In any hour during which any Unit is out of service, the Owners shall bear equally the cost of energy used by the out-of-service Unit's auxiliaries during such hour, which may be provided by the applicable local utility Affiliate of an Owner. Alternatively, the Owners may mutually agree in writing to each provide 50% of such energy.

### ARTICLE THREE

### REPLACEMENTS, ADDITIONS, AND RETIREMENTS

- 3.1 The Owners shall take all actions within their respective control to cause the Operator, pursuant to the O&M Agreement, from time to time to make or cause to be made any necessary or appropriate additions to, replacements of, and retirements of, capitalizable facilities associated with the Mitchell Plant in accordance with the Capital Budget and the O&M Agreement or as may otherwise be mutually agreed upon by the Owners.
- 3.2 Each Owner shall, upon the retirement of the Mitchell Plant (or any individual Unit), (a) cooperate in good faith and take all actions reasonably necessary to facilitate the relevant Decommissioning thereof, including negotiating in good faith any contracts or agreements (including liability transfer arrangements) on behalf of either Owner or Operator,

including transfers, conveyances or assignments of Facility Equipment (as defined in the O&M Agreement), as reasonably requested by either Owner or Operator to facilitate such Decommissioning and (b) take, and/or instruct the Operator pursuant to the O&M Agreement to take, such actions, at the sole cost and expense of WPCo, to continue operating and maintaining the barge loading facilities and gypsum conveyor system at the Mitchell Plant and providing use of such facilities and system to the applicable contract counterparty and its representatives in accordance with, and until the expiration or earlier termination of, the CertainTeed Contract.

### ARTICLE FOUR

### WORKING CAPITAL REQUIREMENTS

- 4.1 The Owners shall periodically mutually determine the amount, timing and invoicing processes for funds required for use as working capital, for operating, capital and other expenses incurred in the operation, maintenance and Decommissioning (including the Decommissioning Costs) of the Mitchell Plant, and in buying equipment, materials, parts, fuel and other supplies and services necessary to operate, maintain and Decommission the Mitchell Plant and to make the timely payments of any expenses required under the O&M Agreement.
- 4.2 Each Owner shall, in accordance with the timing set forth in a determination made pursuant to Section 4.1, promptly provide 50% of any such amount required by the Owners pursuant to Section 4.1, except as otherwise provided for in Section 6.7.
- Each Owner agrees that if such Owner fails at any time during the Term to satisfy the Ratings Requirement, it will, within thirty (30) days of such failure, provide in favor of the other Owner and maintain credit support in the form of (a) a cash deposit, (b) a guaranty issued by an Affiliate of such Owner that satisfies the Ratings Requirement in form and substance reasonably acceptable to the other Owner or (c) a letter of credit in form and substance reasonably acceptable to other Owner, issued by a commercial bank or other financial institution with a Credit Rating of at least "A-" by S&P Global Ratings, or any successor thereto ("S&P") or at least "A3" by Moody's Investors Service, Inc., or any successor thereto ("Moody's"), and in an amount equal to (i) one-half (1/2) of the then-applicable annual operating budget for the Mitchell Plant established pursuant to Section 7.2 from time to time, plus (ii) the sum of such Owner's allocated amount of capital expenditures for such year contained in the then-applicable Capital Budget, plus (iii) an amount equal to the latest estimate of Decommissioning Costs prepared by the Operator, determined on a net present value basis using a discount rate equal to the WACC as of the date of determination. Such credit support posted in favor of an Owner shall be promptly returned within thirty (30) days of the other Owner furnishing written evidence demonstrating that it satisfies the Ratings Requirement.
- 4.4 The Operator shall provide such credit support, including guarantees, cash deposits, letters of credit or other forms of credit support, to third parties (including contractual counterparties and Governmental Authorities) as required for the Owners' ownership, operation, maintenance and Decommissioning of the Mitchell Plant. To the extent that the Operator is required to provide such credit support to a third party in connection with any activity performed

in respect of the Mitchell Plant under this Agreement (including the procurement of fuel as described in Section 5.1), the Owners shall share the reasonable and documented out-of-pocket cost of the third-party credit support incurred by the Operator (including of any credit support furnished by an Affiliate of the Operator) in accordance with their respective Ownership Interests.

### **ARTICLE FIVE**

### INVESTMENT IN FUEL

- 5.1 The Operator shall procure, establish and maintain reserves of coal in common stock piles for the Mitchell Plant of such quality and in such quantities as the Operating Committee shall determine to be required to provide adequate fuel reserves against interruptions of normal fuel supply and as is necessary to maintain the number of tons in such coal stock piles, after taking into account the coal consumption from such coal stock piles by each Unit during each month. For purposes of this Agreement, "consumables" shall be as defined in account 502 of the Uniform System of Accounts administered by the Federal Energy Regulatory Commission ("FERC").
- 5.2 The quality of any coal or consumable product provided by the Operator must be reasonably acceptable to both Owners. Any coal being utilized shall be deemed to be acceptable to the Owners if it meets the following requirements: (a) coal previously utilized at the Mitchell Plant with satisfactory operating performance shall be considered acceptable for use in the Mitchell Plant, unless deemed unacceptable due to a required change of the engineering specifications making the coal no longer viable; (b) coal from any new seam or source shall be acceptable if such supply is shown to perform satisfactorily in the Mitchell Plant and is mutually acceptable to each Owner; or (c) as otherwise mutually agreed to by each Owner. Consumables from any new seam or source shall be acceptable if such supply is shown to perform satisfactorily to both Owners in the Mitchell Plant and conform to the then current engineering specifications for the Mitchell Plant or as otherwise mutually agreed by each Owner.
- 5.3 Each Owner shall be responsible for, and own, 50% of the investment in the common coal stock piles or such percentage as adjusted pursuant to Article Five of the O&M Agreement.
- 5.4 Fuel oil and consumables charged to operation for the Mitchell Plant shall be owned and accounted for between the Owners in the same manner as coal.

### ARTICLE SIX

### APPORTIONMENT OF STATION COSTS

- 6.1 The allocation to the Owners of fuel expense associated with each Unit shall be determined by the Operating Committee as follows:
- (a) In any calendar month, the average unit cost of coal available for consumption from the Mitchell Plant common coal stock piles shall be determined based on the prior month's ending inventory dollar and ton balances plus current month receipts delivered to the Mitchell Plant common coal stock piles. Each Owner's average unit-cost will be the same, and receipts and inventory available for consumption amounts will be allocated to each Owner based on monthly usage.
- (b) The number of tons of coal consumed by the Mitchell Plant in each calendar month from the Mitchell Plant common coal stock piles shall be determined and shall be converted into a dollar amount equal to the product of (i) the average cost per ton of coal associated with the Mitchell Plant in the Mitchell Plant common coal stock pile at the close of such month, and (ii) the number of tons of coal consumed by the Mitchell Plant from the Mitchell Plant common coal stock piles during such month. Such dollar amount shall be credited to the Mitchell Plant fuel in the stock pile and charged to the Mitchell Plant fuel consumed.
- (c) In each calendar month, each Owner's respective shares of the Mitchell Plant fuel consumed expense as determined by the provisions of <u>Section 6.1(b)</u> shall be proportionate to each Owner's dispatch of the Mitchell Plant in such month.
- (d) Fuel oil reserves will be owned and accounted for in the same manner as coal stock piles, and fuel oil consumed will be allocated to the Owners in the same manner as coal consumed.
- 6.2 For each calendar month, the Operator will, to the extent practicable and in accordance with the O&M Agreement, determine all of the Mitchell Plant's operations expenses and associated overheads, as accounted for under the FERC Uniform System of Accounts.
- 6.3 For each calendar month, the Operator will, to the extent practicable and in accordance with the O&M Agreement, determine all the Mitchell Plant's maintenance expenses and associated overheads, as accounted for under the FERC Uniform System of Accounts.
- 6.4 In each calendar month, each Owner's respective shares of operations and maintenance expenses associated with the Mitchell Plant, as determined in accordance with this Article Six, shall be allocated as follows:
- (a) Each Owner's respective share of the Mitchell Plant steam expenses as recorded in FERC Account 502, and emission tons, with allowance expenses as recorded in

FERC Account 509, shall be proportionate to each Owner's dispatch of the Mitchell Plant in such month.

- (b) In each calendar month, the maintenance of boiler plant expenses as recorded in FERC Account 512, and maintenance of electric plant expenses as recorded in FERC Account 513, shall be directly assigned to each Unit or designated as a common expense attributable to both Units. In each calendar month, each Owner's respective share of these expenses shall be proportionate to each Owner's dispatch of the applicable Unit, or both Units in the case of common expenses, over the previous sixty (60) calendar months.
- (c) In each calendar month, each Owner shall be responsible for 50% of all other Steam Power Generation Expenses (FERC Accounts 500 515) not addressed in Section 6.4(a) and Section 6.4(b). Administrative and General Expenses (FERC Accounts 920 935) shall be assigned to the Mitchell Plant through an annual wages and salaries allocator applied to monthly Administrative & General Expenses. Each Owner shall be responsible for 50% of this monthly amount; provided, however, that, for the avoidance of doubt, each Owner shall be individually responsible for any fees, costs or other charges, including but not limited to those imposed by PJM Interconnection, L.L.C. ("PJM") or any regional transmission operator or any other Governmental Authority in respect of, or which are attributable to, the sale or transmission of the capacity or energy associated with its Ownership Interest, as the case may be.
- (d) Notwithstanding the foregoing clauses (a) through (c) or anything else in this Agreement or the O&M Agreement to the contrary, in each calendar month, any operations and maintenance or other expenses to the extent attributable to any ELG Upgrade (regardless of the FERC Account to which it is charged) shall be allocated exclusively to and paid by WPCo.
- (e) In each calendar month, each Owner's respective share of Construction Work In Progress charged to FERC Account 107 shall be allocated on the same basis as capital expenditures, as set forth in <u>Section 6.7</u>.
- (f) In each calendar month, the net change in Mitchell Plant storeroom inventory (inventory purchases less issuances of inventory) charged to FERC Account 154 shall be allocated 50% to each Owner.
- (g) Each Owner shall be charged 50% of Operating Costs, as defined in and in accordance with Section 7.2 of the O&M Agreement, except to the extent a different allocation for specific FERC Accounts or otherwise is specified in this Article Six.
- 6.5 All taxes, duties or assessments levied against or with respect to each Owner's Ownership Interest, or an Owner's purchase, use, ownership or beneficial interest in, or income from, the Mitchell Plant shall be the sole responsibility of, and shall be paid by, the Owner upon whose purchase, use, ownership interest or beneficial interest or income said taxes or assessments are levied. Without limiting the foregoing, in each calendar month, each Owner's respective share of Employee Payroll Taxes charged to FERC Account 408 shall be 50%.
- 6.6 Notwithstanding any other provision of this Agreement or any other agreement to the contrary, each Owner hereby acknowledges and agrees that (a) each Owner prior to the Effective Date has treated, and subsequent to such date shall continue to treat, the co-ownership

and operation of the Mitchell Plant as excluded from Subchapter K of the Internal Revenue Code of 1986, as amended (the "<u>Tax Code</u>"), pursuant to Section 761(a) thereof, for all federal, state and local income tax purposes, (b) each Owner prior to the Effective Date affirmatively elected not to apply any of the provisions of Subchapter K of the Tax Code to such Owner's interest in the Mitchell Plant, with such election having been formally filed in connection with the Owners' applicable income tax returns for the taxable year ending on December 31, 2020 and each Owner has taken all actions necessary to implement such election and (c) each Owner prior to the Effective Date has reported, and subsequent to such date shall report, its share of all income, gains, deductions, losses, credits, etc. from its Ownership Interest on its tax returns consistent with such exclusion from the provisions of Subchapter K of the Tax Code.

- 6.7 Subject to <u>clauses (b)</u> and <u>(c)</u> below the cost of any replacement, addition, improvement or upgrade of each Unit or any portion of the Mitchell Plant, and any restoration or remediation required in connection therewith, shall be allocated between the Owners in accordance with the allocations for such capital items contained in the Capital Budget. With respect to any such capital item not contained in the Capital Budget, the costs of such capital item shall be allocated as follows, unless the Operating Committee agrees upon a different allocation:
- (a) Capital expenditures (other than ELG Capital Expenditures) that the Operating Committee determines have been or will be incurred exclusively for one Owner shall be allocated exclusively to, and paid for by, that Owner.
- (b) Notwithstanding anything to the contrary herein, ELG Capital Expenditures shall be allocated exclusively to, and paid for exclusively by, WPCo and CCR Capital Expenditures shall be allocated 50% to (and paid for by) each Owner; provided, that, the Operating Committee shall engage or retain a Technical Expert to make recommendations with respect to determining which capital expenditures are ELG Capital Expenditures and which capital expenditures are CCR Capital Expenditures.
- (c) Notwithstanding anything to the contrary herein, if the in-service date of a capital item is reasonably anticipated by the Operating Committee to be after December 31, 2028, then the capital expenditures for such capital item shall be allocated exclusively to, and paid for by, WPCo.
- (d) If the Operating Committee determines, including based on Depreciable Lives of similar assets previously approved by applicable Governmental Authorities, that a capital item (other than an ELG Upgrade) has a Depreciable Life that extends beyond December 31, 2028, then (i) KPCo shall be responsible for and shall pay 50% of the expenditures for such capital item, multiplied by (A) the number of months (not to exceed the Depreciable Life of such capital item) between the reasonably anticipated in-service date of such capital item and December 31, 2028, divided by (B) the Depreciable Life of such capital item and (ii) WPCo shall be responsible for the remaining amount of such capital expenditure not allocated to KPCo pursuant to the foregoing clause (i).

- (e) Any other capital expenditures shall be allocated 50% to (and paid for by) each Owner, subject to the written approval of the Operating Committee for budget overruns to the extent required pursuant to Section 5.3.2 of the O&M Agreement.
- 6.8 Each Owner shall be responsible for 50% of all Decommissioning Costs, unless a different allocation is expressly specified for such item in the Capital Budget (as agreed by the Owners) or the Owners mutually agree to allocate such costs in another manner.
- 6.9 Notwithstanding anything contained in this Agreement, an Owner's obligation to pay its obligations under this Agreement shall not in any way be conditioned upon or affected by any regulatory order or other determination disallowing, limiting or deferring rate recovery of the costs and expenses paid or payable by an Owner in respect of its Ownership Interest.

### ARTICLE SEVEN

### **OPERATING COMMITTEE AND OPERATIONS**

- 7.1 By written notice to each other, each Owner shall name one representative (the "Operating Representative") and one alternate to act for it in matters pertaining to operating arrangements under this Agreement and the O&M Agreement. An Owner may change its Operating Representative or alternate at any time by written notice to the other Owner. The Operating Representatives for the respective Owners, or their alternates, shall comprise the "Operating Committee". All decisions, directives, or other actions by the Operating Committee must be by unanimous agreement of the Operating Representatives of the Owners. If the Operating Representatives are unable to agree on any matter, such matter will be resolved through the dispute resolution procedures set forth in Article Twelve.
- 7.2 The Operating Committee shall have the following responsibilities, which decisions are reserved exclusively for the Operating Committee and may not be made individually by the Operator or any Owner:
- (a) Review and approval of any amendments to the Capital Budget, and adoption of an annual operating budget, annual operating plan and a six-year forecast of operating and capital expenses, each as delivered to the Operating Committee by the Operator pursuant to Section 7.8, including determination of the emission allowances required to be acquired by each Owner with respect to their Ownership Interests; provided, that an Owner's Operating Representative shall have the right to amend the Capital Budget solely to include any capital expenditures for which such Owner shall be allocated greater than 75% of the costs pursuant to Section 6.7, up to an aggregate amount of such capital expenditures that does not exceed \$3 million per year allocated to the other Owner. Allocations of new capital expenditures added to the Capital Budget shall be consistent with Section 6.7; provided, that if the Operating Committee cannot agree upon the Depreciable Life of a capital item or the allocation of a capital expenditure between the Owners (including determining which capital expenditures), the matter shall be resolved in accordance with the Technical Dispute resolution procedures set forth in Section 12.1 and Section 12.3 and the Owners shall implement any resolution of the Technical

Dispute through adjustments or true-up payments, as appropriate. If the Operating Committee fails to adopt an annual operating budget, the approved annual operating budget from the previous year (other than one-time or other non-recurring or inapplicable items) shall apply until such time as the new annual operating budget is approved.

- (b) Establishment and monitoring of procedures for communication and coordination with respect to the Mitchell Plant capacity availability, fuel-firing options, and scheduling of outages for maintenance, repairs, equipment replacements, scheduled inspections, and other foreseeable cause of outages at the Mitchell Plant, as well as the return the Mitchell Plant to availability following an unplanned outage. The Operating Committee shall use commercially reasonable efforts, consistent with Prudent Operation and Maintenance Practices (as defined in the O&M Agreement), to schedule the implementation of ELG Upgrades during planned maintenance and repair outages so as to eliminate or minimize incremental outages.
- (c) To the extent not included in the Capital Budget, decisions on capital projects, including Unit upgrades and re-powering, except that an Owner's Operating Representative shall have the right to approve any such capital projects for which such Owner shall be allocated greater than 75% of the costs pursuant to Section 6.7 and Section 7.2(a).
- (d) Determinations as to allocations between the Owners of expenses pursuant to Section 6.1.
  - (e) Determinations as to changes in the Unit capability.
- (f) Establishment and modification of billing procedures under this Agreement or under the O&M Agreement.
  - (g) Approval of material contracts for fuel supply or transportation.
- (h) Establishment and modification of specifications of fuels; oversight of fuel procurement, inspection and certification arrangements, policies and procedures; and management of fuel inventories for the Mitchell Plant.
- (i) Establishment of, termination of, and approval of any change or amendment to the operating arrangements (including the O&M Agreement) between the Owners and the Operator (or any successor Operator or replacement third-party Operator) and selection of any replacement Operator, except as otherwise permitted by Section 7.9.
- (j) Review and approval of plans and procedures designed to ensure compliance at the Mitchell Plant with all Applicable Law, including procedures for allocating and using emission allowances or for any programs that permit averaging at more than one Unit for compliance.
- (k) Amendment, termination, extension or modification of the O&M Agreement, and supervision of the performance of, and provision of direction as needed to, the Operator.

- (l) Decisions regarding the retirement, permanent removal from service or Decommissioning of a Unit or any material portion of the Mitchell Plant and any restoration or remediation required in connection therewith.
- (m) Establishment of an insurance program to provide property and general liability insurance on behalf of each Owner, to be procured by the Operator pursuant to the O&M Agreement.
  - (o) Other duties as assigned by agreement of the Owners.
- 7.3 The Operating Committee shall meet at least quarterly, or at such other frequency as determined by the Operating Committee, and at such other times as an Owner may reasonably request. The Operator shall provide operations reports to the Operating Committee each month (presented on a monthly basis) and each quarter (presented on a quarterly basis) substantially in the form of Exhibit B hereto.
- 7.4 The Owners and the Operator shall cooperate in providing to the Operating Committee the information it reasonably needs to carry out its duties, and to supplement or correct such information on a timely basis.
- 7.5 The Owners will each make an initial unit commitment one business day ahead of real-time dispatch.
- 7.6 Application of this <u>Section 7.6</u> (including subsections) is subject to (i) the receipt of any necessary regulatory approvals or waivers expressly granted for this <u>Section 7.6</u>; and (ii the Operating Committee establishing and approving procedures and systems for dispatch. As used in this Section and subsections of this Section, the terms "Party" or "Parties" refers only to KPCo and WPCo, or both of them, as the case may be.
- 7.6.1 If Mitchell Unit 1 or Unit 2 is designated to be committed by both Parties, such unit will be brought on line or kept on line. If neither Party designates Mitchell Unit 1 or Unit 2 to be committed, such unit will remain offline or be taken offline.
- 7.6.2 When a Mitchell Unit is designated to be committed by one Party, but designated not to be committed by the other Party, the unit will be brought on line or kept on line if the Party designating the unit for commitment undertakes to pay any applicable start-up costs for the unit, as well as any applicable minimum running costs for the unit thereafter, in which event the unit shall be brought on line or kept on line, as the case may be. The Party so designating the unit to be committed shall have the right to schedule and dispatch up to all of the Available Capacity of the unit. Available Capacity means that portion of the Owners' aggregate Assigned Capacity that is currently capable of being dispatched. The Party exercising this right shall be referred to as the "Calling Party," and the capacity called by that Party in excess of its Assigned Capacity Percentage of the Available Capacity of that unit shall be referred to as its "Called Capacity." The other Party shall be referred to as the "Non-Calling Party". The Calling Party shall provide reasonable notice to the Non-Calling Party of its call, including any start-up or shut-down time for the Unit. For purposes of this Agreement, KPCo's Assigned Capacity Percentage shall be

50%, and WPCo's Assigned Capacity Percentage shall be 50% as adjusted pursuant to section 7.9 of the O&M Agreement.

- 7.6.3 The Non-Calling Party can reclaim any Called Capacity attributable to its Assigned Capacity share by giving the Calling Party notice equal to the normal cold start-up time for the unit. At the end of the notice period, the Non-Calling Party shall have the right to schedule and dispatch the recalled capacity. At that point, the Non-Calling Party shall resume its responsibility for its share of any applicable start-up costs for the unit and prospectively shall bear its responsibility for the costs associated with its Assigned Capacity from the unit.
- 7.6.4 If any capacity remains available but is not dispatched from a Party's Available Capacity committed as a result of the initial unit commitment, the other Party may schedule and dispatch such capacity pursuant to 7.6.1 through 7.6.3, above.
- Emission Allowances. Prior to the earlier of any Buyout Transaction or December 31, 2028 (or earlier retirement of the Facility), to the extent that emission allowances issued by the U.S. Environmental Protection Agency ("USEPA") pursuant to Title IV of the Clean Air Act Amendments of 1990 and any regulations thereunder, and any other emission allowance trading program created under the Clean Air Act and administered by USEPA or the State of West Virginia, including but not limited to the Cross-State Air Pollution Rule 40 C.F.R. Part 97, and any amendments thereto (the "Emission Allowances"), are required for operation of the Mitchell Plant, each Owner will be entitled to receive for its own benefit 50% of any Emissions Allowances allocated to the Mitchell Plant. Each Owner will be responsible for acquiring any additional Emission Allowances needed to satisfy the Emission Allowances required because of such Owner's dispatch of energy from the Mitchell Plant. Additionally, prior to such time, each Owner will be responsible for acquiring the Emission Allowances required, to the extent necessary in addition to its share of the Emissions Allowances allocated to the Mitchell Plant, to satisfy 50% of the Emission Allowance surrender obligations attributable to the Mitchell Plant imposed under the Consent Decree entered in United States, et al. v. American Electric Power Service Corp., et al., Civil Action Nos. C2-99-1182 and C2-99-1250 and United States, et al. v. American Electric Power Service Corp., et al., Civil Action No. C2 05-360 and Ohio Citizen Action, et al. v. American Electric Power Service Corp., Civil Action No. C2-04-1098 dated December 10, 2007 as subsequently modified or amended, it being understood that the Owners may be subject to additional rights and obligations under any applicable agreement among the Owners (and/or their Affiliates) and American Electric Power Company, Inc. (and/or its Affiliates) pertaining to the allocation of emission limitations associated with the Mitchell Plant. As early as possible, but no later than three business days after the deadline for submitting final electronic data to the EPA for compliance purposes, the Operator shall notify each Owner of the number of annual or seasonal Emission Allowances that are needed to offset each Owner's share of emissions for the previous year or season. Each Owner shall supply its respective share of allowances, with a reasonable compliance margin as determined by the Operating Committee, by transferring the applicable allowances to the Mitchell Plant's Allowance Facility Account on or before 15 days prior to the remittance date. In the event that an Owner fails to surrender the required number of Emission Allowances in accordance with the prior paragraph, the other Owner shall have the option to purchase the required number of Emission Allowances, and the Owner that failed to surrender the required number of Emission Allowances shall reimburse the other Owner for any amounts it shall have incurred to make such purchases, with interest at the

"Federal Funds Rate" (as published by the Board of Governors of the Federal Reserve System as from time to time in effect) running from the date of such purchases to the date of payment. The Operating Committee will develop procedures to be implemented after the end of each calendar year to account for each Owner's share of the Emission Allowances required by the use of the Mitchell Plant and to correct any imbalance between the Emission Allowances supplied and the Emission Allowances used through the end of the preceding year by settlement or payment.

- At least ninety (90) days before the start of each operating year, the Operator shall submit to the Operating Committee any proposed amendments to the Capital Budget and an annual operating budget for such operating year with respect to the Mitchell Plant, a proposed annual operating plan with respect to the Mitchell Plant for such operating year, and a forecast of operating and capital costs to be incurred during the next six-year period. The annual operating budget and amendments to the Capital Budget shall be presented on a month-by-month basis, and shall include an operating budget, a capital budget, and an estimate of the cost of any major repairs or improvements that are anticipated to occur during the relevant period with respect to the Mitchell Plant, and an itemized estimate of all projected fixed and variable operating expenses relating to the operation of the Mitchell Plant during that operating year. The members of the Operating Committee will meet and work in good faith to agree upon the final annual operating budget, final annual operating plan and any amendments to the Capital Budget. Once approved, the annual operating budget and annual operating plan shall remain in effect throughout the applicable operating year, subject to such changes, revisions, amendments, and updating as the Operating Committee may determine. If the Operating Committee determines to retire the Mitchell Plant prior to December 31, 2028, the members of the Operating Committee will meet and work in good faith to amend the Capital Budget to remove any future ELG Capital Expenditures and any other future capital expenditures no longer required, to the extent practicable and consistent with Applicable Law. The Capital Budget shall remain in effect throughout the Term, subject to such amendments as the Operating Committee may determine.
- Notwithstanding anything in this Agreement to the contrary, (i) in the case of the 7.9 O&M Agreement or any other agreement relating to the Mitchell Plant that is entered into jointly by or on behalf of the Owners, on one hand, with an Affiliate of an Owner (or with an Owner itself, as in the case of the O&M Agreement) on the other hand, the non-Affiliate Owner shall have the sole and exclusive right to exercise any and all affirmative or elective rights of the Owners, including remedies (including delivering notices of and pursuing or settling disputes or delivering notices of default or making and pursuing claims for indemnification) and any termination rights (including rights of termination for convenience, if any) thereunder (for the avoidance of doubt, without first obtaining the consent of the other Owner or the Operating Committee) and (ii) in the case the O&M Agreement is terminated pursuant to Section 8.2 thereof, KPCo shall have the sole and exclusive right to select and designate any successor "Operator" or replacement third-party Operator, in each case so long as such successor replacement is a "Qualified Replacement Operator" (as defined in the O&M Agreement); provided, however, that notice of any such action described in this Section 7.9 shall be sent to the other Owner at the time such action is taken if such other Owner is not the Operator. For purposes of this Agreement, "Affiliate" shall mean, with respect to any person or entity, any other person or entity that directly or indirectly, controls, is controlled by, or is under common control with such person or entity. As used in this definition, "control" (including, with its correlative meanings, "controlled by" and "under common control with") means the possession,

directly or indirectly, of the power to direct or cause the direction of the management or policies of a person or entity, whether through the ownership of securities or partnership or other ownership interests, by contract or otherwise.

### ARTICLE EIGHT

### EFFECTIVE DATE AND TERM

- 8.1 This Agreement shall be effective as of the Effective Date.
- 8.2 Subject to FERC approval or acceptance of any termination, if necessary, this Agreement shall remain in force until the date on which this Agreement is terminated by mutual agreement of the Owners (the period from the Effective Date through such date, the "Term").

### ARTICLE NINE

### TRANSFERS

Neither Owner may assign, transfer or otherwise dispose of its Ownership Interest, either in whole or part, whether by sale, lease, division, declaration or creation of a trust, by operation of law or otherwise ("Dispose" or a "Disposition") to any person or entity (the "Proposed Purchaser") without the prior written consent of the other Owner (the "Non-Offering Owner" and the Owner proposing the Disposition, the "Offering Owner"), which consent may be granted or withheld in the Non-Offering Owner's sole discretion. Notwithstanding the foregoing, either Owner may Dispose of, all (but not less than all) of its Ownership Interest to a state regulated utility Affiliate, provided that (i) the Disposition shall not relieve the Offering Owner of its obligations under this Agreement, (ii) the Disposition shall be made in compliance with the Consent Decree entered in United States, et al. v. American Electric Power Service Corp., et al., Civil Action Nos. C2-99-1182 and C2-99-1250 and United States, et al. v. American Electric Power Service Corp., et al., Civil Action Nos. C2-04-1098 and C2-05-360, and all amendments or modifications thereto, as in effect as of the date of the Disposition, (iii) the Proposed Purchaser shall agree to and assume, in respect of the Ownership Interest subject to the Disposition, the rights and obligations of the Offering Owner and its Affiliates under any applicable agreement with American Electric Power Company, Inc. (and/or its Affiliates) pertaining to the allocation of emission limitations associated with the Mitchell Plant, and (iv) in the event the Offering Owner (or any Affiliate thereof) shall be the Operator, the Proposed Purchaser shall also have been assigned, and agreed to have assumed, the rights and obligations of the Operator under this Agreement and the O&M Agreement as of the effective date of such Disposition; provided, that in the case of this clause (iv), a written consent from the Non-Offering Owner (which consent shall not be unreasonably withheld, conditioned or delayed) shall be obtained prior to such Disposition to the extent such Disposition results in the change of the Operator.

- 9.2 No Disposition shall be made unless all requisite regulatory and other approvals, consents and authorizations from all Governmental Authorities that are required to be obtained in connection with such Disposition have been obtained and as to which all conditions to the consummation of Disposition thereunder have been satisfied.
- 9.3 All costs associated with any Disposition of an Ownership Interest by an Owner shall be borne solely by the Offering Owner, <u>provided</u> that the foregoing shall not limit the Offering Owner's right to seek reimbursement of any costs from the Proposed Purchaser in connection with any such Disposition.
- Each Owner shall have the right to seek financing for all or a portion of such Owner's Ownership Interest and to provide general security for such financing of its Ownership Interest, including through the creation of any Encumbrance thereon (and the right of the beneficiary thereof to enforce thereon, but not to foreclose upon or transfer such Owner's Ownership Interest without the prior written consent of the other Owner), without the prior consent of the other Owner. Each Owner further agrees to cooperate reasonably and in good faith, and to cause its Affiliates to so cooperate, with an Owner seeking financing in connection with such modifications and other rights and consents customary in transactions of such type, and not unreasonably to withhold its consent to such modifications as may be reasonably necessary or appropriate to allow such Owner to obtain such financing upon reasonably competitive terms, including obtaining consents to the assignment of such Owner's Ownership Interest in any of the Project Assets reasonably requested by such Owner's lender; provided that none of such proposed modifications shall (a) relieve the financing Owner of any of its obligations under this Agreement, the O&M Agreement or any other agreement related to the Mitchell Plant or any Project Asset, (b) decrease the economic benefits, or increase the costs, of the ownership and operation of the Mitchell Plant to the other Owner, (c) create any increased economic or legal risk to the other Owner in connection with the ownership and operation of the Mitchell Plant, (d) permit or allow any Encumbrances relating to any such financing to be placed upon any portion of or interest in the Project Assets other than the financing Owner's Ownership Interest; or (e) permit partition of the Project Assets or any of them, including any partition upon a default by the financing Owner under any of the relevant financing documents or .
- 9.5 Notwithstanding anything else herein to the contrary, no Disposition shall constitute a release of the Offering Owner from any liabilities to the Non-Offering Owner or the Operator arising from events occurring prior to or in connection with the Disposition, except as may be.

### ARTICLE TEN

### DEFAULTS AND REMEDIES

10.1 An Owner shall be deemed to be in default hereunder upon the occurrence of any of the following events with respect to such Owner (each of the following events to be referred to as an "Event of Default," the Owner in default to be referred to as the "Defaulting Owner" and the Owner not in default to be referred to as the "Non-Defaulting Owner"):

- (a) an Owner fails to make any payment required by it as and when due and payable in accordance with the terms of this Agreement, the O&M Agreement or any other agreement related to the Mitchell Plant or any Project Asset and such failure is not remedied within ten (10) days after receipt of written notice thereof by such Owner from the other Owner; provided, that any such notice shall include a statement of the amount the Defaulting Owner has failed to pay (a "Payment Default"); or
- (b) an Owner fails to perform any material obligation (other than as described in Section 10.1(a)) imposed upon such Owner under this Agreement and such failure is not remedied within thirty (30) days after such Owner receives written notice thereof from the other; provided that, if such thirty (30) day period is not sufficient to enable the remedy or cure of such failure in performance, and such Owner shall have upon receipt of the initial notice promptly commenced and diligently continues thereafter to remedy such failure, then such Owner shall have a reasonable additional period of time (but in no event longer than an additional ninety (90) days from the end of the initial thirty (30) day cure period) to remedy or cure such failure; provided, however, that an Owner shall not be in default of its obligations hereunder to the extent such failure is caused by or is otherwise attributable to a breach by the other Owner of its obligations under this Agreement.
- 10.2 Without limiting the rights and remedies available to the Non-Defaulting Owner under Applicable Law, in the case of an Event of Default, the Non-Defaulting Owner shall have the right (but not the obligation) to (x) pay all or a portion of the amounts that were the subject of the Payment Default on behalf of the Defaulting Owner and (y) perform the obligation(s) which the Defaulting Owner has failed to perform on behalf of and at the expense of the Defaulting Owner (in any such case subject to all limits on liability benefiting the Defaulting Owner as set forth in this Agreement); and, if such payment is made (the portion as so paid or expended in connection with such performance, the "Paid Amount"), to:
- (a) charge the Defaulting Owner interest with respect to the Paid Amount, from the day the payment was made by the Non-Defaulting Owner until it is paid in full by the Defaulting Owner to the Non-Defaulting Owner, at the rate equal to the prime rate as published from time to time in *The Wall Street Journal* (or any successor publication) plus five (5) percentage points per annum, calculated daily, regardless of whether the Non-Defaulting Owner has notified the Defaulting Owner in advance of its intention to charge interest with respect to such Paid Amount;
- (b) set off against the Paid Amount any sums due or accruing to the Defaulting Owner by the Non-Defaulting Owner in accordance with this Agreement;
- (c) maintain an action or actions for the Paid Amount and interest thereon on a continuing basis as the Paid Amount becomes payable but is not paid by the Defaulting Owner, as if the obligation to pay those amounts and the interest thereon was a liquidated demand due and payable on the date the amounts were due to be paid, without any right or resort of the Defaulting Owner to set-off or counter-claim against the Non-Defaulting Owner; and any obligation to pay interest under this <u>Section 10.2</u> shall apply until the Payment Default is rectified or remedied; and

at the Non-Defaulting Owner's option, (i) draw on any letter of credit posted by the Defaulting Owner pursuant to Section 4.3 in an amount equal to the Paid Amount, including all interest accrued thereon or (ii) receive one hundred percent (100%) of any revenues arising from or attributable to the sale of capacity, energy, ancillary services or other energy products from the Mitchell Plant that the Defaulting Owner would otherwise be entitled to receive in respect of its Assigned Capacity until the Non-Defaulting Owner receives an amount equal to the Paid Amount, including all interest accrued thereon, plus all costs of collection incurred in connection therewith, and the Owners shall cooperate with each other, the Operator, applicable Governmental Authorities (including in respect of securing any regulatory approvals) or other third parties (including lenders) as may be reasonably necessary to facilitate the Non Defaulting Owner's right to be paid and receive the revenues attributable to the Defaulting Owner's Assigned Capacity until the applicable Paid Amount, including all interest accrued thereon and all costs of collection incurred in connection therewith has been paid to the Non-Defaulting Owner in full, including facilitating any appropriate changes in the applicable settlement accounts with respect to which market revenues are credited or paid by PJM or other applicable regional transmission organizations and executing any documents required to assign over such market revenues to the Non-Defaulting Owner.

### ARTICLE ELEVEN

### LIMITATION OF LIABILITY

11.1 Without limiting any other provision of this Agreement, each Owner's liability under this Agreement shall be limited to direct actual damages only. Such direct actual damages shall be the sole and exclusive remedy with respect to all claims arising under this Agreement and all other remedies or damages at law or in equity with respect to claims arising under this Agreement are waived, and unless expressly provided herein, no Owner shall be liable for consequential, punitive, incidental, exemplary or indirect damages, lost profits or other business interruption damages, by statute, in tort or in contract, under any indemnity provision or otherwise, with respect to claims arising under this Agreement. It is the intent of the Owners that the limitations herein imposed on remedies and the measure of damages be without regard to the cause or causes related thereto, including the negligence of any Owner, whether such negligence be sole, joint or concurrent, or active or passive. Notwithstanding anything herein to the contrary, the limitations set forth in this Section 11.1 shall not limit or preclude any indemnification obligations of an Owner pursuant to Article Ten of the O&M Agreement, including with respect to indemnification for third-party claims.

### ARTICLE TWELVE

### DISPUTE RESOLUTION

12.1 If either Owner believes that a dispute (including a Technical Dispute ) has arisen as to the meaning or application of this Agreement, it shall submit a written description of the

disputed matter to the Operating Committee, and shall provide a copy of that submission to the other Owner.

- 12.2 If the Operating Committee does not reach agreement on the resolution of a dispute not constituting a Technical Dispute submitted to the Operating Committee pursuant to Section 12.1 within thirty (30) days after the dispute is presented to it, the matter shall be referred to senior executive officers with the authority to resolve such dispute of each of the Owners for resolution in the manner that such individuals shall agree is appropriate; provided, however, that either Owner may exercise any and all rights at law or equity at any time after the end of the thirty (30) day period provided for the Operating Committee to reach agreement if the Operating Committee has not reached agreement.
- If the Operating Committee does not reach agreement on the resolution of a Technical Dispute submitted to the Operating Committee within ten (10) business days after such Technical Dispute is presented to it, then either Owner may refer such Technical Dispute to a Technical Expert. Within ten (10) business days following receipt of an Owner's notice referring a Technical Dispute to a Technical Expert, the Operating Representatives shall confer to agree upon a Technical Expert to hear the Technical Dispute. If the Owners are unable to agree upon the appointment of a Technical Expert, then at the end of such ten (10) business day period each Owner shall, within five (5) business days, notify the other Owner in writing of its designation of a proposed Technical Expert. The two proposed Technical Experts shall, within five (5) business days, select a Technical Expert (who may be one of the two Technical Experts designated by the Owners or another Technical Expert) and such Technical Expert shall hear the Technical Dispute. Each Owner shall be required to put forth and endorse one proposal, budget or solution, as the case may be, as its proposed resolution to the Technical Dispute, based on an agreed statement of the nature of the Technical Dispute and agreed facts surrounding such Technical Dispute. Each Owner's proposal, budget or solution shall be delivered to the Technical Expert and the other Owner no later than twenty (20) business days after the date of the notice of the Owner submitting the Technical Dispute to the Technical Expert. The Technical Expert shall be guided by consideration of (a) this Agreement, (b) all other agreements between the Owners relating to the Mitchell Plant, including the O&M Agreement and (c) Prudent Operation and Maintenance Practices (as defined in the O&M Agreement), and be required to select one of the proposals, budgets or solutions, as the case may be, and shall not be able to select any other proposal, budget or solution, except to the extent mutually agreed by the Owners. The Technical Expert shall render a decision resolving the matter within forty-five (45) days of the date of the notice of the Owner submitting such matter. The Technical Expert shall not award to either Owner any relief greater than that initially sought by such Owner. The decision of the Technical Expert shall be final and binding upon the Owners and not subject to appeal or review. The Owners shall bear equally all costs and expenses of the Technical Expert procedure and the Technical Expert shall not have the authority to award costs or attorneys' fees to either Owner. The Technical Expert shall act as an expert and not as an arbitrator and the provisions of the Federal Arbitration Act and the laws relating to arbitration shall not apply to the Technical Expert or the Technical Expert's determination or the procedure by which a determination is reached. Except as provided in Section 7.2(a), the Technical Expert's decision shall not in any event result in deviations from the agreed allocations of costs between the Owners as set forth in this Agreement.

- 12.4 Except as provided in this <u>Article Twelve</u>, the existence, contents, or results of any settlement negotiations or the results thereof under this <u>Article Twelve</u> may not be disclosed without the prior written consent of the Owners, <u>provided</u>, <u>however</u>, that either Owner may make disclosures as may be required to fulfill regulatory obligations to any Governmental Authority having jurisdiction, and may inform its lenders, affiliates, auditors, and insurers, as necessary, under pledge of confidentiality, and may consult with expert consultants as required in connection with any proceeding under pledge of confidentiality.
- 12.5 Nothing in this Agreement shall be construed to preclude either Owner from filing a petition or complaint with FERC with respect to any claim over which FERC has jurisdiction. In such case, the other Owner may request that FERC reject the petition or complaint or otherwise decline to exercise its jurisdiction. If FERC declines to act with respect to all or part of a claim, the portion of the claim not so accepted by FERC may be resolved through an action at law or equity. To the extent that FERC asserts or accepts jurisdiction over all or part of a claim, the decisions, findings of fact, or orders of FERC shall be final and binding, subject to judicial review under the Federal Power Act, 16 U.S.C. §§ 791a et seq., as amended from time to time, and any proceedings that may have commenced prior to the assertion or acceptance of jurisdiction by FERC shall be stayed, pending the outcome of FERC proceedings. To the extent that any decisions, findings of fact, or orders of FERC do not provide a final or complete remedy to an Owner seeking relief, such Owner may proceed at law or equity to secure such a remedy, subject to any FERC decisions, findings, or orders.
- 12.6 If an Owner (the "Contesting Owner") contests in good faith any amount paid pursuant to the terms of this Agreement following receipt of the written notice of the other Owner delivered pursuant to Section 10.1(a), and any portion of such amount is determined or resolved (including pursuant to the dispute resolution procedures of this Article Twelve) to be in excess of the actual amount due pursuant to the terms of this Agreement, then the Contesting Owner may charge the other Owner interest with respect to such excess amount from the day the payment was made until it is repaid to the Contesting Owner, at the rate equal to the prime rate as published from time to time in The Wall Street Journal (or any successor publication) plus five (5) percentage points per annum, calculated daily, regardless of whether the Contesting Owner has notified the other Owner in advance of its intention to charge interest with respect to such excess amount, and the other Owner shall make payment in full in respect of such excess amount and interest within thirty (30) days of written demand therefor.

### ARTICLE THIRTEEN

### **GENERAL**

13.1 This Agreement shall inure to the benefit of and be binding upon the signatories hereto and their respective successors and permitted assigns, but this Agreement may not be assigned by any signatory without the written consent of the other parties hereto or as permitted by Article Nine hereof.

- 13.2 This Agreement and all of its terms, including those addressing future events, is subject to the regulatory authority of any state or federal agency having jurisdiction, including the KPSC and the WVPSC. Each Owner is responsible for compliance with the orders of its state public service commission regarding ELG Upgrades and associated costs, including maintenance capital and landfill capital expense, incurred to keep the Mitchell Plant operating after the date the Mitchell Plant would otherwise be required to retire under the ELG Rule in the absence of the ELG Upgrades, including KPSC Case Nos. 2021-00004 and 2021-00421, with respect to KPCo and WVPSC Case Nos. 20-1040-E-CN and 21-0810-E-PC with respect to WPCo.
- The interpretation and performance of this Agreement is governed by and shall be construed in accordance with the laws of the State of New York, exclusive of the conflicts of law provisions thereof that would require the application of the laws of a different jurisdiction. Each Owner hereby agrees that any Action arising out of or relating to this Agreement brought by an Owner (or any of their respective successors or assigns) shall be brought and determined in any state or federal court sitting in the State of New York, within the Borough of Manhattan, City of New York, and the Owners hereby irrevocably submit to the exclusive jurisdiction of the aforesaid courts for themselves and with respect to their property, generally and unconditionally, with regard to any such Action arising out of or relating to this Agreement and the transactions contemplated hereby, and the appellate courts from any thereof in connection with any action arising out of or relating to this Agreement or any other agreement related to the Mitchell Plant or any Project Asset and the transactions contemplated hereby, and consents that any such action may be brought in such courts and waives any objection it may now or hereafter have to the venue of any such action in any such court or that such action was brought in an inconvenient court. EACH OWNER HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ALL RIGHTS TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE O&M AGREEMENT, OR ANY OTHER AGREEMENT RELATED TO THE MITCHELL PLANT OR ANY PROJECT ASSET.
- 13.4 This Agreement supersedes all previous representations, understandings, negotiations, and agreements, either written or oral between the signatories hereto or their representatives with respect to operation of the Mitchell Plant, including the Original Operating Agreement. Notwithstanding the foregoing, the amendment and restatement of the Original Operating Agreement effected hereby shall not relieve any party thereto of any undischarged obligation or liability of such party in respect of the period prior to the Effective Date under the Original Operating Agreement. This Agreement, together with the O&M Agreement (and any replacements thereof), constitutes the entire agreement of the signatories hereto with respect to the operation of the Mitchell Plant and the ownership thereof. The signatories hereto hereby agree that this Agreement shall amend the Original Operating Agreement to irrevocably remove AEPSC as a party thereto and, on and after the Effective Date, AEPSC shall no longer be a party thereto or hereto or entitled to rights, or subject to obligations, as a party to this Agreement; provided, however, that Operator shall be permitted to delegate any of its rights, duties and obligations under this Agreement and the O&M Agreement to AEPSC without the consent of KPCo, but without relieving Operator of any of its obligations hereunder.

21

- 13.5 No amendments or modifications of this Agreement are valid unless in writing and signed by duly authorized representatives of the Owners.
- 13.6 Each Owner shall designate in writing a representative to receive any and all notices required under this Agreement. Notices shall be in writing and shall be given to the representative designated to receive them, either by personal delivery, certified mail, e-mail or any similar means, properly addressed to such representative at the address specified below:

[]
[]
Attn:
Phone: []
Email:[]
WHEELING POWER COMPANY
[]
[]
Attn:
Phone: []
Email:[]

KENTUCKY POWER COMPANY

All notices shall be deemed to have been given (a) when personally delivered, (b) when transmitted (except if not a Business Day then the next Business Day) via electronic mail (provided that no error message or other notification of non-delivery is generated with respect to the intended recipient), (c) the day following the day (except if not a Business Day then the next Business Day) on which the same has been delivered prepaid to a reputable national overnight air courier service or (d) the third Business Day following the day on which the same is sent by certified or registered mail, postage prepaid, in each case to the respective parties hereto at the address set forth below, or at such other address as such Owner may specify by written notice to the other Owner (or at such other address for an Owner as shall be specified in a notice given in accordance with this Section 13.6). Each Owner may, by written notice to the other Owner, change the representative or the address to which such notices are to be sent.

13.7 This Agreement may be executed in any number of counterparts, and each such counterpart hereof shall be deemed to be an original instrument, but all of such counterparts shall

constitute for all purposes one agreement. Any signature hereto delivered by a party hereto by facsimile or other electronic transmission shall be deemed an original signature hereto.

- 13.8 Except as otherwise specifically provided, all fees, costs and expenses incurred by the parties hereto in negotiating this Agreement shall be paid by the party incurring the same, including legal and accounting fees, costs and expenses.
- 13.9 Any of the terms, covenants, or conditions hereof may be waived only by a written instrument executed by or on behalf of the Owners waiving compliance. No course of dealing on the part of any Owners, or its respective officers, employees, agents, accountants, attorneys, investment bankers, consultants or other authorized representatives, nor any failure by an Owner to exercise any of its rights under this Agreement shall operate as a waiver thereof or affect in any way the right of such Owner at a later time to enforce the performance of such provision. No waiver by any Owner of any condition, or any breach of any term or covenant contained in this Agreement, in any one or more instances, shall be deemed to be or construed as a further or continuing waiver of any such condition or breach or a waiver of any other condition or of any breach of any other term or covenant. The rights of the Owners under this Agreement shall be cumulative, and the exercise or partial exercise of any such right shall not preclude the exercise of any other right.
- 13.10 This Agreement shall be binding upon and inure to the benefit of the Owners and their respective successors and permitted assigns.
- 13.11 No Owner will issue, or permit any of its Affiliates, its or its Affiliate's directors, officers, employees, consultants, agents or other representatives to issue, any press releases or otherwise make, or cause any of its Affiliates, its or its Affiliate's directors, officers, employees, consultants, agents or other representatives to make, any public statements or other public disclosures with respect to this Agreement, or the transactions contemplated hereby without the prior written consent of the other Owner; provided, however, that the foregoing requirement to obtain prior written consent shall not apply where such release, statement or disclosure is deemed in good faith by the releasing or disclosing Owner to be required by Applicable Law or under the rules and regulations of a recognized stock exchange on which shares of such Owner (or any of its Affiliates) are listed, so long as prior to making any such release, statement or disclosure and to the extent legally permitted, the releasing or disclosing Owner shall provide prompt notice to the other Owner, consult the other Owner as to the form, contents and timing of such release or disclosure and, when available, provide a copy of such release, statement or disclosure containing such information to the other Owner.
- 13.12 If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Owners shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Owners as closely as possible in an acceptable manner to the end that the transactions contemplated hereby are fulfilled to the extent possible.

13.13 Each Owner acknowledges that it shall be inadequate or impossible, or both, to measure in money the damage to the Members if any of them or any transferee or any legal representative of any Owner fails to comply with any of the restrictions or obligations imposed by Article Nine that every such restriction and obligation is material, and that in the event of any such failure, the Owners shall not have an adequate remedy at law or in damages. Therefore, each Owner consents to the issuance of an injunction or the enforcement of other equitable remedies against such Owner at the suit of an aggrieved party without the posting of any bond or other security, to compel specific performance of all of the terms of Article Nine and to prevent any Disposition in contravention of any terms of Article Nine, and waives any defenses thereto, including the defenses of: (i) failure of consideration, (ii) breach of any other provision of this Agreement and (iii) availability of relief in monetary damages.

### ARTICLE FOURTEEN

### **DEFINITIONS**

For all purposes of this Agreement (including the preceding sections and recitals), unless otherwise required by the context in which any defined term appears or otherwise defined in the body of this Agreement, capitalized terms have the meanings specified in this Article Fourteen. In this Agreement, unless expressly stated otherwise: (a) reference to any agreement (including this Agreement), document or instrument means such agreement, document or instrument as has been, or may be, amended, supplemented or otherwise modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof; (b) reference to any Applicable Law means such Applicable Law as has been, or may be, amended, modified, codified or reenacted, in whole or in part, and in effect from time to time, including rules and regulations, promulgated thereunder; (c) the singular includes the plural, as the context requires; (d) the terms "includes" and "including" mean "including, but not limited to"; (e) "Day" (regardless of capitalization) shall mean a calendar day, unless specifically designated as a Business Day or business day; (f) "Month" (regardless of capitalization) shall mean a calendar month; (vii) references to articles, sections and appendices mean the articles and sections of, and appendices to, this Agreement.

"AEPSC" shall have the meaning given to such term in the Preamble. "Agreement" shall have the meaning given to such term in the Preamble.

"Applicable Law" shall mean all laws (including common law), statutes, codes, acts, treaties, ordinances, orders, judgments, writs, decrees, injunctions, rules, regulations, governmental approvals, permits, directives, and requirements of all Governmental Authorities (including with respect to the environment) having jurisdiction over an Owner, any other person or entity (as to that person or entity), this Agreement, any Project Asset or the Mitchell Plant, as applicable

"Assigned Capacity" shall have the meaning given to such term in Section 2.3.

"Capital Budget" shall have the meaning given to such term in Section 1.7.

"CertainTeed Contract" shall mean that certain Supply Agreement dated March 11, 2005, by and between CertainTeed Gypsum West Virginia, Inc. (formerly BPB West Virginia Inc.) and KPCo (as assignee of Ohio Power Company), as amended by Amendment No. 2010-1 dated August 2, 2010, as further amended by Amendment No. 2012-1 dated February 20, 2012 and as further amended by Amendment No. 2013-1 dated June 5, 2013, as may be amended, amended and restated, supplemented or modified from time to time, and as may be assigned to Operator or an Affiliate of Operator.

"CCR Capital Expenditures" shall mean all capital expenditures associated with implementation of the CCR Upgrades.

"<u>CCR Rule</u>" means the Coal Combustion Residuals Rule, 40 CFR Part 257 (April 17, 2015, as amended), and any regulations thereunder promulgated by the USEPA or the State of West Virginia.

"CCR Upgrades" shall mean any improvements or upgrades to the Mitchell Plant to enable KPCo and WPCo to comply with the CCR Rule.

"Control" shall have the meaning given to such term in Section 7.9.

"Credit Rating" means with respect to any entity, the rating then assigned to such entity's unsecured, senior long-term debt or deposit obligations (not supported by third party credit enhancements) by S&P or Moody's. If no rating is assigned to such entity's unsecured, senior long-term debt or deposit obligations by S&P or Moody's, then "Credit Rating" means the general corporate credit rating or long-term issuer rating assigned to such entity by S&P or Moody's. If an entity is rated by both S&P and Moody's and the ratings are at different levels, then "Credit Rating" means the lowest such rating.

"<u>Decommission</u>" or "<u>Decommissioning</u>" shall mean the retirement, dismantlement and permanent removal of the Units and other property, plant, and equipment comprising the Mitchell Plant, including any common facilities associated with each Unit that are to be permanently removed from service, the restoration of the Mitchell Plant site and the removal or remediation of any hazardous materials or other contaminated equipment, materials, coal ash or wastes associated therewith, in a manner that meets the requirements of Applicable Law.

"Decommissioning Costs" shall mean all costs and obligations expended or incurred in the performance of all work reasonably necessary or undertaken to Decommission the Mitchell Plant, including work associated with the preparation and implementation of Decommissioning plans and the preparation, submittal and prosecution of all necessary applications with Governmental Authorities as required to Decommission the Mitchell Plant in accordance with Applicable Law.

"Defaulting Owner" shall have the meaning given to such term in Section 10.1.

"<u>Depreciable Life</u>" means, with respect to a capital item, the shorter of (a) the reasonably expected depreciable life (in months) of such capital item and (b) the number of months between

the anticipated in-service date of such capital item and December 31, 2040 (or such earlier anticipated date of the permanent cessation of operations of the Units filed with the WVPSC).

"Dispose" or "Disposition" shall have the meaning given to such term in Section 9.1.

Effective Date" shall have the meaning given to such term in the Preamble.

"ELG Capital Expenditures" shall mean all capital expenditures associated with implementation of the ELG Upgrades.

"ELG Rule" shall mean the Steam Electric Reconsideration Rule, 85 Fed. Reg. 64,650 (Oct. 13, 2020), and any regulations thereunder promulgated by the USEPA or the State of West Virginia.

"ELG Upgrades" shall mean any improvements or upgrades to the Mitchell Plant to enable WPCo to comply with the ELG Rule.

"Emission Allowances" shall have the meaning given to such term in Section 7.7.

"Encumbrance" shall mean with respect to any property or asset (a) any mortgage, deed of trust, charge, lien, pledge, hypothecation, title retention arrangement or other security interest, as or in effect as security for the payment of a monetary obligation or the observance of any other obligation; (b) any easement, servitude, restrictive covenant, equity or interest in the nature of an encumbrance, garnishee order, writ of execution, right of set-off, lease, license to use or occupy, assignment of income or monetary claim, whether or not filed, recorded or otherwise perfected under Applicable Law; and (c) any agreement to create any of the foregoing or allow any of the foregoing to exist.

"Event of Default" shall have the meaning given to such term in Section 10.1.

"FERC" shall have the meaning given to such term in Section 5.1.

"FERC Accounting Requirements" means the accounting requirements of FERC, including with respect to the Uniform System of Accounts, established by FERC under the FPA.

"FPA" means the Federal Power Act.

"Governmental Authority" means any federal, national, regional, state, municipal or local government authority, tribunal, court, agency, body, board or instrumentality, or any regulatory, administrative or other department, commission, bureau or agency, taxing authority or power, or any political or other subdivision, department or branch of the foregoing, including any independent system operator, regional transmission organization or electric reliability organization.

"KPCo" shall have the meaning given to such term in the Preamble.

"KPSC" shall mean the Kentucky Public Service Commission.

"Mitchell Plant" shall mean the Mitchell Power Generation Facility, which consists of the Units and associated plant, equipment, real estate and other related facilities, located in Moundsville, West Virginia, but excluding the real property and operation known as the Conner Run Fly Ash Impoundment and Dam.

"Moody's" shall have the meaning given to such term in Section 4.3.

"Non-Defaulting Owner" shall have the meaning given to such term in Section 10.1.

"Non-Offering Owner" shall have the meaning given to such term in Section 9.1.

"O&M Agreement" shall have the meaning given to such term in the Recitals.

"Offering Owner" shall have the meaning given to such term in Section 9.1.

"Operating Committee" shall have the meaning given to such term in Section 7.1.

"Operating. Representative" shall have the meaning given to such term in Section 7.1.

"Operator" shall have the meaning given to such term in the Recitals.

"Original Operating Agreement" shall have the meaning given to such term in the Recitals.

"Owner" or "Owners" shall have the meaning given to such term in the Preamble.

"Ownership Interest" shall have the meaning given to such term in the Recitals.

"Paid Amount" shall have the meaning given to such term in Section 10.2.

"Payment Default" shall have the meaning given to such term in Section 10.1(a).

"Project Assets" shall have the meaning given to such term in Section 1.1.

"Proposed Purchaser" shall have the meaning given to such term in Section 9.1.

"Ratings Requirement" shall mean a Credit Rating for such Owner (or if such Owner has provided a guaranty issued by an Affiliate to satisfy its obligations under this Section 4.3, such Owner's Affiliate guarantor) of at least "BBB-" by S&P or at least Baa3 by Moody's, and if such Credit Rating is "BBB-" by S&P or "Baa3" by Moody's then such Credit Rating must not be on negative credit watch by S&P or Moody's.

"S&P" shall have the meaning given to such term in Section 4.3.

"Tax Code" shall have the meaning given to such term in Section 6.6.

"<u>Technical Dispute</u>" shall mean any dispute which this Agreement expressly provides shall be a Technical Dispute.

"Technical Expert" shall mean any individual selected in accordance with the procedure specified in Section 12.3 and who (a) has significant professional qualifications and practical experience in the subject matter of the Technical Dispute, (b) has no interest, financial or otherwise, or duty which conflicts or may conflict with such individual's functions as a Technical Expert (such individual being required to fully disclose any such interest or duty prior to any appointment) and (c) is not currently and has not been (i) during the five (5) years prior to the date of appointment, an employee of any of the Owners or any of their Affiliates and (ii) during the three (3) years prior to the date of appointment, a contractor or consultant of either of the Owners or any of their Affiliates, unless otherwise mutually agreed by the Owners.

"Term" shall have the meaning given to such term in Section 8.2.

"Total Net Capability" shall have the meaning given to such term in Section 2.1.

"Total Net Generation" shall have the meaning given to such term in Section 2.2.

"Unit" shall have the meaning given to such term in the Recitals.

"USEPA" shall have the meaning given to such term in Section 7.7.

"WACC" shall mean, as of any date, WPCo's then-applicable WVPSC-authorized weighted average cost of capital, compounded semiannually (consistent with the compounding of Allowance for Funds Used During Construction (AFUDC)).

"WPCo" shall have the meaning given to such term in the Preamble. "WVPSC" shall mean the Public Service Commission of West Virginia

[Signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their officers thereunto duly authorized as of the date first above written.

Ву:	
Title:	
WHEELING POWER COMPANY	
Ву:	
Title:	

KENTUCKY POWER COMPANY

Solely with respect to <u>Section 13.4</u> :
AMERICAN ELECTRIC POWER SERVICE CORPORATION
By: Title:

## Exhibit A

### Capital Budget, Initial Budgets and Forecast

[To Be Attached as of the Effective Date.]

## Exhibit B

## Form of Monthly Sample Report

[To Be Attached as of the Effective Date.]

### PUBLIC SERVICE COMMISSION OF WEST VIRGINIA CHARLESTON

CASE NO. 21-0810-E-PC (Closed Filing)

APPALACHIAN POWER COMPANY and WHEELING POWER COMPANY, public utilities.

Petition for Commission Consent and Approval to Enter into Ownership and Operating Agreements for the Mitchell Plant Pursuant to W. Va. Code § 24-2-12

### **CERTIFICATE OF SERVICE**

I, Anne C. Blankenship, counsel for Appalachian Power Company and Wheeling Power Company, hereby certify that true copies of the foregoing filing were provided electronically on this 11th day of July, 2022, addressed to the following:

Lucas R. Head, Esquire
Public Service Commission
of West Virginia
201 Brooks Street
Charleston, West Virginia 25301
Counsel for Public Service Commission

Susan J. Riggs, Esquire
Jason C. Pizatella, Esquire
Spilman Thomas & Battle, PLLC
300 Kanawha Blvd., East
Charleston, WV 25301
Counsel for West Virginia Economic
Development Authority

H. Brann Altmeyer, Esquire
Jacob C. Altmeyer, Esquire
Phillips, Gardill, Kaiser & Altmeyer, PLLC
61 Fourteenth Street
Wheeling, WV 26003
Counsel for West Virginia Coal Association, Inc.

Robert F. Williams, Esquire
Heather B. Osborn, Esquire
John Auville, Esquire
Consumer Advocate Division
300 Capitol Street, Suite 810
Charleston, West Virginia 25301
Counsel for Consumer Advocate Division

Derrick P. Williamson, Esquire Barry A. Naum, Esquire Spilman Thomas & Battle, PLLC 1100 Bent Creek Blvd., Suite 101 Mechanicsburg, PA 17050 Counsel for WVEUG

Emmett Pepper, Esquire Pepper & Nason 8 Hale Street Charleston, WV 25301 Counsel for CAG/SUN/EEWV Raghava Murthy, Esquire Shannon W. Fisk, Esquire Earthjustice 48 Wall St., 15th Floor New York, NY 10005 Counsel for CAG/SUN/EEWV

Anne C. Blankenshir

# EXHIBIT 2





ANNE C. BLANKENSHIP ATTORNEY AT LAW

P.O. BOX 1791 CHARLESTON, WV 25326

CHARLESTON OFFICE: (304) 344-5800 DIRECT DIAL: (304) 347-8352 FACSIMILE: (304) 344-9566 E-MAIL: acb@ramlaw.com

July 8, 2022

Via Electronic Filing

Karen Buckley, Executive Secretary Public Service Commission of West Virginia 201 Brooks Street Charleston, WV 25301

Re: Case No. 21-0810-E-PC (Closed Entry)
Appalachian Power Company and
Wheeling Power Company

Dear Ms. Buckley:

On behalf of Appalachian Power Company and Wheeling Power Company (together, "the Companies"), please find enclosed for filing in the above-referenced case, the **Motion for Extension of Deadline**. A copy of the motion has been provided to all parties of record.

Thank you for your assistance. Please do not hesitate to contact me with any questions.

Very truly yours,

Anne C. Blankenship

Counsel for Appalachian Power Company and

Wheeling Power Company

Enclosure cc: Service List

# PUBLIC SERVICE COMMISSION OF WEST VIRGINIA CHARLESTON

CASE NO. 21-0810-E-PC (Closed Filing)

APPALACHIAN POWER COMPANY and WHEELING POWER COMPANY, public utilities.

Petition for Commission Consent and Approval to Enter into Ownership and Operating Agreements for the Mitchell Plant Pursuant to W. Va. Code § 24-2-12

### MOTION FOR EXTENSION OF DEADLINE

COME NOW Appalachian Power Company ("APCo") and Wheeling Power Company ("WPCo") (together, "the Companies"), and file this motion seeking an extension of time to file a public version of responses to the discovery requests filed by the Consumer Advocate Division ("CAD") as required in the Commission's July 1, 2022 Order in this proceeding ("the Order"). In support of this motion, the Companies state the following:

1. On Friday, July 1, 2022, the Commission entered its Order in this case, which, among other things, addressed the Companies' Motion for Protective Treatment of confidential responses to the CAD's discovery requests 1.4 and 1.5. Because the responses to those requests were extremely voluminous and the responses were directed to be filed within five (5) days, it was impracticable for redactions to be made and the Companies did not provide public, redacted versions of the confidential information. However, the Companies filed non-public, unredacted versions under seal with the Commission and provided those to the parties who had executed interim protective agreements with the Companies.

- 2. In its July 1, 2022 Order, the Commission required the Companies to provide redacted public versions of their responses to CAD requests 1.4 and 1.5 within ten (10) days of the date of the Order. (The Order also required public versions of Commission Post Hearing Exhibit 1, Attachment 1. However, the Companies have already filed both redacted and unredacted versions of that document.) The confidential discovery responses include the following: (1) a Confidential Information Memorandum related to the sale of KPCo to Liberty Utilities Co. ("Liberty") and involving pre-sale strategy; (2) multiple drafts of the Stock Purchase Agreement, Disclosure Schedules, Mitchell Plant Ownership Agreement, and Mitchell Plant Operations and Maintenance Agreement exchanged between American Electric Power Company, Inc. ("AEP") and Liberty in the process of developing the final versions that were originally filed in both redacted and unredacted versions.
- 3. The Companies intend to comply with the Commission's Order but will need additional time in order to do so. Having received the Order late on Friday, July 1, 2022, the Companies have only had a few working days before the redacted documents are due. The documents contain approximately 2,600 pages and will require a thorough and time-consuming review and analysis to determine and make the necessary redactions. The Companies already have personnel engaged in this process and intend to complete it as expeditiously as possible, but they require additional time to do so. Accordingly, the Companies respectfully seek an extension until August 10, 2022, to submit the redacted documents, with the hope that they can complete the task sooner than that and make the filing as soon as the redacted documents have been completed.

WHEREFORE, the Companies respectfully request that the Commission grant this motion and extend the deadline for the Companies to provide the public redacted discovery responses to August 10, 2022.

### Respectfully submitted,

# APPALACHIAN POWER COMPANY WHEELING POWER COMPANY

By Counsel,

William C. Porth (WV State Bar #2943) wcp@ramlaw.com Anne C. Blankenship (WV State Bar #9044) acb@ramlaw.com

Robinson & McElwee PLLC

P.O. Box 1791

Charleston, WV 25326

James R. Bacha American Electric Power Service Corporation 1 Riverside Plaza Columbus, Ohio 43215 jrbacha@aep.com

Keith D. Fisher (WV State Bar #11346) American Electric Power Service Corporation 500 Lee Street East, Suite 800 Charleston, West Virginia 25301 (304) 348-4154 kdfisher@aep.com

Counsel for Appalachian Power Company and Wheeling Power Company

Dated: July 8, 2022

# PUBLIC SERVICE COMMISSION OF WEST VIRGINIA CHARLESTON

CASE NO. 21-0810-E-PC (Closed Filing)

APPALACHIAN POWER COMPANY and WHEELING POWER COMPANY, public utilities.

Petition for Commission Consent and Approval to Enter into Ownership and Operating Agreements for the Mitchell Plant Pursuant to W. Va. Code § 24-2-12

#### CERTIFICATE OF SERVICE

I, Anne C. Blankenship, counsel for Appalachian Power Company and Wheeling Power Company, hereby certify that true copies of the foregoing filing were provided electronically on this 8<sup>th</sup> day of July, 2022, addressed to the following:

Lucas R. Head, Esquire
Public Service Commission
of West Virginia
201 Brooks Street
Charleston, West Virginia 25301
Counsel for Public Service Commission

Susan J. Riggs, Esquire
Jason C. Pizatella, Esquire
Spilman Thomas & Battle, PLLC
300 Kanawha Blvd., East
Charleston, WV 25301
Counsel for West Virginia Economic
Development Authority

H. Brann Altmeyer, Esquire
Jacob C. Altmeyer, Esquire
Phillips, Gardill, Kaiser & Altmeyer, PLLC
61 Fourteenth Street
Wheeling, WV 26003
Counsel for West Virginia Coal Association, Inc.

Robert F. Williams, Esquire Heather B. Osborn, Esquire John Auville, Esquire Consumer Advocate Division 300 Capitol Street, Suite 810 Charleston, West Virginia 25301 Counsel for Consumer Advocate Division

Derrick P. Williamson, Esquire Barry A. Naum, Esquire Spilman Thomas & Battle, PLLC 1100 Bent Creek Blvd., Suite 101 Mechanicsburg, PA 17050 Counsel for WVEUG

Emmett Pepper, Esquire
Pepper & Nason
8 Hale Street
Charleston, WV 25301
Counsel for CAG/SUN/EEWV

Raghava Murthy, Esquire Shannon W. Fisk, Esquire Earthjustice 48 Wall St., 15th Floor New York, NY 10005 Counsel for CAG/SUN/EEWV

Anne C. Blankenship

# EXHIBIT 3

# PUBLIC SERVICE COMMISSION OF WEST VIRGINIA CHARLESTON

At a session of the PUBLIC SERVICE COMMISSION OF WEST VIRGINIA in the City of Charleston on the 8<sup>th</sup> day of July 2022.

CASE NO. 21-0810-E-PC

APPALACHIAN POWER COMPANY and WHEELING POWER COMPANY, public utilities.

Petition for consent and approval to enter into Ownership and Operating Agreements for the Mitchell Plant.

### **COMMISSION ORDER**

The Commission corrects an error in the final Commission Order entered on July 1, 2022.

### **BACKGROUND**

On November 19, 2021, Appalachian Power Company and Wheeling Power Company (together, Companies) filed a petition (Petition) for consent and approval to enter into Ownership and Operating Agreements for the Mitchell Plant that is the subject of this case (sometimes referred to as Mitchell). As a part of that Petition, the Companies requested exemption from the requirement to provide the financial information specified in Rule 21 of the Commission Rules of Practice and Procedure (Procedural Rules), 150 C.S.R. 1. Petition at 2.

On July 1, 2022, the Commission entered a final order in this case.

### **DISCUSSION**

The July 1, 2022 Commission Order, in part, required the Companies to file a public version of responses to Consumer Advocate Division (CAD) discovery requests 1.4 and 1.5 as well as Commission Post Hearing Exh. 1, Attachment 1. A public version of Commission Post Hearing Exh. 1, Attachment 1 was filed by the Companies on April 15, 2022. By this Order, the Commission corrects the fifth ordering paragraph of the July 1, 2022 Commission Order to require public versions only of CAD discovery requests 1.4 and 1.5.

### **ORDER**

IT IS THEREFORE ORDERED that the fifth ordering paragraph of the July 1, 2022 Commission Order is corrected to order:

IT IS FURTHER ORDERED that the Companies shall file, within ten days of the date of this Order, a public version of responses to CAD discovery requests 1.4 and 1.5.

IT IS FURTHER ORDERED that upon entry of this Order this case shall be removed from the Commission's docket of open cases.

IT IS FURTHER ORDERED that the Executive Secretary of the Commission serve a copy of this Order by electronic service on all parties of record who have filed an e-service agreement, by United States First Class Mail on all parties of record who have not filed an e-service agreement, and on Staff by hand delivery.

A True Copy, Teste,

Karen Buckley, Executive Secretary

Varen Bucklay

SMS/ksf 210810ce

### **VERIFICATION**

The undersigned, Brian K. West, being duly sworn, deposes and says he is the Vice President, Regulatory & Finance for Kentucky Power Company, that he has personal knowledge of the matters set forth in the foregoing responses and the information contained therein is true and correct to the best of his information, knowledge, and belief.

		Brian K. West		
Commonwealth of Kentucky	)	Case No. 2021-00370		
County of Boyd	)	Case No. 2021-00370		
Subscribed and sworn before me, a Notary Public, by Brian K. West this 11 <sup>th</sup> day of July, 2022.				
Scott f. Bush Notary Public	op	SCCRT 3. 5/3/42P Notary Public		
My Commission Expires	yne.	Commonwealth of Kentucky Commission Number KYNP32110 My Commission Expires Jun 24, 2025		

Notary ID Number: <u>LYNP 32 110</u>